#### UNIVERSITY OF CALIFORNIA, SAN DIEGO

# **Shadows and Intersections**

A dissertation submitted in partial satisfaction of the requirements for the degree Doctor of Philosophy

in

Mathematics

by

Ameerah Naz Chowdhury

Committee in charge:

Professor Jacques Verstraëte, Chair Professor Fan Chung Graham Professor Ron Graham Professor Russell Impagliazzo Professor Jeff Remmel

2012

Copyright Ameerah Naz Chowdhury, 2012 All rights reserved. The dissertation of Ameerah Naz Chowdhury is approved, and it is acceptable in quality and form for publication on microfilm and electronically:

Chair

University of California, San Diego

2012

# DEDICATION

To my mother.

# TABLE OF CONTENTS

Signature Pag	ge.			iii
Dedication .	••••			iv
Table of Cont	ents			v
List of Figure	s			viii
Acknowledge	ments	5		ix
Vita and Publ	icatio	ns		x
Abstract of th	e Diss	sertation		xi
Chapter 1	Intro	oduction		1
-	1.1	Outlin	e of Thesis	2
	1.2	Main I	Results and Significance	2
		1.2.1	Shadows and Intersections in Vector Spaces	3
			1.2.1.1 Results	3
			1.2.1.2 Significance	4
			1.2.1.3 Future Work	4
		1.2.2	On a Conjecture of Frankl and Füredi	4
			1.2.2.1 Results	4
			1.2.2.2 Significance	5
			1.2.2.3 Future Work	5
		1.2.3	On the Manickam-Miklós-Singhi Conjecture	5
			1.2.3.1 Results	6
	1.3	Notati	on and Terminology	6
Chapter 2	Com	binatori	al Techniques	10
1	2.1	Shado	WS	11
		2.1.1	The Kruskal-Katona Theorem	11
		2.1.2	Lovász's Version of the Kruskal-Katona Theorem	13
		2.1.3	Shifting	13
		2.1.4	Properties of the Shift Operator	14
		2.1.5	Lovász's Result via Shifting	16
		2.1.6	Keevash's Proof of Lovász's Result	17
	2.2	Interse	ction Theorems	20
		2.2.1	The Erdős-Ko-Rado Theorem	20
		2.2.2	More Properties of the Shift Operator	20
		2.2.3	Erdős-Ko-Rado via Shifting	22
		2.2.4	Erdős-Ko-Rado via Shadows	23

		2.2.5	Erdős-Ko-Rado via Cyclic Permutations	24	
		2.2.6	Frankl's <i>r</i> -wise Intersection Theorem	25	
		2.2.7	Forbidding Triangles	27	
		2.2.8	On <i>t</i> -intersecting Families	28	
		2.2.9	The Hilton-Milner Theorem	29	
	2.3	Fisher	's Inequality	29	
		2.3.1	The de Bruijn-Erdős Theorem	30	
		2.3.2	Proof of the de Bruijn-Erdős Theorem	31	
		2.3.3	The Frankl-Füredi Conjecture	33	
		2.3.4	Convexity	33	
Chanton 2	A 1 a a 1	hundia Tr	achairman	26	
Chapter 5	Algei	Draic Ie		30	
	3.1	Fisher	s inequality via Linear Algebra	30	
		3.1.1	The Extremal Families in Fisher's Inequality	3/	
		3.1.2	Symmetric Designs	38	
		3.1.3	The $\lambda$ -design Conjecture	40	
	3.2	L-inte	rsecting Families	41	
		3.2.1	Polynomial Spaces	41	
		3.2.2	Proof of the Frankl-Wilson Theorem	43	
	3.3	Eigenv	values	44	
		3.3.1	Independent Sets	44	
		3.3.2	Erdős-Ko-Rado via Eigenvalues	46	
	3.4	Linear	Programming	47	
		3.4.1	Duality	48	
		3.4.2	The Fundamental Theorem of Linear Programming .	50	
		3.4.3	Integer Programming	51	
		3.4.4	Flows	52	
		3.4.5	Baranyai's Theorem	55	
Chapter 4	Proie	ctive P	rerequisites	58	
Chapter	4 1	Vector	Snace Analogs	59	
	7.1	<i>A</i> 1 1	What are Vector Space Analogs?	59	
		4.1.1 1 1 2	Significance	5) 60	
		4.1.2	Methods	60	
	12	4.1.5 The a	hinomial coefficients	61	
	4.2	The <i>q</i> -binomial coefficients $\dots \dots \dots$			
	4.5	The $q$ -Pascal Rule			
	4.4				
	4.5	Spread		00	
	4.0	w nich	10018 WOFK /	08	
		4.0.1		09	
			$4.0.1.1  \text{Snifting} \dots \dots$	69	
			4.6.1.2 Daykin's Proot	69	
			4.6.1.3 No Analog of the Colex Order	70	

	4.	6.2 Katona's	S Cyclic Permutation Method	71
	4.	6.3 Algebra	c Successes	72
		4.6.3.1	Modular Frankl-Wilson Theorem	73
		4.6.3.2	Bollobás's Two Families Theorem	74
		4.6.3.3	On <i>t</i> -intersecting Families	75
Chapter 5	Shadow	vs and Intersect	tions in Vector Spaces	76
	5.1 SI	hadows and an	Analog of Lovász's Theorem	78
	5.2 A	nalog of Frank	l's <i>r</i> -wise Intersection Theorem	81
	5.3 C	haracterizing E	Equality in the Base Case	84
	5.4 O	ther Results of	the Author	90
	5.	4.1 Analog	of the Hilton-Milner Theorem	90
	5.	4.2 Coloring	g the $q$ -Kneser Graph	92
	5.5 O	pen Problems		93
	5.	5.1 Forbiddi	ng Triangles	94
	5.	5.2 Baranya	i's Theorem	94
	5.	5.3 Cliques	in the $q$ -Kneser Graph $\ldots$	95
Chapter 6	On a Co	onjecture of Fr	ankl and Füredi	97
	6.1 R	esults		98
	6.2 O	old and New Co	onjectures	99
	6.3 Pi	roof of Theore	$m 6.1.1 \ldots \ldots$	100
	6.4 Pi	roof of Theore	$m 6.1.2 \ldots \ldots$	103
	6.5 A	ppendix		109
Chapter 7	On the	Manickam-Mi	klós-Singhi Conjecture	116
	7.1 N	onnegative Su	ms	116
	7.2 N	otation		117
	7.3 R	esults		118
	7.4 R	eductions to B	ase Cases	119
	7.5 T	he MMS Conje	ecture when $k$ is Three	122
	7.6 T	he MMS Conje	ecture when $k$ is Four $\ldots$	124
	7.7 O	pen Problems		125
Bibliography				128

# LIST OF FIGURES

Figure 2.1:	The Fano plane is an example of a projective plane	30
Figure 3.1:	A transportation network with flow and capacity denoted $f/c$	53

#### ACKNOWLEDGEMENTS

Over the course of my graduate career, many mathematicians have provided crucial advice and assistance. I am grateful to my advisor, Jacques Verstraëte, for his generosity, kindness, and patience. Ada Chan, Chris Godsil, and Rick Wilson deserve special thanks for instilling me with a love for mathematics and for their ongoing support. I am indebted to Benny Sudakov for allowing me to be a part of his tightly run ship at UCLA. My studies in Hungarian-style combinatorics have been enriched by László Babai's and Péter Frankl's wonderful unpublished manuscript *Linear Algebra Methods in Combinatorics* and by conversations with many members of the Rényi Institute, such as my coauthor, Balázs Patkós.

A number of funding agencies and institutes made my research visits productive and memorable. Specifically, I thank the NSF, CESRI, Fulbright, and the Rényi Institute for providing me with valuable experiences abroad. I am grateful to IPAM for hosting me during their CMA 2009 program, and especially to their associate director, Amber Puha, for her encouragement.

I am fortunate to have friends like Matthew Drescher and Tony Huynh, who fought valiantly beside me in the trenches, and who lived through it all, from cookies to cobras. Thanking Matthew with words for his exemplariness as a human being is impossible, but a backrub might suffice. Claudine Chen, Tristan McLoughlin, Carlos Pin, and Beverly Tomov always provided a friendly ear and a helping hand. Lastly and most importantly, I am grateful to my family for all of their sacrifices, big and small.

Chapter 5, in part, is a reprint of the material as it appears in "Shadows and intersections in vector spaces," 2010. Chowdhury, Ameera; Patkós, Balázs. *J. Combin. Theory Ser. A*, 117(8):1095–1106, 2010. The dissertation author was the primary investigator and author of this paper.

Chapter 6, in part, is a reprint of the material as it appears in "On a conjecture of Frankl and Füredi," 2011. Chowdhury, Ameera. *Electron. J. Combin.*, 18(1):Paper 56, 16, 2011. The dissertation author was the primary investigator and author of this paper.

Chapter 7, in part, is currently being prepared for submission for publication of the material. Chowdhury, Ameera. The dissertation author was the primary investigator and author of this material.

#### VITA

2004	B. S. in Mathematics, California Institute of Technology
2005	MMath in Combinatorics and Optimization, University of Water- loo
2010	Fulbright Fellow, Alfréd Rényi Institute of Mathematics
2012	Ph. D. in Mathematics, University of California, San Diego

#### PUBLICATIONS

A. Chowdhury, C. Godsil, and G. Royle, "Colouring Lines in Projective Space," *Journal of Combinatorial Theory, Series A*, 113(1): 39-52, 2006.

A. Blokhuis, A.E. Brouwer, A. Chowdhury, P. Frankl, T. Mussche, B. Patkós, and T. Szőnyi, "A Hilton-Milner Theorem for Vector Spaces," *Electr. J. Combin.* 17(1) (2010), R71.

A. Chowdhury and B. Patkós, "Shadows and Intersections in Vector Spaces," *Journal of Combinatorial Theory, Series A*, 117(8): 1095-1106, 2010.

A. Chowdhury, "On a Conjecture of Frankl and Füredi," *Electr. J. Combin.* 18(1) (2011), P56.

A. Chowdhury, "On the Manickam-Miklós-Singhi Conjecture," in preparation.

### ABSTRACT OF THE DISSERTATION

#### **Shadows and Intersections**

by

Ameerah Naz Chowdhury

Doctor of Philosophy in Mathematics

University of California, San Diego, 2012

Professor Jacques Verstraëte, Chair

This thesis makes contributions to extremal combinatorics, specifically extremal set theory questions and their analogs in other structures. Extremal set theory studies how large or small a family of subsets of a finite set X can be under various constraints. By replacing the set X with another finite object, one can pose similar questions about families of other structures. Remarkably, a question and its analogs essentially have the same answer, regardless of the object. Despite these similarities, not much is known about analogs because standard techniques do not always apply. Our main results establish analogs of extremal set theory results for structures such as vector spaces and subsums of a finite sum. We also study intersecting families and shadows in their classical context of sets by researching a conjecture of Frankl and Füredi.

# Chapter 1

# Introduction

This thesis makes contributions to the field of extremal combinatorics, specifically extremal set theory questions and their analogs for other discrete structures. Given a finite set X, the general problem in extremal set theory asks how large or small a family of subsets of X can be if it satisfies certain restrictions. Naturally, this type of question appears throughout mathematics, and so extremal set theory can be applied in areas ranging from topology [74] to theoretical computer science [40]. On the other hand, extremal set theory borrows tools from algebra and probability, and its connections to other branches of mathematics is one of its most beautiful features.

Two core concepts in extremal set theory are intersecting families and shadows. The main results for intersecting families are the Erdős-Ko-Rado and Hilton-Milner theorems, and the principal result for shadows is the Kruskal-Katona theorem. By defining suitable notions of "intersecting" and "shadow," one can find remarkable analogs of these theorems for other structures such as vector spaces and permutations. Tantalizingly, while many results about sets should generalize to different settings, not much is known about analogs because standard techniques do not always apply.

This thesis aims to further understanding of shadows and intersecting families in sets and other structures. In the latter case, the goals are to identify objects with analogs of the Erdős-Ko-Rado, Hilton-Milner, and Kruskal-Katona theorems and to find integrated approaches to their proofs. The ultimate objective is to have a unified theory that characterizes the structures for which analogs exist and that proves results simultaneously for broad classes of objects. Since this research applies to a variety of structures, it injects new questions, applications, and techniques into many areas. Vector space analogs, for example, influence the fields of finite geometry, algebraic combinatorics, and coding theory; analogs for permutations affect the theory of group representations. Each new object with an analog touches further areas.

# **1.1 Outline of Thesis**

We now outline the chapters of this thesis. We first present our main results in Section 1.2 and compile a list of frequently used notation in Section 1.3. In Chapter 2, we discuss combinatorial methods in extremal set theory, paying particular attention to the shifting technique. We then examine the use of algebraic methods in extremal set theory in Chapter 3. Next, we give background on vector spaces over finite fields in Chapter 4; we also highlight the difficulties in generalizing purely combinatorial techniques in extremal set theory to vector spaces, and discuss algebraic methods that have been successfully used to prove theorems about both sets and vector spaces. In Chapter 5, we demonstrate new combinatorial techniques for vector spaces by proving vector space analogs of both Lovász's version of the Kruskal-Katona theorem and Frankl's *r*-wise intersection theorem; our proof of the latter also yields a new proof of the Erdős-Ko-Rado theorem for vector spaces. We end Chapter 5 with remarks on three of our favorite open problems in this area. In Chapter 6, we discuss our results on a conjecture of Frankl and Füredi. Finally, in Chapter 7, we end by presenting some new results on the Manickam-Miklós-Singhi conjecture and discussing some related open problems.

# **1.2 Main Results and Significance**

A classical question in extremal set theory is to bound the size of a family of subsets of X whose members have size k and pairwise intersect; such a family is called *intersecting*. Erdős, Ko, and Rado [47] showed that if X is large enough, then the unique intersecting family of maximum size consists of the k-element subsets containing a fixed point. If no point of X may lie in all sets, then the Hilton-Milner theorem [70] determines

the largest intersecting family in this case.

Another important property of a family  $\mathscr{F}$  of *k*-element subsets of *X* is its *shadow*, which consists of all (k-1)-element subsets of *X* contained in at least one member of  $\mathscr{F}$ . Kruskal and Katona [75, 80] determined the minimum size of the shadow, and their result implies many others, such as the Erdős-Ko-Rado theorem. In practice, however, their theorem is not used in its full generality and a weaker but more convenient version due to Lovász [84, Ex 13.31(b)] is applied.

## **1.2.1** Shadows and Intersections in Vector Spaces

In Chapter 5, our main results show that striking analogs of extremal set theory results exist for vector spaces. Our proofs are not straightforward generalizations of the corresponding ones for sets because standard techniques do not often apply; one example is that the complement of a subset of X is another subset while the complement of a subspace of V is not another subspace. Consequently, we develop new methods.

#### 1.2.1.1 Results

Patkós and I [27] discovered a unified proof of Lovász's version of the Kruskal-Katona theorem that works for sets and vector spaces [77]. Our result, Theorem 5.0.12, is one of the first about shadows in vector spaces although the problem is over thirty years old [67]. Since Lovász's theorem has many corollaries, its analog provides a way to extend them to vector spaces. Applications are not straightforward, however, because combinatorial techniques do not often apply. Our Theorem 5.0.14 yields unified proofs of the Erdős-Ko-Rado and Frankl's *r*-wise intersecting theorems for sets and vector spaces as corollaries. Three nice features of our proofs are that they are inductive, don't involve tedious computations, and characterize the case of equality.

We also state, but do not prove our other results in this area, namely obtaining a vector space analog of the Hilton-Milner theorem, Theorem 5.4.1, and determining the chromatic number of the q-Kneser graph, Theorem 5.4.5 and Theorem 5.4.6. The latter results were motivated by the longstanding problem of coloring the Kneser graph, whose solution involved a novel use of algebraic topology [9, 83].

#### 1.2.1.2 Significance

Vector space analogs bring new questions and techniques to finite geometry since many of its problems can be reformulated in these terms. They also provide applications for the *q*-analog identities studied by algebraic combinatorialists. Recently, coding theorists such as Vardy are studying vector space analogs because they imply results about projective codes [18, 49, 50]. Since codes are used in communication systems, research in this area may yield practical applications.

#### 1.2.1.3 Future Work

Our Lovász analog, Theorem 5.0.12, establishes shadows as a viable method for proving vector space analogs such as Theorem 5.0.14, and more applications are expected. Recently, Wang [113] used Theorem 5.0.12 to prove a conjecture of Erdős, Faigle, and Kern [48]. The method of proof in our Frankl analog, Theorem 5.0.14, has also been used to prove results on *r*-cross intersecting families of sets [58]; Patkós and I are currently working with Frankl and Tokushige to extend these results to vector spaces.

## **1.2.2** On a Conjecture of Frankl and Füredi

Fisher's Inequality bounds the size of  $\lambda$ -intersecting families, which are families of subsets of X whose members pairwise intersect in exactly  $\lambda$  points. Fisher's Inequality proves that if  $\mathscr{F}$  is  $\lambda$ -intersecting, then the number of points in X that are covered by a set in  $\mathscr{F}$  is at least  $|\mathscr{F}|$ . Inspired by Fisher's Inequality, Frankl and Füredi conjectured that if  $\mathscr{F}$  is a  $\lambda$ -intersecting family, for which no point in X lies in all the sets in  $\mathscr{F}$ , then the number of pairs of points in X that are covered by some set in  $\mathscr{F}$  is at least  $\binom{|\mathscr{F}|}{2}$ .

#### 1.2.2.1 Results

We reformulated the Frankl-Füredi conjecture in terms of shadows, and proved it in some special cases using linear programming [25]. We first show in Theorem 6.1.1 that Frankl and Füredi's conjecture holds for nontrivial  $\lambda$ -intersecting families that satisfy a reasonable extra condition and characterize the extremal families. If a conjecture of Hall [68] is true, Theorem 6.1.1 would verify the Frankl-Füredi conjecture for  $\lambda$ intersecting families whose members have fixed size k, when k is large with respect to  $\lambda$ . While Hall's conjecture remains open, we applied Theorem 6.1.1 in Theorem 6.1.2
to prove the Frankl-Füredi conjecture, when  $\mathscr{F}$  is additionally required to be uniform
and  $\lambda$  is small.

#### 1.2.2.2 Significance

The  $\lambda$ -intersecting family in the Frankl-Füredi conjecture is a central concept in coding theory so any new results about it may have applications in communication systems. A proof of the Frankl-Füredi conjecture would be interesting from a mathematical standpoint because it would yield a new proof of Fisher's Inequality. As we have seen, it also relates to many other conjectures in design theory such as Hall's.

#### 1.2.2.3 Future Work

Not much is known about the Frankl-Füredi conjecture, although it is twenty years old, and my paper is the first to consider it since it was published. I pose a conjecture about 2-intersecting families in my paper, and if this conjecture is true, it would imply the Frankl-Füredi conjecture when  $\lambda = 2$ . I intend to continue working on both of these conjectures.

### 1.2.3 On the Manickam-Miklós-Singhi Conjecture

We used combinatorial and algebraic methods for the previous two questions. The probabilistic method, however, also plays a role in extremal set theory problems and their analogs. Alon, Huang, and Sudakov's recent work [1] on the Manickam-Miklós-Singhi conjecture [88, 89] uses probabilistic arguments to prove partial analogs of the Erdős-Ko-Rado and Hilton-Milner theorems for subsums of a finite sum. More precisely, they showed that if  $n \ge 33k^2$ , then every set of *n* real numbers with nonnegative sum has at least  $\binom{n-1}{k-1}$  *k*-element subsets whose sum is also nonnegative. The conjecture is thought to hold when  $n \ge 4k$ , and Chapter 7 presents ideas for verifying it when *k* is small and for improving Alon, Huang, and Sudakov's result.

#### 1.2.3.1 Results

For  $k \in \mathbb{Z}^+$ , let f(k) be the minimum integer  $N \in \mathbb{Z}^+$  such that for all  $n \ge N$ , every set of *n* real numbers with nonnegative sum has at least  $\binom{n-1}{k-1}$  *k*-element subsets whose sum is also nonnegative. Manickam, Miklós, and Singhi [88, 89] proved that f(k)exists and conjectured that  $f(k) \le 4k$ . Alon, Huang, and Sudakov [1] recently showed that  $f(k) \le \min\{33k^2, 2k^3\}$ , which substantially improves previous bounds. We prove f(3) = 11 and  $f(4) \le 24$ , which improves the previous upper bounds of  $f(3) \le 12$  from [87, 91] and  $f(4) \le 128$  from [1]. Although Conjecture 7.1.1 for k = 3 was previously tackled, our result is stronger because we determine f(3) exactly and we characterize the case of equality; moreover our proof is simpler and gives a nice application of the Kruskal-Katona theorem. We also show how our method could potentially yield a quadratic upper bound on f(k) that improves on that of Alon, Huang, and Sudakov. We end Chapter 7 by discussing a related open problem, the vector space analog of the Manickam-Miklós-Singhi conjecture.

# **1.3** Notation and Terminology

Here, we collect a list of frequently used notation and terminology.

- Number Systems
  - 1.  $\mathbb{Z}$  denotes the integers.  $\mathbb{Z}^+$  denotes the positive integers.
  - 2. If  $a, b \in \mathbb{Z}$ , then a | b if there exists  $x \in \mathbb{Z}$  such that ax = b.
  - 3.  $\mathbb{N} = \{0, 1, ...\}$  denotes the natural numbers.
  - 4.  $\mathbb{Q}$  denotes the rationals.
  - 5.  $\mathbb{R}$  denotes the real numbers.  $\mathbb{R}^+$  denotes the positive real numbers.
  - 6.  $\mathbb{R}_{>0} := \{x \in \mathbb{R} : x \ge 0\}$
  - 7. If  $a, b \in \mathbb{R}$  and a < b, then  $[a, b] := \{x \in \mathbb{R} : a \le x \le b\}$ .
  - 8.  $\mathbb{F}_q$  denotes the finite field of order q.
  - 9. If A is a set, then  $A^d = A \times \cdots \times A$  denotes the *d*-fold product of A with itself.

- Sets.
  - 1.  $X := [n] = \{1, 2, \dots, n\}.$
  - 2. Ø denotes the *empty set*.
  - 3.  $S_X$  denotes the family of all permutations of the set X.
  - 4. |X| denotes the *cardinality* of *X*.
  - 5.  $x \in X$  denotes an *element* of *X*.
  - 6.  $X \cup Y$  denotes the *union* of *X* and *Y*.
  - 7.  $X \cap Y$  denotes the *intersection* of X and Y.
  - 8.  $X \setminus Y$  denotes the set complement of Y with respect to X.
- Families of Sets
  - 1.  $2^X$  is the family of all subsets of *X*.
  - 2.  $\mathscr{F} \subset 2^X$  denotes a family of sets. When using the notation  $\mathscr{F} = \{F_1, \ldots, F_m\}$ , the sets  $F_i$  are assumed to be distinct, unless otherwise stated.
  - 3.  $\binom{X}{k}$  denotes the family of all *k*-element subsets of *X*. We also call  $\binom{X}{k}$  the complete *k*-uniform hypergraph on *n* vertices.
  - 4.  $\binom{n}{k} = \frac{n(n-1)\cdots(n-k+1)}{k!}$  is the binomial coefficient. Note  $\binom{X}{k} = \binom{n}{k!}$ .
  - 5.  $\mathscr{C}_k^m$  denotes the first *m* sets of  $\binom{X}{k}$  in the colex order.
- Properties of Families of Sets
  - 1. A family  $\mathscr{F} \subset 2^X$  is *k*-uniform if  $\mathscr{F} \subset {X \choose k}$ .
  - 2. For  $\mathscr{F} \subset 2^X$  and  $x \in X$ , the *degree* of x is  $deg(x) := |\{F \in \mathscr{F} : x \in F\}|$ , the number of elements in  $\mathscr{F}$  that contain x.
  - 3. For  $\mathscr{F} \subset 2^X$  and  $S \subset X$ , the *co-degree* of *S* is codeg  $(S) := |\{F \in \mathscr{F} : S \subset F\}|$ , the number of sets in  $\mathscr{F}$  that contain *S*.
  - 4. A family  $\mathscr{F} \subset 2^X$  is *r*-regular if deg(x) = r for all  $x \in X$ .
  - 5. A family  $\mathscr{F} \subset 2^X$  is *trivial* if there exists  $x \in X$  with deg $(x) = |\mathscr{F}|$ , and is *nontrivial* otherwise.

- Intersection Properties of Families of Sets
  - 1. A family  $\mathscr{F} \subset 2^X$  is intersecting if  $F \cap F' \neq \emptyset$  for any  $F, F' \in \mathscr{F}$ .
  - 2. A family  $\mathscr{F} \subset 2^X$  is *r*-wise intersecting if, for all  $F_1, \ldots, F_r \in \mathscr{F}$ , we have  $\bigcap_{i=1}^r F_i \neq \emptyset$ .
  - 3. For  $t \in \mathbb{Z}^+$ , a family  $\mathscr{F} \subset 2^X$  is *t*-intersecting if  $|F \cap F'| \ge t$  when  $F, F' \in \mathscr{F}$ .
  - 4. For  $\lambda \in \mathbb{N}$ , a family  $\mathscr{F} \subset 2^X$  is  $\lambda$ -intersecting if  $|F_1 \cap F_2| = \lambda$  for any  $\{F_1, F_2\} \in \binom{\mathscr{F}}{2}$ .
  - 5. For a finite set  $L \subset \mathbb{N}$ , a family  $\mathscr{F} \subset 2^X$  is *L*-intersecting if  $|F_1 \cap F_2| \in L$  for any  $\{F_1, F_2\} \in \binom{\mathscr{F}}{2}$ .
- Operators on Families of Sets
  - 1. If  $\mathscr{F} \subset 2^X$  then  $\partial^i \mathscr{F} := \left\{ E \in \binom{X}{i} : E \subset F \in \mathscr{F} \right\}$  is the *i*-shadow of  $\mathscr{F}$ .
  - 2. If  $\mathscr{F} \subset 2^X$  and  $\{i, j\} \in {X \choose 2}$ , then  $\tilde{S}_{ij}(\mathscr{F})$  is the *shift operator* that replaces the element *j* by the element *i* whenever possible.
  - 3. A family  $\mathscr{F} \subset 2^X$  has an associated  $|\mathscr{F}| \times |X|$  incidence matrix *M*.

#### • Graphs

- 1. G = (V, E) denotes an undirected graph.
- 2. V(G) = V denotes the set of vertices in the context of a graph.
- 3.  $E \subset {\binom{V}{2}}$  denotes the family of *edges* of a graph.
- 4.  $v \sim w$  denotes vertex *adjacency*.
- 5. A graph G = (V, E) has an associated  $|V| \times |V|$  adjacency matrix A.
- 6.  $\chi(G)$  denotes the chromatic number of a graph.
- 7. If G = (V, E) is a graph, then the *degree* of a vertex  $v \in G$ , denoted deg(v), is the number of vertices adjacent to v; that is, deg $(v) := |\{w : v \sim w\}|$ .
- 8. A graph G = (V, E) is *k*-regular if all vertices  $v \in V$  have degree deg(v) = k.
- 9.  $K_{n:k}$  denotes the Kneser graph.
- 10.  $qK_{n:k}$  denotes the *q*-Kneser graph.

- Digraphs
  - 1. D = (V,A) denotes a digraph
  - 2.  $A \subset V \times V$  is a family of ordered pairs of vertices called *arcs*.
  - 3. If a = (v, w) is an arc in the digraph D = (V, A), then *a* is said to be directed from *v* to *w*. We call *v* the *tail* of *a* and *w* the *head*.
  - 4. A digraph D = (V, A) has an associated  $|V| \times |A|$  incidence matrix M.
- Vector Spaces
  - 1. V denotes an *n*-dimensional vector space over the finite field  $\mathbb{F}_q$  in the context of vector spaces.
  - 2.  $\{0\}$  denotes the zero subspace.
  - 3.  $\dim V$  denotes the dimension of the vector space V over its underlying field.
  - 4.  $\begin{bmatrix} V \\ k \end{bmatrix}$  denotes the family of all *k*-dimensional subspaces of *V*.
  - 5. For  $a \in \mathbb{R}$ ,  $q \in \mathbb{R}^+$ , and  $k \in \mathbb{Z}^+$ , the symbol  $\begin{bmatrix} a \\ k \end{bmatrix}_q := \prod_{0 \le i < k} \frac{q^{a-i}-1}{q^{k-i}-1}$  denotes the *q*-binomial coefficient. When k = 1, we write  $[a]_q := \begin{bmatrix} a \\ 1 \end{bmatrix}_q$ .
  - 6. If  $A \subset V$  is a subspace of V, let  $A^{\perp}$  denote the orthogonal complement of A.
  - If A, B ⊂ V are subspaces of V, then A ∨ B denotes the *linear span* of A and
     B. That is, A ∨ B is the smallest subspace that contains both A and B.
  - 8. GL(V) denotes the family of all nonsingular linear transformations of V.
- Intersection Properties of Families of Subspaces
  - 1. A family  $\mathscr{F} \subset \begin{bmatrix} V \\ k \end{bmatrix}$  is intersecting if dim $(F \cap F') \neq 0$  for any  $F, F' \in \mathscr{F}$ .
  - 2. A family  $\mathscr{F} \subset \begin{bmatrix} V \\ k \end{bmatrix}$  is called *r*-wise intersecting if dim $(\bigcap_{i=1}^r F_i) \neq 0$  whenever  $F_1, \ldots, F_r \in \mathscr{F}$ .
  - 3. For  $t \in \mathbb{Z}^+$ , a family  $\mathscr{F} \subset \begin{bmatrix} V \\ k \end{bmatrix}$  is *t*-intersecting if  $\dim(F \cap F') \ge t$  when  $F, F' \in \mathscr{F}$ .

# Chapter 2

# **Combinatorial Techniques**

Let *X* denote the set  $[n] := \{1, ..., n\}$ , which we take as our underlying set, and let  $2^X$  denote the family of all subsets of *X*. The general problem in extremal set theory asks for the maximum or minimum size of a family  $\mathscr{F} \subset 2^X$  of subsets of *X* that satisfies certain restrictions. It is common, for example, to restrict the family to be *k*-uniform; that is, all sets in the family have size *k*. We use the symbol  $\binom{X}{k}$  to denote the family of all *k*-subsets of *X*.

The sections of this chapter fall into three major themes. The first topic is shadows, and we shall discuss the Kruskal-Katona theorem [75, 80]. We next discuss intersection theorems such as the Erdős-Ko-Rado theorem [47] and its generalizations. Both of these topics allow us to demonstrate the power of the shifting technique. In this chapter, we stress the importance and usefulness of the shifting technique in extremal theory because we will see in Chapter 4 that no analog of it for vector spaces exists yet despite several attempts.

The last theme of this chapter is Fisher's inequality [17, 51, 73, 85]. We give de Bruijn and Erdős' proof [34] of the nonuniform Fisher inequality in the case  $\lambda = 1$ . Finally, we end the chapter with a proof of a special case of a conjecture of Frankl and Füredi [55], which generalizes de Bruijn and Erdős' result and proof, and makes use of convexity arguments. Our aim in this chapter is to give an overview of some combinatorial techniques that are frequently used in extremal set theory and in this thesis.

# 2.1 Shadows

Shadows are a fundamental concept in extremal set theory and appear throughout this thesis. For example, we will use shadows in the proofs of the Erdős-Ko-Rado theorem and Frankl's r-wise intersection theorem [52] in Section 2.2.4 and Section 2.2.6 respectively. We will generalize shadows to vector spaces in Chapter 5, study a conjecture of Frankl and Füredi on shadows in Chapter 6, and finally use them to prove special cases of the Manickam-Miklós-Singhi conjecture [88, 89] in Chapter 7.

In Section 2.1.1, we formally define shadows and state the main result on them, the Kruskal-Katona theorem, in its full generality. While we will need the complete Kruskal-Katona theorem in Chapter 7, in practice, it suffices to use a weaker but more convenient version of the theorem due to Lovász [84, Ex 13.31(b)]. In Section 2.1.2, we explain the statement of Lovász's version of the Kruskal-Katona theorem. We will give two proofs of Lovász's version of the Kruskal-Katona theorem in Section 2.1.5 and Section 2.1.6 respectively. The first proof is due to Frankl [53] and will make use of the shifting technique. The second is a recent and elegant proof due to Keevash [77], which we will generalize to vector spaces in Chapter 5.

## 2.1.1 The Kruskal-Katona Theorem

The Kruskal-Katona theorem [75, 80] gives a tight lower bound on the size of the shadow of a family  $\mathscr{F} \subset {X \choose k}$ .

**Definition 2.1.1.** For a family  $\mathscr{F} \subset 2^X$ , we define the <u>i-shadow</u> of  $\mathscr{F}$ , denoted  $\partial^i \mathscr{F}$ , to consist of those i-subsets of X contained in at least one member of  $\mathscr{F}$ ,

$$\partial^i \mathscr{F} := \left\{ E \in \binom{X}{i} : E \subset F \in \mathscr{F} \right\}.$$

When  $\mathscr{F} \subset \binom{X}{k}$ , we define the <u>shadow</u> of  $\mathscr{F}$ , denoted  $\partial \mathscr{F}$ , to be  $\partial \mathscr{F} = \partial^{k-1} \mathscr{F}$ .

The Kruskal-Katona theorem also describes the structure of set-systems with minimum shadow over all set-systems with the same cardinality. To characterize these families, we need to define the colex order on  $\binom{X}{k}$ .

**Definition 2.1.2.** Given  $A = \{a_1, ..., a_k\}$  and  $B = \{b_1, ..., b_k\}$  in  $\binom{X}{k}$  where  $a_i < a_{i+1}$ and  $b_i < b_{i+1}$  for all  $i \in [k-1]$ , we say that A < B in the <u>colex order</u> if  $A \neq B$ , and for  $s = \max\{t : a_t \neq b_t\}$ , we have  $a_s < b_s$ . For  $1 \le m \le \binom{n}{k}$ , let  $\mathscr{C}_m^k$  denote the first m elements of the colex order on  $\binom{X}{k}$ .

Note that the colex order is a total order on  $\binom{X}{k}$ , so the definition of  $\mathscr{C}_m^k$  makes sense.

We now prove a lemma that demonstrates that every positive integer  $m \in \mathbb{Z}^+$  has a *k*-binomial representation, and this lemma will allow us to compute the size of the shadow of the first *m* sets in the colex order on  $\binom{X}{k}$ .

**Lemma 2.1.3.** Given positive integers  $m, k \in \mathbb{Z}^+$ , there exists a unique representation of *m* in the form

$$m = \binom{a_k}{k} + \binom{a_{k-1}}{k-1} + \dots + \binom{a_t}{t},$$

where  $a_k > a_{k-1} > \cdots > a_t \ge t \ge 1$ . (This representation of *m* is called the *k*-binomial representation of *m*.)

*Proof.* We show by induction on *k* that such a representation exists and is unique. If k = 1, then the theorem is trivially true. Now assume that there exists a unique such representation for any  $m \in \mathbb{Z}^+$  and k = l - 1. To show that a unique such representation exists for  $m \in \mathbb{Z}^+$  and k = l, first note that t = k and  $a_t = k$  is a unique such representation for m = 1. Suppose then that  $m \ge 2$ . Since for  $r \ge k$ , we have

$$\sum_{j=1}^{k} \binom{r-k+j}{j} = \binom{r+1}{k} - 1 < \binom{r+1}{k},$$

we must have  $a_k = \max\{r : \binom{r}{k} \le m\}$ . If  $m = \binom{a_k}{k}$  then set t = k. If  $m > \binom{a_k}{k}$ , then by the induction hypothesis, we have unique positive integers  $a_{k-1} > \cdots > a_t \ge t \ge 1$  such that

$$m - \binom{a_k}{k} = \binom{a_{k-1}}{k} + \dots + \binom{a_t}{t}.$$

Hence  $m = \binom{a_k}{k} + \binom{a_{k-1}}{k} + \cdots + \binom{a_t}{t}$  is a unique such representation.

We are now ready to state the Kruskal-Katona theorem.

**Theorem 2.1.4** (Kruskal-Katona). Let  $\mathscr{F} \subset {\binom{X}{k}}$  be a family of size *m* and suppose that  $m = \sum_{j=t}^{k} {a_j \choose j}$  is the *k*-binomial representation of *m*. Then

$$|\partial \mathscr{F}| \ge |\partial \mathscr{C}_m^k| = \sum_{j=t}^k \binom{a_j}{j-1};$$

in other words, the size of the shadow of  $\mathscr{F}$  is at least the size of the shadow of the first m sets in the colex order on  $\binom{X}{k}$ . Moreover, letting  $m' := \sum_{j=t}^{k} \binom{a_j}{j-1}$ , we have  $\partial \mathscr{C}_m^k = \mathscr{C}_{m'}^{k-1}$ ; that is, the shadow of the first m sets in the colex order on  $\binom{X}{k}$  consists of the first m' sets in the colex order on  $\binom{X}{k-1}$ .

## 2.1.2 Lovász's Version of the Kruskal-Katona Theorem

While we will need the Kruskal-Katona theorem in its full generality in Chapter 7, Lovász's weaker but more convenient version [84, Ex 13.31(b)] suffices for many applications. For example, we will use Lovász's version in the proofs of the Erdős-Ko-Rado theorem and Frankl's *r*-wise intersection theorem in Section 2.2.4 and Section 2.2.6 respectively. We also generalize Lovász's theorem to vector spaces in Chapter 5. In this section, we explain the statement of Lovász's result.

Recall that the binomial coefficient

$$\binom{n}{k} := \frac{n(n-1)\cdots(n-k+1)}{k!}$$

can be defined for all  $n \in \mathbb{R}$  and  $k \in \mathbb{Z}^+$ . If we fix k, then  $\binom{n}{k}$  is a continuous function of n that is positive and increasing; hence, by the intermediate value theorem, if  $r \ge 1$  is a real number, then there exists a unique real number  $n_r \ge k$  such that  $r = \binom{n_r}{k}$ .

**Theorem 2.1.5** (Lovász). Let  $\mathscr{F} \subset {X \choose k}$  and let  $y \ge k$  be the real number defined by  $|\mathscr{F}| = {y \choose k}$ . Then  $|\partial \mathscr{F}| \ge {y \choose k-1}$ . If equality holds, then  $y \in \mathbb{Z}^+$  and  $\mathscr{F} = {Y \choose k}$ , where Y is a y-subset of X.

### 2.1.3 Shifting

Our first proof of Lovász's version of the Kruskal-Katona theorem will make use of the shifting technique. Also known as compression, the method was introduced by Erdős, Ko, and Rado [47] and is one of the most important tools in extremal set theory. We will use the shifting technique again in our proof of the Erdős-Ko-Rado theorem in Section 2.2.3.

**Definition 2.1.6.** *For*  $F \subset X$  *and distinct*  $i, j \in X$  *define* 

$$S_{ij}(F) := \begin{cases} (F \setminus \{j\}) \cup \{i\} & \text{if } i \notin F, \ j \in F \\ F & \text{otherwise.} \end{cases}$$

We see that  $S_{ij}$  replaces the element j by the element i whenever possible. Note that  $S_{ij}$  is not a one-to-one map since for  $E \subset X \setminus \{i, j\}$  one has

$$S_{ij}(E \cup \{i\}) = S_{ij}(E \cup \{j\}) = E \cup \{i\}.$$

We would like to associate with  $S_{ij}$  a map  $\tilde{S}_{ij}$ , which sends a set system to another set system of the same size. This motivates the following definition.

**Definition 2.1.7.** For a family  $\mathscr{F} \subset 2^X$  and distinct  $i, j \in X$  define the shift operator

$$\tilde{S}_{ij}(\mathscr{F}) := \{S_{ij}(F) : F \in \mathscr{F}\} \cup \{F : F, S_{ij}(F) \in \mathscr{F}\}.$$

We note two immediate properties of  $S_{ij}$  and its associated shift operator  $\tilde{S}_{ij}$ .

**Proposition 2.1.8.** We have  $|F| = |S_{ij}(F)|$  and  $|\mathscr{F}| = |\tilde{S}_{ij}(\mathscr{F})|$ .

We will see in Section 2.1.1 and Section 2.2.1 that the shift operator  $\tilde{S}_{ij}$  allows us to transform our original set system into a more structured one while still preserving important properties of the original.

### 2.1.4 **Properties of the Shift Operator**

Towards a proof of Lovász's version of the Kruskal-Katona theorem via shifting, we first observe that the shifting operator  $\tilde{S}_{ij}$  does not increase the size of the shadow.

**Lemma 2.1.9.** For  $\mathscr{F} \subset {\binom{X}{k}}$  and distinct  $i, j \in X$ , we have  $|\partial \mathscr{F}| \geq |\partial \tilde{S}_{ij}(\mathscr{F})|$ .

*Proof.* We show that  $S_{ji}: \partial \tilde{S}_{ij}(\mathscr{F}) \setminus \partial \mathscr{F} \to \partial \mathscr{F} \setminus \partial \tilde{S}_{ij}(\mathscr{F})$  is an injective map. Observe that if  $E \in \partial \tilde{S}_{ij}(\mathscr{F}) \setminus \partial \mathscr{F}$  then  $E \subset S_{ij}(F)$  for some  $F \in \mathscr{F}$  for which  $S_{ij}(F) \neq F$ . Hence

 $j \notin E$ . If  $i \notin E$ , then  $E = S_{ij}(F) \setminus \{i\} = F \setminus \{j\}$  so  $E \in \partial \mathscr{F}$ , which is impossible by our assumption; hence  $i \in E$ . We have shown that every  $E \in \partial \tilde{S}_{ij}(\mathscr{F}) \setminus \partial \mathscr{F}$  has  $i \in E$  and  $j \notin E$ .

We claim that  $S_{ji}(E) \in \partial \mathscr{F} \setminus \partial \tilde{S}_{ij}(\mathscr{F})$ . Clearly,  $S_{ji}(E) \in \partial \mathscr{F}$ . Suppose, for a contradiction, that  $(E \setminus \{i\}) \cup \{j\} = S_{ji}(E) \in \partial \tilde{S}_{ij}(\mathscr{F})$ . Hence, for some  $x \in X$  that is not in  $(E \setminus \{i\}) \cup \{j\}$ , we have  $(E \setminus \{i\}) \cup \{j, x\} \in \tilde{S}_{ij}(\mathscr{F})$ . If x = i, then  $E \cup \{j\} \in \tilde{S}_{ij}(\mathscr{F})$  and Definition 2.1.7 implies that  $E \cup \{j\} \in \mathscr{F}$  so  $E \in \partial \mathscr{F}$ , which is a contradiction. If  $x \neq i$ , then since  $i \notin (E \setminus \{i\}) \cup \{j, x\}$  and  $j \in (E \setminus \{i\}) \cup \{j, x\} \in \tilde{S}_{ij}(\mathscr{F})$ , we must have

$$E \cup \{x\} = S_{ij}((E \setminus \{i\}) \cup \{j, x\}) \in \mathscr{F},$$

so  $E \in \partial \mathscr{F}$ , which is a contradiction. This proves that  $S_{ji}(E) \in \partial \mathscr{F} \setminus \partial \tilde{S}_{ij}(\mathscr{F})$ .

We showed in the first paragraph that every  $E \in \partial \tilde{S}_{ij}(\mathscr{F}) \setminus \partial \mathscr{F}$  has  $i \in E$  and  $j \notin E$ . Hence, the map  $S_{ji} : \partial \tilde{S}_{ij}(\mathscr{F}) \setminus \mathscr{F} \to \partial \mathscr{F} \setminus \partial \tilde{S}_{ij}(\mathscr{F})$  is injective. Consequently,

$$\begin{split} |\partial \mathscr{F}| &= |\mathscr{F} \cap \partial \widetilde{S}_{ij}(\mathscr{F})| + |\partial \mathscr{F} \setminus \partial \widetilde{S}_{ij}(\mathscr{F})| \\ &\geq |\partial \widetilde{S}_{ij}(\mathscr{F}) \cap \mathscr{F}| + |\partial \widetilde{S}_{ij}(\mathscr{F}) \setminus \mathscr{F}| = |\partial \widetilde{S}_{ij}(\mathscr{F})|, \end{split}$$

as desired.

Since shifting does not increase the size of a family's shadow, the next two lemmas show how we can shift our original family to one that is more structured.

**Lemma 2.1.10.** Let  $\mathscr{F} \subset {X \choose k}$ . Define  $\mathscr{H}_1 = \mathscr{F}$  and  $\mathscr{H}_i = \tilde{S}_{1i}(\mathscr{H}_{i-1})$  for  $i \in X \setminus \{1\}$ . We then have  $\tilde{S}_{1i}(\mathscr{H}_n) = \mathscr{H}_n$  for all  $i \in X \setminus \{1\}$ .

*Proof.* Suppose, for a contradiction, that there exists  $i \in X \setminus \{1\}$  such that  $\tilde{S}_{1i}(\mathscr{H}_n) \neq \mathscr{H}_n$ . Hence, there exists  $H \in \mathscr{H}_n$  such that  $i \in H$ ,  $1 \notin H$ , and  $S_{1i}(H) \notin \mathscr{H}_n$ . As  $1 \notin H$ , we have  $H \in \mathscr{H}_{i-1} \cap \mathscr{H}_i$ . This implies  $S_{1i}(H) \in \mathscr{H}_i$  and hence  $S_{1i}(H) \in \mathscr{H}_n$ , which is a contradiction. Consequently,  $\tilde{S}_{1i}(\mathscr{H}_n) = \mathscr{H}_n$  for all  $i \in X \setminus \{1\}$ .

If  $\mathscr{F} \subset {X \choose k}$ , then we can partition  $\mathscr{F}$  into two subfamilies according to whether  $1 \in F \in \mathscr{F}$ . We show that this partition has a special property, when  $\mathscr{F} \subset {X \choose k}$  is a family that satisfies the conclusion of Lemma 2.1.10, namely that  $\tilde{S}_{1i}(\mathscr{F}) = \mathscr{F}$  for each  $i \in X \setminus \{1\}$ . We will exploit this property when we inductively prove Lovász's version of the Kruskal-Katona theorem.

**Lemma 2.1.11.** Suppose  $\mathscr{F} \subset {X \choose k}$  and  $\tilde{S}_{1i}(\mathscr{F}) = \mathscr{F}$  for each  $i \in X \setminus \{1\}$ . Let

 $\mathscr{F}_0 := \{ F \in \mathscr{F} : 1 \notin F \}, \ \mathscr{F}_1 := \{ F \setminus \{1\} : 1 \in F \in \mathscr{F} \}.$ 

We then have  $|\mathscr{F}_1| \geq |\partial \mathscr{F}_0|$ .

*Proof.* If  $E \in \partial \mathscr{F}_0$  then  $E \cup \{i\} \in \mathscr{F}$  for some  $i \in X \setminus \{1\}$ . Since  $\tilde{S}_{1i}(\mathscr{F}) = \mathscr{F}$ , we must have  $S_{1i}(E \cup \{i\}) = E \cup \{1\} \in \mathscr{F}$ . Consequently,  $E \in \mathscr{F}_1$ . This proves  $|\mathscr{F}_1| \ge |\partial \mathscr{F}_0|$ .  $\Box$ 

## 2.1.5 Lovász's Result via Shifting

We now present Frankl's shifting proof of Lovász's version of the Kruskal-Katona theorem [53].

**Proof of Theorem 2.1.5.** Let  $\mathscr{F} \subset {\binom{X}{k}}$  and let  $y \ge k$  be the real number that is defined by  $|\mathscr{F}| = {\binom{y}{k}}$ . Define  $\mathscr{H}_1 = \mathscr{F}$  and  $\mathscr{H}_i = \widetilde{S}_{1i}(\mathscr{H}_{i-1})$  for  $i \in X \setminus \{1\}$ ; for notational convenience, let  $\mathscr{H}^* := \mathscr{H}_n$ . We have  $|\mathscr{H}^*| = |\mathscr{F}| = {\binom{y}{k}}$  and  $|\partial \mathscr{F}| \ge |\partial \mathscr{H}^*|$  by Proposition 2.1.8 and Lemma 2.1.9. Hence, the theorem will be proved for  $\mathscr{F}$  if we can prove it for  $\mathscr{H}^*$ .

We use double induction on *k* and  $m = |\mathscr{F}| = |\mathscr{H}^*|$ . Define

$$\mathscr{H}_0^* := \{ H \in \mathscr{H}^* : 1 \notin H \}, \ \mathscr{H}_1^* := \{ H \setminus \{1\} : 1 \in H \in \mathscr{H}^* \}.$$

Suppose, for a contradiction, that  $|\mathscr{H}_1^*| < {y-1 \choose k-1}$ . We then have

$$|\mathscr{H}_0^*| = |\mathscr{H}^*| - |\mathscr{H}_1^*| > \binom{y}{k} - \binom{y-1}{k-1} = \binom{y-1}{k}.$$

Hence, by the induction hypothesis,  $|\partial \mathscr{H}_0^*| \ge {\binom{y-1}{k-1}}$ . Lemma 2.1.10 and Lemma 2.1.11 then imply that  $|\mathscr{H}_1^*| \ge |\partial \mathscr{H}_0^*| \ge {\binom{y-1}{k-1}}$ , which contradicts our original assumption. Therefore, we must have  $|\mathscr{H}_1^*| \ge {\binom{y-1}{k-1}}$ , and so  $|\partial \mathscr{H}_1^*| \ge {\binom{y-1}{k-2}}$  by the induction hypothesis. As  $\partial \mathscr{H}_0^* \subset \mathscr{H}_1^*$  by Lemma 2.1.11, we obtain the desired conclusion that

$$|\partial \mathscr{F}| \ge |\partial \mathscr{H}^*| = |\mathscr{H}_1^*| + |\partial \mathscr{H}_1^*| \ge {\binom{y-1}{k-1}} + {\binom{y-1}{k-2}} = {\binom{y}{k-1}}.$$
 (2.1.1)

We now characterize the case of equality in (2.1.1). If  $|\partial \mathscr{F}| = {y \choose k-1}$ , then equality holds everywhere in (2.1.1). We then have that  $|\partial \mathscr{H}_1^*| = {y-1 \choose k-1}$ , and hence

 $|\mathscr{H}_0^*| = {\binom{y-1}{k}}$ . The induction hypothesis and Lemma 2.1.11 imply that  $|\partial \mathscr{H}_0^*| = {\binom{y-1}{k-1}}$ and  $\partial \mathscr{H}_0^* = \mathscr{H}_1^*$ . Since  $|\mathscr{H}_0^*| = {\binom{y-1}{k}}$  and  $|\partial \mathscr{H}_0^*| = {\binom{y-1}{k-1}}$ , the induction hypothesis implies that  $\mathscr{H}_0^* = {\binom{Z}{k}}$  where |Z| = y - 1. Hence  $\mathscr{H}^* = {\binom{Y'}{k}}$ , where  $Y' = Z \cup \{1\}$ , as  $\mathscr{H}^* = \mathscr{H}_0^* \cup \{H \cup \{1\} : H \in \mathscr{H}_1^*\}$  and  $\partial \mathscr{H}_0^* = \mathscr{H}_1^*$ .

If  $\mathscr{F} = \mathscr{H}^*$ , then setting Y = Y', we are done since |Y| = |Y'| = y. If  $\mathscr{F} \neq \mathscr{H}^*$ , let  $i = \min\{j : \mathscr{H}_j = \mathscr{H}^*\}$ . Of course  $i \ge 2$ , since  $\mathscr{H}_1 = \mathscr{F}$  and we have assumed that  $\mathscr{F} \neq \mathscr{H}^*$ . When we apply the shift operator  $\tilde{S}_{1i}$  to  $\mathscr{H}_{i-1}$  only the element *i* can be deleted; we consequently have  $\partial^1 \mathscr{H}_{i-1} \subset Z \cup \{i\}$  because  $\mathscr{H}_i = \mathscr{H}^* = \binom{Z \cup \{1\}}{k}$ . Hence,  $|\partial^1 \mathscr{H}_{i-1}| \le |Z \cup \{i\}| = y$ . We have that  $|\partial \mathscr{F}| = \binom{y}{k-1} = |\partial \mathscr{H}^*|$ , so Lemma 2.1.9 implies that  $|\partial \mathscr{H}_{i-1}| = \binom{y}{k-1}$ . By (2.1.1), we have  $|\partial^1 \mathscr{H}_{i-1}| \ge y$ , so  $|\partial^1 \mathscr{H}_{i-1}| = |Z \cup \{i\}| = y$ and  $\partial^1 \mathscr{H}_{i-1} = Z \cup \{i\}$ . Consequently,  $\mathscr{H}_{i-1} = \binom{Z \cup \{i\}}{k}$ . Since  $1 \notin Z \cup \{i\}$ , Definition 2.1.7 implies that  $\mathscr{F} = \binom{Z \cup \{i\}}{k}$  so setting  $Y = Z \cup \{i\}$  finishes the proof.

# 2.1.6 Keevash's Proof of Lovász's Result

As we mentioned previously, no analog of the shifting technique for vector spaces exists yet. Consequently, it is not clear whether Frankl's proof in Section 2.1.5 can be generalized to vector spaces. We will give more examples of problems that arise when trying to generalize purely combinatorial proofs about sets to vector spaces in Chapter 4. Since the techniques that tend to work for both sets and vector spaces are typically algebraic in nature, we were pleasantly surprised to find that Keevash's purely combinatorial proof of Lovász's result [77] generalizes to vector spaces. We present Keevash's proof now, and discuss its generalization in Chapter 5.

We first collect definitions that will be used in Keevash's proof of Theorem 2.1.5.

**Definition 2.1.12.** *For*  $\mathscr{F} \subset {\binom{X}{k}}$  *and*  $x \in X$ *, define* 

$$K_{k+1}^k(\mathscr{F}) := \left\{ T \in \binom{X}{k+1} : \binom{T}{k} \subset \mathscr{F} \right\}$$

to be the family of (k+1)-subsets of X all of whose k-subsets lie in  $\mathscr{F}$  and

$$K_{k+1}^k(x) := \{T \in K_{k+1}^k(\mathscr{F}) : x \in T\}$$

to be the family of (k+1)-subsets in  $K_{k+1}^k(\mathscr{F})$  that contain x.

**Definition 2.1.13.** For  $\mathscr{F} \subset 2^X$  and  $x \in X$ , define the <u>degree</u> of x, which is denoted by deg(x), to be the number of elements of  $\mathscr{F}$  that contain x,

$$\deg(x) := |\{F \in \mathscr{F} : x \in F\}|.$$

**Definition 2.1.14.** For  $\mathscr{F} \subset {X \choose k}$  and  $x \in X$ , define the <u>link</u> of x to be the family of (k-1)-subsets in  $X \setminus \{x\}$  whose union with x is an element of  $\mathscr{F}$ ,

$$L(x) := \{A \subset X \setminus \{x\} : |A| = k - 1, A \cup \{x\} \in \mathscr{F}\} \subset \begin{bmatrix} X \\ k - 1 \end{bmatrix}.$$

In Theorem 2.1.15, we establish an upper bound on  $|K_{k+1}^k(\mathscr{F})|$  in terms of  $|\mathscr{F}|$ ; we will see that Theorem 2.1.5 follows as a simple corollary.

**Theorem 2.1.15** (Keevash). Let  $\mathscr{F} \subset {X \choose k}$  and let  $y \ge k$  be the real number defined by  $|\mathscr{F}| = {y \choose k}$ . Then

$$|K_{k+1}^k(\mathscr{F})| \le \binom{y}{k+1}.$$

Equality holds if and only if  $y \in \mathbb{Z}^+$  and  $\mathscr{F} = \binom{Y}{k}$  for some y-subset  $Y \subset X$ .

*Proof.* We argue by induction on k. The base case k = 1 is easy, so assume the statement is true for k - 1. We first show that if  $x \in X$ , then  $|K_{k+1}^k(x)| \le ((y-k)/k) \deg(x)$ ; we will then sum this inequality over all  $x \in X$  and double count to obtain the desired upper bound on  $|K_{k+1}^k(\mathscr{F})|$ . If deg(x) = 0, then clearly  $|K_{k+1}^k(x)| \le ((y-k)/k) \deg(x)$ , so we will assume that deg $(x) \ne 0$ . We will need to consider the cases where deg(x) is large and where deg(x) is small separately.

First, let's consider the case when  $\deg(x) \ge {\binom{y-1}{k-1}}$ . If  $F \cup \{x\} \in K_{k+1}^k(x)$ , then  $F \in \mathscr{F}$  does not contain *x*; consequently,

$$K_{k+1}^{k}(x)| \leq |\mathscr{F}| - \deg(x) \leq {\binom{y}{k}} - {\binom{y-1}{k-1}} = {\binom{y-1}{k}}$$
$$= \frac{y-k}{k} {\binom{y-1}{k-1}} \leq \frac{y-k}{k} \deg(x).$$
(2.1.2)

We have equality in (2.1.2) if and only if  $\deg(x) = {\binom{y-1}{k-1}}$  since  $\deg(x) \neq 0$ .

Now we'll consider the case where  $\deg(x) \leq {\binom{y-1}{k-1}}$ . If  $F \cup \{v\} \in K_{k+1}^k(x)$ , then  $F \in K_k^{k-1}(L(x))$  and so  $|K_{k+1}^k(x)| \leq |K_{k-1}^k(L(x))|$ . We define the real number  $y_x \geq k$ 

by  $\deg(x) = {\binom{y_x-1}{k-1}}$ . Since  $L(x) \subset {\binom{X}{k-1}}$  and  $|L(x)| = \deg(x) = {\binom{y_y-1}{k-1}}$ , the induction hypothesis yields that

$$|K_{k+1}^k(x)| \le |K_{k-1}^k(L(x))| \le \binom{y_x - 1}{k} = \frac{y_x - k}{k} \binom{y_x - 1}{k - 1} \le \frac{y - k}{k} \deg(x), \quad (2.1.3)$$

where the last inequality follows because  $y_x \le y$ , by our assumption on deg(*x*). We have equality in (2.1.3) if and only if deg(*x*) =  $\binom{y-1}{k-1}$  by our assumption that deg(*x*)  $\ne 0$ .

To finish the proof, we sum the inequality  $|K_{k+1}^k(x)| \le ((y-k)/k) \deg(x)$  over all  $x \in X$  and double-count to obtain the desired inequality on  $|K_{k+1}^k(\mathscr{F})|$ . We have

$$(k+1)|K_{k+1}^{k}(\mathscr{F})| = \sum_{x \in X} |K_{k+1}^{k}(x)| \le \frac{y-k}{k} \sum_{x \in X} \deg(x) = \frac{y-k}{k} \sum_{F \in \mathscr{F}} |F| \qquad (2.1.4)$$
$$= \frac{y-k}{k} (k|\mathscr{F}|) = (y-k) {\binom{y}{k}} = (k+1) {\binom{y}{k+1}}.$$

We now characterize the case of equality in (2.1.4). We see that equality holds in (2.1.4) if and only if  $|K_{k+1}^k(x)| = \frac{y-k}{k} \deg(x)$  for all  $x \in X$ . We saw previously that if  $\deg(x) \neq 0$ , then equality holds in (2.1.2) and (2.1.3) if and only if  $\deg(x) = {y-1 \choose k-1}$ . Consequently,

$$|\partial^{1}\mathscr{F}|\binom{y-1}{k-1} = \sum_{x \in X} \deg(x) = \sum_{F \in \mathscr{F}} |F| = k\binom{y}{k}$$

so  $|\partial^1 \mathscr{F}| = y$ . As  $\binom{y}{k} = |\mathscr{F}| \le \binom{|\partial^1 \mathscr{F}|}{k} = \binom{y}{k}$ , we have  $\mathscr{F} = \binom{Y}{k}$ , where  $Y = \partial^1 \mathscr{F}$ .  $\Box$ 

We now show that Theorem 2.1.5 follows as a corollary of Theorem 2.1.15.

**Keevash's Proof of Theorem 2.1.5** Let  $\mathscr{F}$  be as in Theorem 2.1.5, and let  $x \ge k-1$  be the real number defined by  $|\partial \mathscr{F}| = \binom{x}{k-1}$ . By Theorem 2.1.15,

$$\binom{y}{k} = |\mathscr{F}| \le |K_k^{k-1}(\partial \mathscr{F})| \le \binom{x}{k}$$

because  $\mathscr{F} \subset K_k^{k-1}(\partial \mathscr{F})$ . Hence  $x \ge y$ , so  $|\partial \mathscr{F}| = \binom{x}{k-1} \ge \binom{y}{k-1}$ . If  $|\partial \mathscr{F}| = \binom{y}{k-1}$  then x = y. Hence,  $|K_k^{k-1}(\partial \mathscr{F})| = \binom{y}{k}$  and  $\mathscr{F} = K_k^{k-1}(\partial \mathscr{F})$ . By Theorem 2.1.15, this implies  $y \in \mathbb{Z}^+$  and  $\partial \mathscr{F} = \binom{Y}{k-1}$  for some y-subset  $Y \subset X$ . Clearly,  $\binom{Y}{k} = K_k^{k-1}(\partial \mathscr{F}) = \mathscr{F}$ .  $\Box$ 

# **2.2 Intersection Theorems**

We introduce the Erdős-Ko-Rado theorem in Section 2.2.1 and its extension to *t*-intersecting families [116] in Section 2.2.8. We will give three proofs of the Erdős-Ko-Rado theorem in Section 2.2.3, Section 2.2.4, and Section 2.2.5. We also discuss generalizations of the Erdős-Ko-Rado theorem such as Frankl's *r*-wise intersection theorem [52] in Section 2.2.6 and the Hilton-Milner theorem [70] in Section 2.2.9. In Chapter 4 and Chapter 5, we will see vector space analogs of all these theorems.

# 2.2.1 The Erdős-Ko-Rado Theorem

We present three proofs of the Erdős-Ko-Rado theorem. The first is the original proof by Erdős, Ko, and Rado [47], which introduced the shifting technique. The second is Daykin's elegant proof [33], which shows that the Erdős-Ko-Rado theorem is a simple corollary of Lovász's version of the Kruskal-Katona theorem. The last is a proof by Katona [76], which uses the cyclic permutation method. In Chapter 4 and Chapter 7, we will use arguments similar to Katona's cyclic permutation method to prove special cases of the Erdős-Ko-Rado theorem for vector spaces and the Manickam-Miklós-Singhi conjecture respectively.

**Definition 2.2.1.** We say a family  $\mathscr{F} \subset 2^X$  is intersecting, if any two sets in  $\mathscr{F}$  have nonempty intersection; that is, for all  $F, F' \in \mathscr{F}$  we have  $F \cap F' \neq \emptyset$ .

The Erdős-Ko-Rado problem asks for the maximum size of an intersecting family  $\mathscr{F} \subset {X \choose k}$  of *k*-subsets of *X*. Of course, the problem is interesting only when  $n \ge 2k$ , as otherwise any two sets of  ${X \choose k}$  intersect.

**Theorem 2.2.2** (Erdős-Ko-Rado). Suppose  $\mathscr{F} \subset {\binom{X}{k}}$  is intersecting and  $n \ge 2k$ . Then  $|\mathscr{F}| \le {\binom{n-1}{k-1}}$ . Equality holds if and only if  $\mathscr{F} = \left\{F \in {\binom{X}{k}} : x \in F\right\}$  for some  $x \in X$ , excepting the case n = 2k.

## 2.2.2 More Properties of the Shift Operator

Towards a proof of the Erdős-Ko-Rado theorem using shifting, we first observe that the shifting operator  $\tilde{S}_{ij}$  preserves the intersecting property of set systems.

# **Proposition 2.2.3.** If $\mathscr{F} \subset 2^X$ is intersecting, then so is its shift $\tilde{S}_{ij}(\mathscr{F})$ .

Proof. Suppose, for a contradiction, that  $\mathscr{F} \subset 2^X$  is intersecting but its shift  $\tilde{S}_{ij}(\mathscr{F})$ is not. Let  $F_1, F_2 \in \tilde{S}_{ij}(\mathscr{F})$  be disjoint sets. We cannot have both  $F_1, F_2 \in \mathscr{F}$  since  $\mathscr{F}$  is intersecting and this would contradict our assumption that  $F_1$  and  $F_2$  are disjoint. Observe that Definition 2.1.7 implies that if  $F \in \tilde{S}_{ij}(\mathscr{F})$  and  $F \notin \mathscr{F}$  then  $i \in F$  and there exists  $E \in \mathscr{F}$  such that  $j \in E$ ,  $i \notin E$ , and  $F = S_{ij}(E)$ . Hence, we cannot have both  $F_1, F_2 \notin \mathscr{F}$  as then  $i \in F_1 \cap F_2$ . Without loss of generality, assume  $F_1 \notin \mathscr{F}$  and  $F_2 \in \mathscr{F}$ . Let  $E_1 \in \mathscr{F}$  such that  $j \in E_1$ ,  $i \notin E_1$ , and  $F_1 = S_{ij}(E_1)$ . We have  $E_1 \cap F_2 \neq \emptyset$  as  $\mathscr{F}$ is intersecting and  $E_1, F_2 \in \mathscr{F}$ . Since  $F_1 = S_{ij}(E_1) = (E_1 \setminus \{j\}) \cup \{i\}$  and  $F_1 \cap F_2 = \emptyset$ , we must have  $E_1 \cap F_2 = \{j\}$ . Now  $F_1 \notin \mathscr{F}$  implies that  $i \in F_1$ ; consequently  $i \notin F_2$  as  $F_1 \cap F_2 = \emptyset$ . We have shown that  $i \notin F_2$  and  $j \in F_2$  so  $S_{ij}(F_2) \neq F_2$ . Definition 2.1.7 implies that  $S_{ij}(F_2) \in \mathscr{F}$  as  $F_2 \in \tilde{S}_{ij}(\mathscr{F}) \cap \mathscr{F}$ . Since  $i \notin E_1$  and  $E_1 \cap F_2 = \{j\}$ , we have

$$E_1 \cap S_{ij}(F_2) = E_1 \cap \left( (F_2 \setminus \{j\}) \cup \{i\} \right) = (E_1 \cap F_2) \setminus \{j\} = \emptyset,$$

which contradicts that  $\mathscr{F}$  is intersecting. Hence, if  $\mathscr{F}$  is intersecting then so is  $\tilde{S}_{ii}(\mathscr{F})$ .  $\Box$ 

We now prove a lemma that will enable us to inductively prove the Erdős-Ko-Rado theorem via shifting.

**Lemma 2.2.4.** Suppose  $\mathscr{F} \subset {X \choose k}$  is an intersecting family and  $n \ge 2k$ . Define  $\mathscr{H}_0 = \mathscr{F}$ and  $\mathscr{H}_i = \tilde{S}_{in}(\mathscr{H}_{i-1})$  for  $1 \le i \le n-1$ . For all  $H, H' \in \mathscr{H}_{n-1}$ , we must have

$$H \cap H' \cap [n-1] \neq \emptyset.$$

*Proof.* Suppose, for a contradiction, that there exist  $H, H' \in \mathscr{H}_{n-1}$  such that

$$H \cap H' \cap [n-1] = \emptyset.$$

We have  $\mathscr{H}_{n-1}$  is intersecting by Proposition 2.2.3 and so  $H \cap H' = \{n\}$ . As n > 2kand  $|H \cup H'| = 2k - 1$ , there exists  $i \in [n - 1]$  such that  $i \notin H \cup H'$ . Since  $n \in H$ , we have  $H \in \mathscr{F}$ , and since  $H \in \mathscr{H}_{n-1}$ , we have  $H \in \mathscr{H}_{i-1} \cap \mathscr{H}_i$ . Since  $i \notin H$ ,  $n \in H$ , and  $H \in \mathscr{H}_i = \tilde{S}_{in}(\mathscr{H}_{i-1})$ , Definition 2.1.7 implies that  $\hat{H} := (H \setminus \{n\}) \cup \{i\} \in \mathscr{H}_{i-1}$  and hence  $\hat{H} \in \mathscr{H}_{n-1}$ . As  $H \cap H' = \{n\}$  and  $i \notin H'$ , we have  $\hat{H} \cap H' = \emptyset$  which contradicts  $\mathscr{H}_{n-1}$  being intersecting. This proves that  $H \cap H' \cap [n-1] \neq \emptyset$  for all H, H' in  $\mathscr{H}_{n-1}$ .  $\Box$ 

#### 2.2.3 Erdős-Ko-Rado via Shifting

We now prove the Erdős-Ko-Rado theorem via the shifting technique [47]. We will see that Proposition 2.2.3 and Lemma 2.2.4 allow us to transform our original intersecting family into a more structured intersecting family of the same size.

**Proof of Theorem 2.2.2.** Let  $\mathscr{F} \subset {\binom{X}{k}}$  be an intersecting family. We will apply induction on *n* and prove the theorem simultaneously for all  $k \le n/2$ . When n = 2k, each *k*-set can be paired up with its complement, which is also a *k*-set. Since any intersecting family  $\mathscr{F} \subset {\binom{X}{k}}$  can contain at most one *k*-set from each pair, we have

$$|\mathscr{F}| \le \frac{1}{2} \binom{2k}{k} = \binom{2k-1}{k-1}.$$
(2.2.1)

This argument also shows that there are exponentially many ways for equality to hold in (2.2.1) as any  $\mathscr{F} \subset {X \choose k}$  that chooses exactly one *k*-set from each pair is an extremal intersecting family.

For n > 2k, define  $\mathscr{H}_0 = \mathscr{F}$  and  $\mathscr{H}_i = \widetilde{S}_{in}(\mathscr{H}_{i-1})$  for  $1 \le i \le n-1$ . By Proposition 2.1.8 and Proposition 2.2.3, we have  $|\mathscr{H}_{n-1}| = |\mathscr{F}|$  and  $\mathscr{H}_{n-1}$  is intersecting. Define families  $\mathscr{G}_{k-1} \subset {\binom{[n-1]}{k-1}}$  and  $\mathscr{G}_k \subset {\binom{[n-1]}{k}}$  as follows:

$$\begin{aligned} \mathscr{G}_{k-1} &= \{ H \cap [n-1] : H \in \mathscr{H}_{n-1}, |H \cap [n-1]| = k-1 \}, \\ \mathscr{G}_k &= \{ H \cap [n-1] : H \in \mathscr{H}_{n-1}, |H \cap [n-1]| = k \}. \end{aligned}$$

Lemma 2.2.4 implies that both  $\mathcal{G}_{k-1}$  and  $\mathcal{G}_k$  are intersecting families. By the inductive hypothesis,

$$|\mathscr{G}_{k-1}| \leq \binom{n-2}{k-2}, \ |\mathscr{G}_k| \leq \binom{n-2}{k-1}.$$

Observe that for fixed  $G \in G_{k-1} \cup G_k$ , there is exactly one *k*-set  $H \in \mathscr{H}_{n-1}$  such that  $H \cap [n-1] = G$  so by Pascal's identity

$$|\mathscr{F}| = |\mathscr{H}_{n-1}| = |\mathscr{G}_{k-1}| + |\mathscr{G}_k| \le \binom{n-2}{k-2} + \binom{n-2}{k-1} = \binom{n-1}{k-1}.$$
 (2.2.2)

We have proved that if  $\mathscr{F} \subset {X \choose k}$  is intersecting and  $n \ge 2k$ , then  $|\mathscr{F}| \le {n-1 \choose k-1}$ .

We now show that if equality holds and n > 2k, then  $\mathscr{F} = \left\{ F \in {X \choose k} : x \in F \right\}$  for some  $x \in X$ . We apply induction on k and prove the theorem simultaneously for

all n > 2k. Clearly, the statement is true for k = 2 and n > 4 since, in this case, any intersecting family is contained in  $\mathscr{F}_1 = \binom{[3]}{2}$  or  $\mathscr{F}_x = \left\{F \in \binom{X}{2} : y \in F\right\}$  for some  $y \in X$ . If  $|\mathscr{F}| = \binom{n-1}{k-1}$  then (2.2.2) implies that  $|\mathscr{G}_{k-1}| = \binom{n-2}{k-2}$  and  $|\mathscr{G}_k| = \binom{n-1}{k-1}$ . Since  $\mathscr{G}_{k-1} \subset \binom{[n-1]}{k-1}$  is intersecting and n-1 > 2(k-1), the induction hypothesis implies that  $\mathscr{G}_{k-1} = \left\{G \in \binom{[n-1]}{k-1} : y \in G\right\}$  for some  $y \in [n-1]$ . Lemma 2.2.4 implies that if  $G \in \mathscr{G}_{k-1}$  and  $G' \in \mathscr{G}_k$  then  $G \cap G' \neq \emptyset$  so we conclude that  $\mathscr{G}_k = \left\{G \in \binom{[n-1]}{k} : y \in G\right\}$ and  $\mathscr{H}_{n-1} = \left\{F \in \binom{X}{k} : y \in F\right\}$ . As a result,  $\mathscr{F} = \left\{F \in \binom{X}{k} : x \in F\right\}$  for some  $x \in X$ .  $\Box$ 

## 2.2.4 Erdős-Ko-Rado via Shadows

We present Daykin's elegant proof [33] of the Erdős-Ko-Rado theorem, which shows that it is a simple corollary of Lovász's version of the Kruskal-Katona theorem. We will return to this proof in Chapter 4, as it provides a simple example of a purely combinatorial proof that does not readily generalize to vector spaces.

**Proof of Theorem 2.2.2.** Let  $\mathscr{F} \subset {\binom{X}{k}}$  be an intersecting family. Define

$$\mathscr{G} := \{X \setminus F : F \in \mathscr{F}\} \subset \binom{X}{n-k}$$

to be the family of (n-k)-sets that are the complements of the sets in  $\mathscr{F}$ . Note that  $n-k \ge k$  under our assumption that  $n \ge 2k$ . Since  $\mathscr{F}$  is intersecting, there does not exist  $F \in \mathscr{F}$  such that  $F \subset G$  for some  $G \in \mathscr{G}$ . Hence,  $\partial^k \mathscr{G}$  and  $\mathscr{F}$  are disjoint subfamilies of  $\binom{X}{k}$  so

$$|\partial^k \mathscr{G}| + |\mathscr{F}| \le \binom{n}{k}.$$
(2.2.3)

Suppose, for a contradiction, that  $|\mathscr{F}| > \binom{n-1}{k-1}$ . Hence,  $|\mathscr{G}| = |\mathscr{F}| > \binom{n-1}{k-1} = \binom{n-1}{n-k}$ . Applying Theorem 2.1.5 repeatedly, we find

$$|\partial^{n-k-1}\mathcal{G}| > \binom{n-1}{n-k-1}, \dots, |\partial^k \mathcal{G}| > \binom{n-1}{k}.$$
 (2.2.4)

We arrive at a contradiction by (2.2.3) since

$$\binom{n}{k} \ge |\partial^k \mathscr{G}| + |\partial \mathscr{F}| > \binom{n-1}{k} + \binom{n-1}{k-1} = \binom{n}{k}$$

and so we must have  $|\mathscr{F}| \leq {\binom{n-1}{k-1}}$ . We now characterize the case of equality. Suppose that  $|\mathscr{F}| = {\binom{n-1}{k-1}}$ , then  $|\mathscr{G}| = {\binom{n-1}{k-1}} = {\binom{n-1}{n-k}}$ . We have  $|\partial\mathscr{G}| \geq {\binom{n-1}{n-k-1}}$  by Theorem 2.1.5. On the other hand, (2.2.3) and (2.2.4) imply that  $|\partial\mathscr{G}| \leq {\binom{n-1}{k-1}}$ . We consequently have  $\mathscr{G} \subset {\binom{X}{n-k}}$ ,  $|\mathscr{G}| = {\binom{n-1}{n-k}}$ , and  $|\partial\mathscr{G}| = {\binom{n-1}{n-k-1}}$ . Theorem 2.1.5 therefore implies that  $\mathscr{G} = {\binom{Y}{n-k}}$  for some  $Y \in {\binom{X}{n-1}}$ . Let  $x \in X$  be the unique element in  $X \setminus Y$ ; we then have  $\mathscr{F} = \{F \in {\binom{X}{k}} : x \in F\}$  as desired.

## 2.2.5 Erdős-Ko-Rado via Cyclic Permutations

Our final proof of the Erdős-Ko-Rado theorem in this chapter is via Katona's cyclic permutation method [76]. We will use similar arguments in Chapter 4 and Chapter 7 to prove special cases of the Erdős-Ko-Rado theorem for vector spaces and the Manickam-Miklós-Singhi conjectures respectively. Let  $S_X$  denote the symmetric group on X. We say  $\alpha \in S_X$  is a *cyclic permutation* of X if  $\alpha$  has exactly one cycle when written in cycle notation. Clearly, there are (n-1)! cyclic permutations of X. If  $F \in {X \choose k}$ , we say that  $\alpha$  contains F if the elements of F are consecutive in  $\alpha$ . We first prove a lemma that shows that a cyclic permutation can contain at most k elements of an intersecting family  $\mathscr{F} \subset {X \choose k}$ .

**Lemma 2.2.5.** If  $\alpha \in S_X$  is a cyclic permutation and  $\mathscr{F} \subset {\binom{X}{k}}$  is an intersecting family, then  $\alpha$  contains at most k elements of  $\mathscr{F}$ .

Proof. Suppose that  $F \in \mathscr{F}$  appears as consecutive elements  $x_1, \ldots, x_k$  of  $\alpha$ . Since  $\mathscr{F}$  is intersecting, the only sets of k consecutive elements of  $\alpha$  which can be sets in  $\mathscr{F}$  are the k-1 sets beginning with the elements  $x_2, \ldots, x_{k-1}$  respectively and the k-1 sets ending with the elements  $x_1, \ldots, x_{k-1}$  respectively. Without loss of generality, we can assume that one of the k-1 sets beginning with  $x_2, \ldots, x_k$  lies in  $\mathscr{F}$ ; choose the last such  $x_j$ , where  $2 \le j \le k$ , for which this is true. Since  $\mathscr{F}$  is intersecting, none of the sets of k consecutive elements ending with  $x_1, \ldots, x_{j-1}$  can lie in  $\mathscr{F}$ . Hence, there are at most j + (k-j) = k elements of  $\mathscr{F}$  contained in  $\alpha$ .

**Proof of Theorem 2.2.2.** Let  $\mathscr{F} \subset {X \choose k}$  be an intersecting family. Let

 $S := \{(\alpha, F) : \alpha \in S_X \text{ is a cyclic permutation, } F \in \mathscr{F}, \text{ and } \alpha \text{ contains } F\}.$ 

We will double-count the cardinality of *S*. On one hand, each  $F \in \mathscr{F}$  is contained in k!(n-k)! cyclic permutations  $\alpha$  because there are k! ways to arrange the elements of *F* consecutively and (n-k)! ways to arrange the elements of  $X \setminus F$ . On the other hand, Lemma 2.2.5 implies that each of the (n-1)! cyclic permutations  $\alpha$  can contain at most k elements of  $\mathscr{F}$ . Putting these observations together yields

$$|\mathscr{F}|k!(n-k)! = |S| \le (n-1)!k,$$

and so  $|\mathscr{F}| \leq \binom{n-1}{k-1}$ . For a characterization of equality using the method of cyclic permutations see [65, Theorem 7.8.1].

# 2.2.6 Frankl's r-wise Intersection Theorem

The Erdős-Ko-Rado theorem asserts that if  $\mathscr{F} \subset {\binom{X}{k}}$  is an intersecting family and  $n \ge 2k$ , then  $|\mathscr{F}| \le {\binom{n-1}{k-1}}$ . Frankl [52] showed in his *r*-wise intersection theorem that we can get the same bound on  $|\mathscr{F}|$  for smaller values of *n* if the pairwise intersecting condition is strengthened. We will prove a generalization of Frankl's result for vector spaces in Chapter 5.

**Definition 2.2.6.** A family  $\mathscr{F} \subset 2^X$  is called <u>*r*-wise intersecting</u> if any *r* sets in  $\mathscr{F}$  have nonempty intersection; that is, for all  $F_1, \ldots, F_r \in \mathscr{F}$  we have  $\bigcap_{i=1}^r F_i \neq \emptyset$ .

When r = 2, then *r*-wise will be omitted since, in this case, an *r*-wise intersecting family is simply intersecting. We saw that the Erdős-Ko-Rado problem was only interesting in the case that  $n \ge 2k$ , otherwise any two elements of  $\binom{X}{k}$  intersect. Similarly, the question of determining the largest *r*-wise intersecting family is only interesting when  $n \ge (r/r - 1)k$ , otherwise any *r* elements of  $\binom{X}{k}$  intersect. On the other hand, since an *r*-wise intersecting family  $\mathscr{F} \subset \binom{X}{k}$  is intersecting, the Erdős-Ko-Rado theorem shows that  $|\mathscr{F}| \le \binom{n-1}{k-1}$  when  $n \ge 2k$ . Frankl proved that this same bound on  $|\mathscr{F}|$  holds in the range  $(r/r - 1)k \le n < 2k$ .

**Theorem 2.2.7** (Frankl). Suppose that  $\mathscr{F} \subset {X \choose k}$  is *r*-wise intersecting and  $(r-1)n \ge rk$ . Then  $|\mathscr{F}| \le {n-1 \choose k-1}$ . Equality holds if and only if  $\mathscr{F} = \left\{F \in {X \choose k} : x \in F\right\}$  for some  $x \in X$ , excepting the case r = 2 and n = 2k.
**Lemma 2.2.8** (Kleitman). Let  $k_1, \ldots, k_r \in \mathbb{Z}^+$  such that  $k_1 + \cdots + k_r = n$  and suppose  $\mathscr{F}_i \subset {X \choose k_i}$  is a family of  $k_i$ -subsets of X for  $i \in [r]$ . If there do not exist  $F_i \in \mathscr{F}_i$  with  $i \in [r]$  and  $F_1 \cup \cdots \cup F_r = X$ , then

$$\sum_{i\in [r]} \frac{|\mathscr{F}_i|}{\binom{n}{k_i}} \leq r-1.$$

Moreover, equality holds if and only if for every ordered partition  $G_1 \cup \cdots \cup G_r = X$  with  $|G_i| = k_i$ , there is exactly one  $i \in [r]$  for which  $G_i \notin \mathscr{F}_i$ .

**Proof.** Consider all ordered partitions  $G_1 \cup \cdots \cup G_r = X$  with  $|G_i| = k_i$ . Say there are *T* of them. Define the  $T \times r$  matrix *A* by

$$A_{(G_1\cup\cdots\cup G_r,\mathscr{F}_i)} = \begin{cases} 1 & \text{if } G_i \in \mathscr{F}_i \\ 0 & \text{otherwise.} \end{cases}$$

For a fixed  $F \in {\binom{X}{k_i}}$ , one has  $F \in G_i$  for a fraction  $1/{\binom{n}{k_i}}$  of all these partitions. Hence,  $G_i \in \mathscr{F}_i$  holds for a fraction  $|\mathscr{F}_i|/{\binom{n}{k_i}}$  of these partitions. The number of nonzero entries in the matrix *A* is thus

$$T\sum_{i\in[r]}\frac{|\mathscr{F}_i|}{\binom{n}{k_i}}.$$

By assumption, however, there are at most r-1 nonzero entries in each row of A. Hence,

$$T\sum_{i\in[r]}\frac{|\mathscr{F}_i|}{\binom{n}{k_i}}\leq (r-1)T \ \Rightarrow \sum_{i\in[r]}\frac{|\mathscr{F}_i|}{\binom{n}{k_i}}\leq r-1,$$

and the characterization of equality is clear.

Now we give Frankl's proof of his r-wise intersection theorem [115].

**Proof of Theorem 2.2.7** Define  $\mathscr{F}^c := \{X \setminus F : F \in \mathscr{F}\} \subset {X \choose n-k}$  to be the family of complements of sets in  $\mathscr{F}$  with respect to X. Choose  $k_1, \ldots, k_r \in [n-k]$  such that  $\sum_{i \in [r]} k_i = n$ . For  $i \in [r]$ , define  $\mathscr{F}_i := \partial^{k_i} \mathscr{F}^c$ . As  $\mathscr{F} \subset {X \choose k}$  is *r*-wise intersecting, this implies that  $\mathscr{F}^c \subset {X \choose n-k}$  does not contain *r* sets whose union is *X*. Thus  $\mathscr{F}_1, \ldots, \mathscr{F}_r$  satisfy the assumptions of Lemma 2.2.8.

We have assumed that  $(r-1)n \ge rk$  so first suppose that (r-1)n = rk. In this case, (n-k)r = n so  $k_1 = \cdots = k_r = n-k$  and  $\mathscr{F}_1 = \cdots = \mathscr{F}_r = \mathscr{F}^c$ . Hence, by Lemma 2.2.8,

$$|\mathscr{F}| = |\mathscr{F}^c| \le \frac{r-1}{r} \binom{n}{n-k} = \frac{n}{k} \binom{n}{k} = \binom{n-1}{k-1}, \qquad (2.2.5)$$

which proves the bound in this case.

Now, we characterize the case of equality in (2.2.5). By Lemma 2.2.8, if equality holds, then for every ordered partition  $G_1 \cup \cdots \cup G_r = X$ , there is exactly one  $i \in [r]$  for which  $G_i \notin \mathscr{F}^c$ . As a result, if equality holds in (2.2.5), then there cannot exist two disjoint sets in  $\binom{X}{n-k} \setminus \mathscr{F}^c$ , as then we could form an ordered partition of X with two sets missing from  $\mathscr{F}^c$ . Hence  $\binom{X}{n-k} \setminus \mathscr{F}^c$  is an intersecting family of size

$$\left| \begin{pmatrix} X \\ n-k \end{pmatrix} \setminus \mathscr{F}^c \right| = \left| \begin{pmatrix} X \\ n-k \end{pmatrix} \right| - \left| \mathscr{F}^c \right| = \begin{pmatrix} n \\ n-k \end{pmatrix} - \begin{pmatrix} n-1 \\ n-k \end{pmatrix} = \begin{pmatrix} n-1 \\ n-k-1 \end{pmatrix}.$$

Since n = r(n-k) and  $\binom{X}{n-k} \setminus \mathscr{F}^c \subset \binom{X}{n-k}$ , we see that for  $r \ge 3$ , the uniqueness of the extremal families in the Erdős-Ko-Rado theorem implies that there exists  $x \in X$  such that  $\binom{X}{n-k} \setminus \mathscr{F}^c = \{G \in \binom{X}{n-k} : x \in G\}$ . Consequently, there exists  $\hat{x} \in X$  such that  $\mathscr{F} = \{F \in \binom{X}{k} : \hat{x} \in F\}$ .

Now assume that (r-1)n > rk so that (n-k)r > n and some  $k_i < n-k$ . Suppose that  $|\mathscr{F}| \ge {\binom{n-1}{k-1}}$  so that  $|\mathscr{F}^c| \ge {\binom{n-1}{n-k}}$ . Applying Theorem 2.1.5 repeatedly yields that  $|\mathscr{F}_i| \ge {\binom{n-1}{k_i}}$ . Since  $\sum_{i \in r} k_i = n$ , we thus have

$$\sum_{i \in [r]} \frac{|\mathscr{F}_i|}{\binom{n}{k_i}} \ge \sum_{i \in [r]} 1 - \frac{k_i}{n} = r - 1.$$
(2.2.6)

Lemma 2.2.8 and (2.2.6) imply that  $|\mathscr{F}_i| = \binom{n-1}{k_i}$  and so  $|\mathscr{F}^c| = \binom{n-1}{n-k}$  and  $|\mathscr{F}| = \binom{n-1}{k-1}$ . Moreover, by Theorem 2.1.5, we must have  $\mathscr{F}^c = \binom{X \setminus \{x\}}{n-k}$  for some  $x \in X$  since there exists  $i \in [r]$  such that  $k_i < n-k$ . Consequently,  $\mathscr{F} = \{F \in \binom{X}{k} : x \in F\}$ . We have shown that if  $|\mathscr{F}| \ge \binom{n-1}{k-1}$ , then  $\mathscr{F} = \{F \in \binom{X}{k} : x \in F\}$ . Hence, we always have  $|\mathscr{F}| \le \binom{n-1}{k-1}$ , and if equality holds then  $\mathscr{F} = \{F \in \binom{X}{k} : x \in F\}$ .

#### 2.2.7 Forbidding Triangles

In the Erdős-Ko-Rado theorem, Theorem 2.2.2, and Frankl's *r*-wise intersection theorem, Theorem 2.2.7, the extremal family has the form  $\{F \in {X \choose k} : x \in F\}$  for some

 $x \in X$ . Such a family is called a *star*. What other conditions on a uniform family  $\mathscr{F} \subset {X \choose k}$  imply that the extremal family is a star? One particularly nice example, Theorem 2.2.10, was posed by Erdős and proved by Mubayi and Verstraëte [93].

**Definition 2.2.9.** A <u>triangle</u>  $\mathscr{T} = \{A, B, C\} \subset {X \choose k}$  is a family consisting of three sets A, B, C such that  $A \cap \overline{B}, A \cap \overline{C}, B \cap C$  are each nonempty but  $A \cap B \cap \overline{C} = \emptyset$ .

We may ask for the size of the largest family  $\mathscr{F} \subset {X \choose k}$  not containing a triangle. We see that the question is uninteresting unless  $n \ge 3k/2$ , otherwise  ${X \choose k}$  is the extremal example. If k = 2, then Mantel's theorem [90] asserts that  $|\mathscr{F}| \le \lfloor n^2/4 \rfloor$  and that the extremal family is a complete bipartite graph. The answer is quite different when  $k \ge 3$ .

**Theorem 2.2.10** (Mubayi-Verstraëte). Suppose that  $\mathscr{F} \subset {\binom{X}{k}}$  contains no triangle and that  $k \ge 3$  and  $n \ge 3k/2$ . Equality holds if and only if  $\mathscr{F} = \left\{F \in {\binom{X}{k}} : x \in F\right\}$  for some  $x \in X$ .

Note that if  $3k/2 \le n < 2k$  and  $\mathscr{F}$  contains no triangle, then  $\mathscr{F}$  is 3-wise intersecting. As a result, Frankl's *r*-wise intersection theorem establishes Theorem 2.2.10 in this range. We will return to Theorem 2.2.10 in Section 5.5.

#### 2.2.8 On *t*-intersecting Families

The Erdős-Ko-Rado theorem gives an upper bound on families  $\mathscr{F} \subset {X \choose k}$  for which any two sets in  $\mathscr{F}$  have nonempty intersection. If the intersection condition is strengthened to any two sets in  $\mathscr{F}$  have intersection of size at least t, where  $t \in \mathbb{Z}^+$ , then Erdős, Ko, and Rado [47] showed that a stronger upper bound on  $|\mathscr{F}|$  holds for sufficiently large n. We will see in Chapter 4 that an analog of this result holds for vector spaces.

**Definition 2.2.11.** For a positive integer  $t \in \mathbb{Z}^+$ , a family  $\mathscr{F} \subset 2^X$  is <u>t-intersecting</u> if any two sets in  $\mathscr{F}$  have intersection of size at least t; that is,  $|F \cap F'| \ge t$  for all  $F, F' \in \mathscr{F}$ .

Using algebraic methods similar to those in Section 3.3.2, Wilson [116] proved the following generalization of the Erdős-Ko-Rado theorem. Note that when t = 1, Wilson's result reduces to the Erdős-Ko-Rado theorem. **Theorem 2.2.12** (Wilson). Suppose  $\mathscr{F} \subset {\binom{X}{k}}$  is *t*-intersecting and  $n \ge (t+1)(k-t+1)$ . Then  $|\mathscr{F}| \le {\binom{n-t}{k-t}}$ . Equality holds if and only if  $\mathscr{F} = \{F \in {\binom{X}{k}} : S \subset T\}$  for some *t*-subset  $S \in {\binom{X}{t}}$ , excepting the case n = (t+1)(k-t+1).

#### 2.2.9 The Hilton-Milner Theorem

Finally, we state the Hilton-Milner theorem [70], which is the last generalization of the Erdős-Ko-Rado theorem we will discuss. Like the other results, we will see in Chapter 5 that Hilton and Milner's theorem can also be generalized to vector spaces. Frankl and Füredi [54] gave an elegant proof of the Hilton-Milner theorem using the shifting technique.

**Theorem 2.2.13** (Hilton-Milner). Let  $\mathscr{F} \subset {\binom{X}{k}}$  be an intersecting family with  $k \ge 2$ ,  $n \ge 2k + 1$ , and such that there does not exist  $x \in X$  such that  $\mathscr{F} \subset \{F \in {\binom{X}{k}} : x \in F\}$ . We then have

$$|\mathscr{F}| \le \binom{n-1}{k-1} - \binom{n-k-1}{k-1} + 1.$$

Equality holds if and only if

(i) 
$$\mathscr{F} = \{F\} \cup \{G \in \binom{X}{k} : x \in G, F \cap G \neq \emptyset\}$$
 for some k-subset F and  $x \in X \setminus F$ .

(ii) 
$$\mathscr{F} = \{F \in {X \choose 3} : |F \cap S| \ge 2\}$$
 for some 3-subset S if  $k = 3$ .

## 2.3 Fisher's Inequality

In Section 2.2.1, we were concerned with an upper bound for intersecting families, that is those families  $\mathscr{F} \subset {X \choose k}$  for which  $|F_1 \cap F_2| \ge 1$  for every  $F_1, F_2 \in \mathscr{F}$ . In this section, we will drop the uniformity condition, but insist that distinct elements  $F_1, F_2$ of our family  $\mathscr{F} \subset 2^X$  satisfy  $|F_1 \cap F_2| = \lambda$ , where  $\lambda \in \mathbb{N}$ . Such families are called  $\lambda$ -intersecting.

In Section 2.3.1, we present the de Bruijn-Erdős theorem [34], which bounds the size of maximum 1-intersecting families and characterizes the extremal examples. We then generalize the de-Bruijn Erdős theorem in Section 2.3.3 by stating Fisher's Inequality [17, 51, 73, 85], which handles the same question for general  $\lambda$ . We will also state a conjecture of Frankl and Füredi [55] that generalizes Fisher's Inequality, and then present a proof of their conjecture in the special case that  $\lambda = 1$ . Frankl and Füredi's proof is similar to that of de Bruijn and Erdős and will make use of convexity, an important tool in extremal combinatorics.

#### 2.3.1 The de Bruijn-Erdős Theorem

We will state the de Bruijn-Erdős theorem [34], which bounds the size of maximum 1-intersecting families and characterizes the extremal examples.

**Definition 2.3.1.** Given  $\lambda \in \mathbb{N}$ , a family  $\mathscr{F} \subset 2^X$  is  $\underline{\lambda}$ -intersecting if, for any distinct  $F_1, F_2 \in \mathscr{F}$ , we have  $|F_1 \cap F_2| = \lambda$ .

**Definition 2.3.2.** A family  $\mathscr{F} \subset 2^X$  is k-uniform if  $\mathscr{F} \subset {X \choose k}$ .

**Definition 2.3.3.** A family  $\mathscr{F} \subset 2^X$  is *r*-regular if deg(x) = r for all  $x \in X$ .

We now give some examples of 1-intersecting families.

**Definition 2.3.4.** We say  $\mathscr{F}$  is <u>trivial</u> if there exists  $x \in X$  with deg $(x) = |\mathscr{F}|$ , and is <u>nontrivial</u> otherwise.

**Definition 2.3.5.** A family  $\mathscr{F} \subset 2^X$  is called a near-pencil if

 $\mathscr{F} = \{\{1,2\},\{1,3\},\ldots,\{1,n\},\{2,3,\ldots,n\}\}.$ 

**Definition 2.3.6.** A family  $\mathscr{F} \subset {\binom{X}{k}}$  is called a <u>projective plane</u> if  $\mathscr{F}$  is 1-intersecting, uniform, and  $|\mathscr{F}| = n$ .



Figure 2.1: The Fano plane is an example of a projective plane.

**Theorem 2.3.7** (de Bruijn-Erdős). Suppose  $\mathscr{F} \subset 2^X$  is a 1-intersecting family of size m. We have  $m \leq n$ . Moreover, if equality holds, then exactly one of the following is true:

- (i) there is  $F \in \mathscr{F}$  such that |F| = 1,
- (ii)  $\mathcal{F}$  is a near-pencil,
- *(iii) F is a projective plane.*

## 2.3.2 Proof of the de Bruijn-Erdős Theorem

In this section, we will prove the de Bruijn-Erdős theorem, Theorem 2.3.7. First, we state and prove a lemma to which the theorem reduces.

**Lemma 2.3.8.** Let  $\mathscr{F} \subset 2^X$  be a family of size m. Suppose  $\deg(x) < m$  and |F| < n for every  $x \in X$  and  $F \in \mathscr{F}$ . Assume further that if  $x \notin F$ , then  $\deg(x) \leq |F|$ . We then have  $m \leq n$ .

*Proof.* Suppose, for a contradiction, that m > n. Given any pair (x, F) for which  $x \notin F$ , we then have

$$\frac{\deg(x)}{m - \deg(x)} < \frac{|F|}{n - |F|}.$$

Summing this inequality over all pairs (x, F) for which  $x \notin F$  yields a contradiction,

$$\sum_{F \in \mathscr{F}} |F| = \sum_{x \in X} \deg(x) = \sum_{x \in X} (m - \deg(x)) \frac{\deg(x)}{m - \deg(x)} < \sum_{(x,F), x \notin F} \frac{\deg(x)}{m - \deg(x)} < \sum_{(x,F), x \notin F} \frac{|F|}{n - |F|} = \sum_{F \in \mathscr{F}} (n - |F|) \frac{|F|}{n - |F|} = \sum_{F \in \mathscr{F}} |F|.$$
(2.3.1)

Hence,  $m \le n$ . Also observe that if m = n, then (2.3.1) yields that deg(x) = |F| if  $x \notin F$ .

Now we prove the de Bruijn-Erdős theorem [34].

**Proof of Theorem 2.3.7.** If there exists  $F \in \mathscr{F}$  such that |F| = 1, then all the other sets contain *F* and are disjoint otherwise. It follows that  $m \le (n-1) + 1 = n$ .

Hence, for each  $x \in X$ , we can assume deg(x) < m; otherwise we can add  $\{x\}$  to  $\mathscr{F}$  if it is not there already, and since  $|\{x\}| = 1$ , we are done by the previous paragraph.

We can also assume for each  $F \in \mathscr{F}$  that |F| < n, otherwise m = n = 1. We claim that if  $x \notin F$ , then  $\deg(x) \le |F|$ . This is because every  $E \in \mathscr{F}(x) := \{E \in \mathscr{F} : x \in E\}$  intersects F in precisely one element and no two distinct  $E \in \mathscr{F}(x)$  intersect F in the same element as  $\mathscr{F}$  is 1-intersecting. Hence, we are in the situation of Lemma 2.3.8, so  $m \le n$ .

We now characterize the nontrivial extremal families. We first prove that every pair of points is contained in some  $F \in \mathscr{F}$ . Suppose, for a contradiction, that there exist distinct  $x, y \in X$  such that there is no  $F \in \mathscr{F}$  for which  $x, y \in F$ . Since m = n, we must have deg $(y) \neq 0$ , so choose  $F' \in \mathscr{F}$  such that  $y \in F'$ . We have  $x \notin F'$  by assumption, so Lemma 2.3.8 implies that deg(x) = |F'|. As  $\mathscr{F}$  is 1-intersecting, every  $E \in \mathscr{F}(x)$ intersects F' in precisely one element and no two distinct  $E \in \mathscr{F}(x)$  intersect F' in the same element. Consequently, deg(x) = |F'| implies that some  $E \in \mathscr{F}(x)$  must contain y, which contradicts our initial assumption about x and y. This proves that if  $\mathscr{F}$  is a nontrivial 1-intersecting family with m = n, then every pair of points is contained in some  $F \in \mathscr{F}$ .

We distinguish two cases according to whether there exist distinct  $F_1, F_2 \in \mathscr{F}$ such that  $X = F_1 \cup F_2$ . First, assume there exist distinct  $F_1, F_2 \in \mathscr{F}$  such that  $X = F_1 \cup F_2$ . Suppose, for a contradiction, that  $|F_1|, |F_2| \ge 3$ . Let  $x = F_1 \cap F_2$  and let  $y_1, y_2$  and  $z_1, z_2$ be two other points on  $F_1$  and  $F_2$  respectively. By the result of the previous paragraph, there exist  $F_3, F_4 \in \mathscr{F}$  such that  $y_1, z_1 \in F_3$  and  $y_2, z_2 \in F_4$ . Since  $\mathscr{F}$  is 1-intersecting and  $X = F_1 \cup F_2$ , we see  $F_3 = \{y_1, z_1\}$  and  $F_4 = \{y_2, z_2\}$ . We obtain a contradiction since  $F_3 \cap F_4 = \emptyset$ . Hence, we cannot have both  $|F_1|, |F_2| \ge 3$ . Without loss of generality, assume  $|F_1| < 3$ ; we must have  $|F_1| = 2$  and  $|F_2| = n - 1$  since  $\mathscr{F}$  is nontrivial and  $X = F_1 \cup F_2$ . As every pair of points belongs to some  $F \in \mathscr{F}$ , we see that  $\mathscr{F}$  must be a near-pencil.

Now suppose there does not exist  $F_1, F_2 \in \mathscr{F}$  such that  $X = F_1 \cup F_2$ . Fix some  $F \in \mathscr{F}$  and suppose |F| = k. For any other  $F' \in \mathscr{F}$ , we can choose  $x \in X$  such that  $x \notin F \cup F'$ . By Lemma 2.3.8, we have  $|F'| = \deg(x) = |F| = k$  so  $\mathscr{F}$  is *k*-uniform. Hence  $\mathscr{F}$  is a projective plane.

### 2.3.3 The Frankl-Füredi Conjecture

The de Bruijn-Erdős theorem proved that if  $\mathscr{F} \subset 2^X$  is a 1-intersecting family of size *m*, then  $m \leq n$ . The well-known Fisher's Inequality establishes the same conclusion more generally for  $\lambda$ -intersecting families.

**Theorem 2.3.9** (Fisher's Inequality). *If*  $\lambda \in \mathbb{Z}^+$  *and*  $\mathscr{F} \subset 2^X$  *is a*  $\lambda$ *-intersecting family of size m, then*  $m \leq n$ .

Fisher [51] proved Theorem 2.3.9 under the additional constraint that  $\mathscr{F}$  is regular (and hence uniform). Bose [17] proved Theorem 2.3.9 when  $\mathscr{F}$  is required to be uniform. Majindar [85] first proved Theorem 2.3.9 as stated, and his proof was later rediscovered by Isbell [73]. We will defer a proof of Theorem 2.3.9 until Chapter 3, as its proof was one of the first to demonstrate the power of linear algebra.

Another way to restate Fisher's Inequality is that if  $\mathscr{F} \subset 2^X$  is a  $\lambda$ -intersecting family of size *m* then  $|\partial^1 \mathscr{F}| \ge m$ . Inspired by Fisher's Inequality, Frankl and Füredi [55] conjectured a similar inequality for  $|\partial^2 \mathscr{F}|$  and verified it when  $\lambda = 1$ .

**Conjecture 2.3.10** (Frankl-Füredi). If  $\lambda \in \mathbb{Z}^+$  and  $\mathscr{F} \subset 2^X$  is a nontrivial  $\lambda$ -intersecting family of size *m*, then

$$|\partial^2 \mathscr{F}| \ge \binom{m}{2}.$$

**Theorem 2.3.11** (Frankl-Füredi). If  $\mathscr{F} \subset 2^X$  is a nontrivial 1-intersecting family of size m then  $|\partial^2 \mathscr{F}| \ge {m \choose 2}$ .

#### 2.3.4 Convexity

We will give more background on Conjecture 2.3.10 in Chapter 6. In this section, we will concern ourselves with Frankl and Füredi's proof of Theorem 2.3.11; their argument is similar to de Bruijn and Erdős's proof of Theorem 2.3.7 and uses convexity. We now define the concepts of convexity and Schur convexity and state a theorem that will be used in the proof of Frankl and Füredi's Conjecture 2.3.10 for  $\lambda = 1$ .

**Definition 2.3.12.** We say a function  $f : \mathbb{R} \to \mathbb{R}$  is <u>convex</u> if for all  $x_1, x_2 \in \mathbb{R}$  and any  $t \in [0, 1]$  we have

$$f(tx_1 + (1-t)x_2) \le tf(x_1) + (1-t)f(x_2).$$

**Definition 2.3.13.** For  $\alpha := (\alpha_1 \dots, \alpha_d) \in \mathbb{R}^d$ , we let  $\alpha^* := (\alpha_1^*, \dots, \alpha_d^*)$  denote the vector with the same components as  $\alpha$  but sorted in decreasing order. Given  $\alpha, \beta \in \mathbb{R}^d$ , we say  $\alpha$  majorizes  $\beta$  and we write  $\alpha \succeq \beta$  if for all  $k \in [d]$ , we have

$$\sum_{i=1}^k lpha_i^* \geq \sum_{i=1}^k eta_i^*, \quad \sum_{i=1}^d lpha_i^* = \sum_{i=1}^d eta_i^*.$$

**Definition 2.3.14.** We say  $F : \mathbb{R}^d \to \mathbb{R}$  is <u>Schur convex</u> if  $F(\alpha) \ge F(\beta)$  for all  $\alpha, \beta \in \mathbb{R}^d$  for which  $\alpha \succeq \beta$ .

**Theorem 2.3.15.** Suppose  $f : \mathbb{R} \to \mathbb{R}$  is a convex function. We then have that the function  $F : \mathbb{R}^d \to \mathbb{R}$  defined by the sum

$$F(x_1,\ldots,x_d) = \sum_{k=1}^d f(x_k)$$

is Schur convex.

For a proof of Theorem 2.3.15 see [105, Chapter 13]. We now present Frankl and Füredi's proof [55] of Conjecture 2.3.10 when  $\lambda = 1$ .

**Proof of Theorem 2.3.11** Let  $\mathscr{F} \subset 2^X$  be a nontrivial 1-intersecting family of size m. For  $x_i \in X$  and  $F_j \in \mathscr{F}$ , let  $d_i := \deg(x_i)$  and  $e_j := |F_j|$ . Order  $F_1, \ldots, F_m$  so that  $e_1 \ge \ldots \ge e_m$ . Similarly order  $x_1, \ldots, x_n$  so that  $d_1 \ge \ldots \ge d_n$ . We will show that

$$e_i \ge d_i. \tag{2.3.2}$$

Since  $\mathscr{F}$  is 1-intersecting, we have that if  $x \notin F$ , then  $\deg(x) \leq |F|$ ; this is because every  $E \in \mathscr{F}(x) := \{E \in \mathscr{F} : x \in E\}$  intersects F in precisely one element and no two distinct  $E \in \mathscr{F}(x)$  intersect F in the same element. Let  $F_j \in \mathscr{F}$  be any set that does not contain  $\{x_1, \ldots, x_i\}$ . Hence, for some  $k \in [i]$  we have  $x_k \notin F$  so  $e_j \geq d_k \geq d_i$ . Consequently, (2.3.2) holds if we have i sets  $F \in \mathscr{F}$  that do not contain  $\{x_1, \ldots, x_i\}$ . For i = 1, we certainly have one set  $F \in \mathscr{F}$  that does not contain  $x_1$  since  $\mathscr{F}$  is nontrivial. For i > 1, suppose there are only at most i - 1 sets  $F \in \mathscr{F}$  that do not contain  $\{x_1, \ldots, x_i\}$ . Since  $i \geq 2$  and  $\mathscr{F}$  is 1-intersecting, there can be at most one set in  $\mathscr{F}$  that contains  $\{x_1, \ldots, x_i\}$  so  $m \leq i$ . As there are i sets and  $d_1 \geq \ldots \geq d_i$ , we have  $d_i \leq 1$  so  $e_i \geq 2 > 1 \geq d_i$ . We see that in all cases (2.3.2) holds.

Let  $\delta := (d_1, \dots, d_n)$ ,  $\varepsilon := (e_1, \dots, e_m, 0, \dots, 0) \in \mathbb{R}^n$ . We have  $\varepsilon \succeq \delta$  since  $e_i \ge d_i$ and  $\sum_{i=1}^m e_i = \sum_{i=1}^n d_i$ . The function  $f(x) = {x \choose 2}$  is convex so by Theorem 2.3.15, the function  $F : \mathbb{R}^n \to \mathbb{R}$  defined by

$$F(x_1,\ldots,x_n):=\sum_{i=1}^n \binom{x_i}{2}$$

is Schur convex. Since  ${\mathscr F}$  is 1-intersecting, we have

$$F(\delta) = F(d_1, \dots, d_n) = \sum_{i=1}^n \binom{d_i}{2} = |\{(x, F, F') : F, F' \in \mathscr{F}, F \cap F' = \{x\}\}| = \binom{m}{2},$$
  
$$F(\varepsilon) = F(e_1, \dots, e_m, 0, \dots, 0) = \sum_{i=1}^m \binom{e_i}{2} = |\{(\{x, y\}, F) : F \in \mathscr{F}, x, y \in F\}| = |\partial^2 \mathscr{F}|.$$

We showed  $\varepsilon \succeq \delta$  and so

$$|\partial^2 \mathscr{F}| = F(\varepsilon) > F(\delta) = \binom{m}{2},$$

as desired.

С		۱.
н		L
		L
н		L

# Chapter 3

# **Algebraic Techniques**

In this chapter, we describe algebraic methods that are frequently used in extremal set theory and in this thesis. First, we make good on our promise in Chapter 2 and prove Fisher's Inequality, Theorem 2.3.9. We also discuss the major open problem of characterizing the extremal families in Fisher's Inequality, which is known as the  $\lambda$ -design conjecture [99, 117]. Next, we discuss a generalization of Fisher's Inequality to *L*-intersecting families [59], whose proof illustrates the polynomial method. This result and its variants have powerful combinatorial and geometric consequences; they have been used, for example, to disprove Borsuk's conjecture in topology [74]. We then prove the Erdős-Ko-Rado theorem via the eigenvalue method [66, 116]; variants of this method have yielded Erdős-Ko-Rado analogs in many different structures. Finally, we discuss linear programming and use this technique to prove Baranyai's theorem [10] on decompositions of the complete hypergraph into perfect matchings. We will use Baranyai's theorem in Chapter 5 and Chapter 7.

# **3.1** Fisher's Inequality via Linear Algebra

We prove Fisher's Inequality, Theorem 2.3.9, via linear algebra. Given a family of sets  $\mathscr{F} \subset 2^X$ , a natural way to represent it is via its incidence matrix. Majindar's [85] and Isbell's [73] ingenious proof of Fisher's Inequality analyzes the rank of the incidence matrix.

**Definition 3.1.1.** If  $\mathscr{F} \subset 2^X$  is a family of sets with size  $|\mathscr{F}| = m$ , then its associated incidence matrix M is defined to be the  $m \times n$  matrix with

$$M_{ij} := \begin{cases} 1 & if \quad j \in F_i \\ 0 & if \quad j \notin F_i. \end{cases}$$

**Proof of Theorem 2.3.9.** Let  $\mathscr{F} \subset 2^X$  be a  $\lambda$ -intersecting family. First, we consider the case where some  $F \in \mathscr{F}$  has exactly  $\lambda$  elements. Since  $\mathscr{F}$  is  $\lambda$ -intersecting, all the other sets in  $\mathscr{F}$  contain F and are disjoint otherwise. Hence  $m \leq n+1-\lambda \leq n$  as  $\lambda \in \mathbb{Z}^+$ .

We may thus assume that the numbers  $\gamma_i := |F_i| - \lambda$  are all positive for  $1 \le i \le m$ . Let *M* be the incidence matrix of  $\mathscr{F}$ , and define the matrix  $A := MM^T$ . Observe that  $A_{ij} = |F_i \cap F_j|$ ; since  $\mathscr{F}$  is  $\lambda$ -intersecting, we have  $A = \lambda J + C$ , where *J* is the  $m \times m$  all ones matrix and *C* is the diagonal matrix with entries  $C_{ii} := \gamma_i$ . Note that  $\lambda J$  is a positive semidefinite matrix and that *C* is a positive definite matrix since  $\gamma_i > 0$ . Consequently, *A* is a positive definite matrix so  $m = \operatorname{rank} A \le \operatorname{rank} M \le n$ .

#### **3.1.1** The Extremal Families in Fisher's Inequality

Characterizing the extremal families  $\mathscr{F} \subset 2^X$  in Fisher's Inequality is a major open problem and is known as the  $\lambda$ -design conjecture. In the de Bruijn-Erdős theorem, Theorem 2.3.7, the crucial step in characterizing the extremal families is proving that every pair of points in a nontrivial extremal family  $\mathscr{F}$  is contained in some  $F \in \mathscr{F}$ ; this is the content of Frankl and Füredi's generalization, Theorem 2.3.11. While Babai [3] has shown that Frankl and Füredi's Conjecture 2.3.10 is true for all  $\lambda \ge 1$  in the case that m = n, this does not lead to a characterization of the extremal families in Fisher's Inequality unfortunately. We will partition the extremal families in Fisher's Inequality according to whether they are uniform.

**Definition 3.1.2.** For  $\lambda \in \mathbb{Z}^+$ , a  $\lambda$ -intersecting family  $\mathscr{F} \subset {\binom{X}{k}}$  is a <u>symmetric design</u> if it is uniform and has cardinality  $|\mathscr{F}| = n$ .

**Definition 3.1.3.** For  $\lambda \in \mathbb{Z}^+$ , a  $\lambda$ -intersecting family  $\mathscr{F} \subset 2^X$  is a  $\underline{\lambda}$ -design if  $\mathscr{F}$  is not uniform and  $|\mathscr{F}| = n$ .

#### **3.1.2** Symmetric Designs

We now give Ryser's [98] elegant algebraic proof which shows that a symmetric design  $\mathscr{F} \subset {X \choose k}$  is *k*-regular and that any pair of points  $\{x, y\} \in {X \choose 2}$  is contained in exactly  $\lambda$  sets in  $\mathscr{F}$ . Ryser's proof provides another application of incidence matrices.

**Theorem 3.1.4** (Ryser). If  $\mathscr{F} \subset {\binom{X}{k}}$  is a symmetric design, then  $\mathscr{F}$  is k-regular and any pair of points  $\{x, y\} \in {\binom{X}{2}}$  is contained in exactly  $\lambda$  sets in  $\mathscr{F}$ .

**Proof.** Let *M* be the  $n \times n$  incidence matrix of  $\mathscr{F}$  and observe that

$$M^{T}M_{ij} := \begin{cases} \deg(x_{i}) & \text{if } i = j \\ \operatorname{codeg}(x_{i}, x_{j}) & \text{if } i \neq j, \end{cases}$$
(3.1.1)

where  $\operatorname{codeg}(x_i, x_j) := |\{F \in \mathscr{F} : \{x_i, x_j\} \in F\}|$  denotes the number of sets in  $\mathscr{F}$  that contain  $\{x_i, x_j\}$ . We will show that  $M^T M = \lambda J + (k - \lambda)I$ ; hence (3.1.1) implies that  $\mathscr{F}$ is *k*-regular and that any pair of points  $\{x, y\} \in {X \choose 2}$  is contained in exactly  $\lambda$  sets in  $\mathscr{F}$ . Note that  $MM_{ij}^T = |F_i \cap F_j|$  so  $MM^T = \lambda J + (k - \lambda)I$  since  $\mathscr{F} \subset {X \choose k}$  is  $\lambda$ -intersecting. As in the proof of Theorem 2.3.9, we conclude that  $MM^T$  is positive definite so  $M^{-1}$ exists. We thus have that

$$M^{T} = M^{-1}(\lambda J + (k - \lambda)I) \implies M^{T}M = \lambda M^{-1}JM + (k - \lambda)I.$$
(3.1.2)

Since  $\mathscr{F}$  is *k*-uniform, we have MJ = kJ, which implies that  $M^{-1}J = k^{-1}J$ . Also note that  $JM_{ij} = \deg(x_j)$ . Hence, by (3.1.2), we have

$$M^{T}M_{ij} := \begin{cases} \lambda k^{-1} \deg(x_{i}) + (k - \lambda) & \text{if } i = j \\ \lambda k^{-1} \deg(x_{j}) & \text{if } i \neq j. \end{cases}$$
(3.1.3)

Since  $M^T M$  is symmetric, (3.1.3) implies that  $\deg(x_1) = \deg(x_j)$  for all  $2 \le j \le n$ . We also have  $nk = \sum_{x \in X} \deg(x) = n \deg(x_1)$  since the sum of the rows of M equals the sum of the columns of M; hence  $\mathscr{F}$  is *k*-regular. By (3.1.1), we see

$$\operatorname{codeg}(x_i, x_j) = M^T M_{ij} = \lambda k^{-1} \operatorname{deg}(x_j) = \lambda.$$

Hence, any pair of points  $\{x, y\} \in {\binom{X}{2}}$  is contained in exactly  $\lambda$  sets in  $\mathscr{F}$ .

Recall that projective planes are symmetric designs for  $\lambda = 1$ . By Theorem 3.1.4, every pair of points in a projective plane  $\mathscr{F}$  is contained in exactly one set so

$$\binom{n}{2} = n\binom{k}{2} \implies n = (k-1)^2 + (k-1) + 1.$$

This observation motivates the definition of the order of a projective plane.

**Definition 3.1.5.** If  $\mathscr{F} \subset {\binom{X}{k}}$  is a projective plane, then its <u>order</u> is k-1.

The projective plane of order 1 is the triangle  $\binom{[3]}{2}$  and the projective plane of order 2 is the Fano plane, Figure 2.1. We now show that if  $q \in \mathbb{Z}^+$  is a prime power, then a projective plane of order q exists.

## **Lemma 3.1.6.** If $q \in \mathbb{Z}^+$ is a prime power, then a projective plane of order q exists.

**Proof.** Since *q* is a prime power, there exists a finite field  $\mathbb{F}_q$  of order *q*. Let  $V = \mathbb{F}_q^3$  be a three-dimensional vector space over  $\mathbb{F}_q$ . Let  $\begin{bmatrix} V \\ i \end{bmatrix} := \{S \subset V : \dim(S) = i\}$  be the family of *i*-dimensional subspaces of *V* for  $i \in [3]$ . Let  $X = \begin{bmatrix} V \\ 1 \end{bmatrix}$  and let  $\mathscr{F} := \begin{bmatrix} V \\ 2 \end{bmatrix}$ . We have

$$|X| = \frac{q^3 - 1}{q - 1} = q^2 + q + 1 = \frac{(q^3 - 1)(q^3 - q)}{(q^2 - 1)(q^2 - q)} = |\mathscr{F}|$$

If  $F \in \mathscr{F}$  then  $|F| = (q^2 - 1)/(q - 1) = q + 1$ . We have  $\dim(F_1 \cap F_2) = 1$  for distinct  $F_1, F_2 \in \mathscr{F}$  since  $\dim V = 3$  and  $\dim F_1 = \dim F_2 = 2$ . Consequently,  $\mathscr{F} \subset {X \choose q+1}$  is a 1-intersecting family with  $|\mathscr{F}| = |X|$ , so  $\mathscr{F}$  is a projective plane of order q.

By Lemma 3.1.6, there is a projective plane of order  $q \in \{2,3,4,5,7,8,9\}$ . Is there a projective plane of order 6 or of order 10? The celebrated Bruck-Ryser-Chowla theorem [23, 28] answers the first question in the negative, and is the definitive tool in proving the nonexistence of symmetric designs. Note that the existence of a projective plane of order 10 is not ruled out by the Bruck-Ryser-Chowla theorem, but has been ruled out by a massive computer search [81].

**Theorem 3.1.7** (Bruck-Ryser-Chowla). Let  $\mathscr{F} \subset {\binom{X}{k}}$  be a symmetric design.

- (i) If  $|\mathscr{F}|$  is even, then  $k \lambda$  is a square.
- (ii) If  $|\mathscr{F}|$  is odd, then the equation  $z^2 = (k \lambda)x^2 + (-1)^{(|\mathscr{F}| 1)/2}\lambda y^2$  has a solution in integers x, y, z, not all zero.

Lemma 3.1.6 proves there are infinitely many symmetric designs for  $\lambda = 1$ . We do not know if the same is true for  $\lambda > 1$ , and a folklore conjecture asserts that the answer is no.

**Conjecture 3.1.8** (Folklore). *There are finitely many symmetric designs for fixed*  $\lambda > 1$ .

#### **3.1.3** The $\lambda$ -design Conjecture

Ryser [99] and Woodall [117] have shown that the points in a  $\lambda$ -design have only two degrees, say  $r_1$  and  $r_2$ .

**Theorem 3.1.9** (Ryser-Woodall). If  $\mathscr{F} \subset 2^X$  is a  $\lambda$ -design, then deg $(x) \in \{r_1, r_2\}$  and  $r_1 + r_2 = n + 1$ .

Woodall [117] has shown that for each  $\lambda > 1$ , there are only finitely many  $\lambda$  designs, which is interesting because the corresponding problem for symmetric designs, Conjecture 3.1.8, is not yet solved.

**Theorem 3.1.10** (Woodall). If  $\lambda > 1$ , there are finitely many  $\lambda$ -designs.

The only class of nontrivial  $\lambda$ -designs known are the point-complemented block designs.

**Definition 3.1.11.** Let  $\mathscr{F} \subset {X \choose k}$  be a symmetric design. A  $\lambda$ -intersecting family  $\mathscr{G} \subset 2^X$  is a point-complemented block design if there exists  $F' \in \mathscr{F}$  such that

$$\mathscr{G} := \{F'\} \cup \{F \triangle F' : F' \neq F \in \mathscr{F}\}.$$

In the de Bruijn-Erdős theorem, Theorem 2.3.7, the 1-designs of type (ii) are point complements of the symmetric designs  $\binom{X}{n-1}$ . Ryser [99] has shown that the unique 2-design  $\hat{\mathscr{F}}$  is the point complement of the Fano plane,

$$\widehat{\mathscr{F}} := \{\{1,2,4\},\{1,4,6,7\},\{1,2,5,7\},\{1,2,3,6\},\{1,3,4,5\},\{2,3,4,7\},\{2,4,5,6\}\}.$$

The  $\lambda$ -design conjecture [99, 117] asserts that all nontrivial  $\lambda$ -designs are of this type.

**Conjecture 3.1.12** (Ryser-Woodall). All nontrivial  $\lambda$ -designs are point-complemented block designs.

The de Bruijn-Erdős theorem, Theorem 2.3.7, proves Conjecture 3.1.12 in the case  $\lambda = 1$ . Conjecture 3.1.12 has been verified for all  $\lambda \leq 34$  [19, 20, 21, 79, 99, 101, 108, 114], and is also true for a few infinite families of  $\lambda$  [102, 103].

# **3.2** *L*-intersecting Families

We now discuss a generalization of Fisher's Inequality to *L*-intersecting families, whose proof illustrates the polynomial method. This result and its many variants have powerful combinatorial and geometric consequences; they have been used, for example, in the counterexample to Borsuk's conjecture from topology [74].

**Definition 3.2.1.** Given a finite set  $L \subset \mathbb{N}$  of nonnegative integers, we say a family  $\mathscr{F} \subset 2^X$  is L-intersecting if for all distinct  $F_1, F_2 \in \mathscr{F}$ , we have  $|F_1 \cap F_2| \in L$ .

A celebrated theorem of Frankl and Wilson [59] bounds the size of *L*-intersecting families as a function of |L|.

**Theorem 3.2.2** (Frankl-Wilson). Suppose  $L \subset \mathbb{N}$  has size |L| = s. If  $\mathscr{F} \subset 2^X$  is an *L*-intersecting family, then

$$|\mathscr{F}| \le \sum_{i=0}^{s} \binom{n}{i}.$$
(3.2.1)

This result is best possible in terms of the parameters *n* and *s*, as demonstrated by taking  $L = \{0, ..., s - 1\}$  and  $\mathscr{F}$  to be the family of all subsets of *X* of size at most *s*. However, it is possible to get sharper bounds by specifying *L*. For example, if  $\mathscr{F} \subset 2^X$  is  $\lambda$ -intersecting, where  $\lambda \in \mathbb{Z}^+$ , then Fisher's Inequality yields that  $|\mathscr{F}| \le n$ , whereas the *statement* of Theorem 3.2.2 gives the weaker bound  $|\mathscr{F}| \le n+1$  since  $|L| = |\{\lambda\}| = 1$ . We shall see, however, that the *proof* of Theorem 3.2.2 can be modified in this case to yield that  $|\mathscr{F}| \le n$ . In general, though, it is an open problem to even determine the order of magnitude of the largest *L*-intersecting family for a specific set *L*.

### 3.2.1 Polynomial Spaces

Given a field  $\mathbb{F}$ , the polynomial ring  $\mathbb{F}[x_1, \dots, x_n]$  in the indeterminates  $x_1, \dots, x_n$  is the set of all finite sums of *monomial terms*, which are elements of the form

$$ax_1^{d_1}\cdots x_n^{d_n},\tag{3.2.2}$$

where  $a \in \mathbb{F}$ , and  $d_i \in \mathbb{N}$ . A monomial is called *monic* if a = 1. In (3.2.2), the exponent  $d_i$  is called the *degree in*  $x_i$  of the term and the sum  $d = d_1 + \dots + d_n$  is called the *degree* of the term. A *polynomial* is a finite sum of nonzero monomial terms, and its degree is the largest degree of any of its monomial terms. A polynomial is *homogeneous of degree* k if each of its monomial terms has degree k. The monomial (3.2.2) is called *multilinear* if each  $d_i \in \{0, 1\}$ , and a *multilinear polynomial* is a finite sum of nonzero multilinear standard abstract algebra texts such as [41] and will be assumed.

We will now show that certain subsets of  $\mathbb{F}[x_1, \ldots, x_n]$  form finite-dimensional vector spaces over  $\mathbb{F}$  and compute their dimension.

**Lemma 3.2.3.** The set T of multilinear homogeneous polynomials of degree k with coefficients in  $\mathbb{F}$  forms a vector space over  $\mathbb{F}$  with dimension dim  $T = \binom{n}{k}$ .

**Proof.** We can more concretely write T as

 $T = \{f \in \mathbb{F}[x_1, \dots, x_n] : f \text{ is multilinear and homogeneous of degree } k\} \cup \{0\}.$ 

Clearly, *T* forms a vector space over  $\mathbb{F}$  and has basis  $\{x_{i_1} \dots x_{i_k} : i_1 < \dots < i_k\}$ . Hence, dim  $T = \binom{n}{k}$ .

A similar argument yields that the space *W* of multilinear polynomials of degree at most *s* and with coefficients in  $\mathbb{F}$  has dimension dim $W = \sum_{i=0}^{s} {n \choose i}$ .

**Lemma 3.2.4.** The set  $W := \{f \in \mathbb{F}[x_1, ..., x_n] : f \text{ is multilinear of degree at most } s\}$  forms a vector space over  $\mathbb{F}$  with dimension

$$\dim W = \sum_{i=0}^{s} \binom{n}{i}.$$

Given a finite set  $\Omega$ , it is easy to see that the set  $\mathbb{F}^{\Omega} := \{f : \Omega \to \mathbb{F}\}$  forms a vector space over  $\mathbb{F}$  called the *function space*. The Triangular Criterion, Lemma 3.2.5, gives a sufficient condition for showing that a set of functions in  $\mathbb{F}^{\Omega}$  is linearly independent.

**Lemma 3.2.5.** For  $i \in [m]$ , let  $f_i : \Omega \to \mathbb{F}$  be functions and  $a_i \in \Omega$  be elements such that

$$f_i(a_j) \begin{cases} \neq 0 & \text{if } i = j \\ = 0 & \text{if } j < i. \end{cases}$$

$$(3.2.3)$$

Then  $f_1, \ldots, f_m$  are linearly independent members of the function space  $\mathbb{F}^{\Omega}$ .

**Proof.** Suppose, for a contradiction, that there exist constants  $c_i \in \mathbb{F}$  not all zero such that  $\sum_{i=1}^{m} c_i f_i = 0$ . Let *i'* be the smallest *i* such that  $c_i \neq 0$ . By (3.2.3),

$$0 = \sum_{i=1}^{m} c_i f_i(a_{i'}) = c_{i'} f_{i'}(a_{i'}),$$

which implies that  $c_{i'} = 0$ , a contradiction. Hence,  $f_1, \ldots, f_m$  are linearly independent members of  $\mathbb{F}^{\Omega}$ .

The last lemma we need before commencing with the proof of Theorem 3.2.2 shows that for any polynomial  $f \in \mathbb{F}[x_1, \dots, x_n]$  of degree at most *s*, there exists a unique multilinear polynomial  $\hat{f} \in \mathbb{F}[x_1, \dots, x_n]$ , which has the same values as *f* on the set  $\{0, 1\}^n$ .

**Lemma 3.2.6.** For any polynomial  $f \in \mathbb{F}[x_1, ..., x_n]$  of degree at most *s*, there exists a unique multilinear polynomial  $\hat{f} \in \mathbb{F}[x_1, ..., x_n]$  of degree at most *s* such that  $f(x) = \hat{f}(x)$  for all  $x \in \{0, 1\}^n$ .

**Proof.** Expand f and use the identity  $x_i^2 = x_i$ , which is valid over  $\{0, 1\}^n$ .

#### **3.2.2 Proof of the Frankl-Wilson Theorem**

We present Babai's [4] elegant proof of Theorem 3.2.2. We first define the notion of the characteristic vector of a set.

**Definition 3.2.7.** The <u>characteristic vector</u> of a set  $F \subset X$ , denoted  $v_F \in \{0,1\}^n$ , is defined by

$$(v_F)_i := \begin{cases} 1 & if \quad i \in F \\ 0 & if \quad i \notin F. \end{cases}$$

**Proof of Theorem 3.2.2** Let  $\mathscr{F} = \{F_1, \ldots, F_m\}$ , where  $|F_1| \leq \cdots \leq |F_m|$ , and suppose  $L = \{l_1, \ldots, l_s\}$ . With each set  $F_i \in \mathscr{F}$ , associate its characteristic vector  $v_i \in \mathbb{R}^n$ . Note that  $v_i \cdot v_j = |F_i \cap F_j|$ . For  $i \in [m]$ , define the polynomial  $f_i \in \mathbb{R}[x_1, \ldots, x_n]$  by

$$f_i(x) := \prod_{k: l_k < |F_i|} (v_i \cdot x - l_k).$$
(3.2.4)

Each polynomial  $f_i$  has degree at most *s* because |L| = s. Since  $\mathscr{F}$  is *L*-intersecting and  $|F_1| \leq \cdots \leq |F_m|$ , we have

$$f_{i}(v_{j}) = \prod_{k:l_{k} < |F_{i}|} (v_{i} \cdot v_{j} - l_{k}) = \prod_{k:l_{k} < |F_{i}|} (|F_{i} \cap F_{j}| - l_{k})$$

$$\begin{cases}
\neq 0 & \text{if } i = j \\
= 0 & \text{if } j < i.
\end{cases}$$
(3.2.5)

We now use Lemma 3.2.6 to replace each  $f_i$  by the unique multilinear polynomial  $\hat{f}_i$  of degree at most *s* such that  $f_i(x) = \hat{f}_i(x)$  for all  $x \in \{0,1\}^n$ . Since  $v_i \in \{0,1\}^n$ , the polynomials  $\hat{f}_1, \ldots, \hat{f}_m$  and the elements  $v_1, \ldots, v_m \in \{0,1\}^n$  satisfy (3.2.5); that is they satisfy the Triangular Criterion, Lemma 3.2.5. Hence,  $\hat{f}_1, \ldots, \hat{f}_m$  are linearly independent members of *W*, the space of multilinear polynomials of degree at most *s* and with coefficients in  $\mathbb{R}$ . By Lemma 3.2.4, we conclude that

$$|\mathscr{F}| = m \le \sum_{i=0}^{s} \binom{n}{i}.$$

If  $\lambda \in \mathbb{Z}^+$  and  $\mathscr{F} \subset 2^X$  is  $\lambda$ -intersecting, then Theorem 3.2.2 gives  $m \leq n+1$ , which is weaker than Fisher's Inequality. However, the proof can easily be modified to give the correct bound. If  $\lambda \in \mathbb{Z}^+$ , then  $\emptyset \notin \mathscr{F}$ . Hence, in (3.2.4),  $f_i(x) = v_i \cdot x - \lambda$  is a multilinear polynomial of degree 1. By Lemma 3.2.3, we conclude that  $m \leq n$ .

## **3.3** Eigenvalues

We give Godsil's and Newman's [66] proof of the Erdős-Ko-Rado theorem via eigenvalues of the Kneser graph, which is closely related to that of Wilson [116]. Variants of this method have yielded Erdős-Ko-Rado analogs for vector spaces [60] and recently, in combination with Fourier analysis, for permutations [46, 64] and graphs [45].

### 3.3.1 Independent Sets

We begin with basic graph theory terminology, and then prove the ratio bound on independent sets [35]. **Definition 3.3.1.** A graph is an ordered pair G = (V, E) comprising a set V of vertices together with a set  $E \subset {V \choose 2}$  of edges.

**Definition 3.3.2.** *Let* G = (V, E) *be a graph. Two vertices*  $v, w \in V$  *are <u>adjacent</u>, denoted*  $v \sim w$ , *if*  $\{v, w\} \in E$ .

**Definition 3.3.3.** Let G = (V, E) be a graph. A subset  $S \subset V(G)$  is <u>independent</u> if no two vertices in S are adjacent.

**Definition 3.3.4.** If G = (V, E) is a graph, then its associated <u>adjacency matrix</u> A is defined to be the  $|V| \times |V|$  matrix with

$$A_{v,w} := \begin{cases} 1 & if \quad v \sim w \\ 0 & if \quad v \not\sim w. \end{cases}$$

**Definition 3.3.5.** If G = (V, E) is a graph, then the <u>degree</u> of a vertex  $v \in G$ , denoted  $\deg(v)$ , is the number of vertices adjacent to v; that is,  $\deg(v) := |\{w : v \sim w\}|$ .

**Definition 3.3.6.** A graph G = (V, E) is <u>k-regular</u> if all vertices  $v \in V$  have degree deg(v) = k.

The ratio bound relates the size of the largest independent set in a regular graph to the least eigenvalue of its adjacency matrix. The original proof by Delsarte uses linear programming techniques, but we will follow Godsil's approach [65, Lemma 9.6.2].

**Theorem 3.3.7** (Delsarte). Let G = (V, E) be a k-regular graph with v vertices and suppose the adjacency matrix A of G has least eigenvalue  $\tau$ . Let  $S \subset V$  be an independent set in G with characteristic vector z. Then

$$|S| \le \frac{v}{1 + \frac{k}{-\tau}};$$

if equality holds then

$$(A-\tau I)\left(z-\frac{|S|}{v}\mathbf{1}\right)=0.$$

**Proof.** Let *M* be the matrix given by

$$M:=A-\tau I-\frac{k-\tau}{v}J.$$

We claim the matrix M is positive semidefinite. It suffices to show that the eigenvalues of M are nonnegative. First observe that, since G is k-regular, the all ones vector,  $\mathbf{1}$ , is an eigenvector of M corresponding to the zero eigenvalue. Let  $\lambda \neq 0$  be an eigenvalue of M distinct from zero. Since M is symmetric, any eigenvector  $f_{\lambda} \in \mathbb{R}^{\nu}$  corresponding to  $\lambda$  must be orthogonal to  $\mathbf{1}$ . Hence,  $Jf_{\lambda} = 0$  so  $f_{\lambda}$  is an eigenvector of  $A - \tau I$ . As  $\tau$  is the least eigenvalue of A, we see that  $A - \tau I$  has nonnegative eigenvalues so  $\lambda \geq 0$ . This proves that M is positive semidefinite.

Let  $S \subset V$  be an independent set in G with characteristic vector z. As M is positive semidefinite, we see

$$0 \leq z^T M z = z^T A z - \tau z^T z - \frac{k - \tau}{v} z^T J z = z^T A z - \tau |S| - \frac{k - \tau}{v} |S|^2.$$

Since *S* is independent, we have  $z^T A z = 0$ , and hence

$$0 \le -\tau |S| - \frac{k - \tau}{\nu} |S|^2 \quad \Rightarrow \quad |S| \le \frac{\nu}{1 + \frac{k}{-\tau}}.$$
(3.3.1)

This yields the bound of the theorem.

If equality holds in (3.3.1), then  $z^T M z = 0$ . Since *M* is positive semidefinite, this implies M z = 0, and accordingly

$$(A - \tau I)z = \frac{k - \tau}{v}Jz = \frac{k - \tau}{v}|S|\mathbf{1}.$$

We also have  $(A - \tau I)\mathbf{1} = (k - \tau)\mathbf{1}$ , which yields the second claim.

## 3.3.2 Erdős-Ko-Rado via Eigenvalues

Godsil's and Newman's proof [66] of the Erdős-Ko-Rado theorem applies the ratio bound, Theorem 3.3.7, to the Kneser graph.

**Definition 3.3.8.** The Kneser graph, denoted  $K_{n:k}$ , has vertex set  $V = {X \choose k}$  and edge set  $E = \{\{A, B\} : A, B \in {X \choose k}, A \cap B = \emptyset\}.$ 

Observe that independent sets in the Kneser graph  $K_{n:k}$  are in bijective correspondence with intersecting families  $\mathscr{F} \subset {X \choose k}$ . Consequently, we can apply the ratio bound, Theorem 3.3.7, to the Kneser graph to prove the Erdős-Ko-Rado theorem. Fortunately, the eigenvalues and their corresponding multiplicities have been computed for the Kneser graph; a derivation can be found in [65, Section 9.4].

**Theorem 3.3.9.** Let  $n \ge 2k$  and let A denote the adjacency matrix of the Kneser graph  $K_{n:k}$ . The eigenvalues  $\lambda_i$  of A and their corresponding multiplicities  $m_i$  are

$$\lambda_i = (-1)^i \binom{n-k-i}{k-i}, \ m_i = \binom{n}{i} - \binom{n}{i-1}, \ i = 0, \dots, k$$

We now give Godsil's and Newman's proof [66] of the Erdős-Ko-Rado theorem; we will prove only the bound though it is possible to use these methods to characterize the case of equality.

**Proof of Theorem 2.2.2.** Suppose  $n \ge 2k$ . Let  $\mathscr{F} \subset {X \choose k}$  be a maximum intersecting family, and let  $S \subset V$  be the independent set in the Kneser graph  $K_{n:k}$  corresponding to  $\mathscr{F}$ . By Theorem 3.3.9, the least eigenvalue of the Kneser graph  $K_{n:k}$  is

$$\tau = \lambda_1 = -\binom{n-k-1}{k-1}.$$

Observe that the Kneser graph  $K_{n:k}$  is  $\binom{n-k}{k}$ -regular since given any *k*-set, there are  $\binom{n-k}{k}$  *k*-sets disjoint from it. Hence, the ratio bound, Theorem 3.3.7, yields

$$|\mathscr{F}| = |S| \le \frac{\binom{n}{k}}{1 + \frac{\binom{n-k}{k}}{\binom{n-k-1}{k-1}}} = \binom{n-1}{k-1},$$

which establishes the bound in Theorem 2.2.2. Using the second claim of Theorem 3.3.7 and the multiplicity  $m_1 = n - 1$  of the least eigenvalue  $\lambda_1$ , it is possible to characterize the case of equality when n > 2k; see [94, Section 5.4] for details.

# 3.4 Linear Programming

We now discuss linear programming and some of its applications to extremal set theory. In Section 3.4.1, we state and prove the weak versions of the Duality and Complementary Slackness theorems; these results are used in Chapter 6 and provide motivation for the strong versions of these theorems [61, 112], which we will state but not prove. In Section 3.4.3, we state the Hoffman-Kruskal theorem [71] on totally unimodular matrices, and use it to prove the integrality theorem on flows [32] in Section 3.4.4. Finally, we use the flow result in Section 3.4.5 to prove Baranyai's theorem [10] on decompositions of the complete hypergraph into perfect matchings. We will need Baranyai's theorem in Chapter 5 and Chapter 7.

#### 3.4.1 Duality

We will prove the weak versions of the Duality and Complementary Slackness theorems for the equality form, which we will use in Chapter 6. The strong versions of these theorems are much more difficult to prove, so we will only state them as they will not be needed for any of the author's results in this thesis.

Consider the problem

Minimize 
$$\sum_{j=1}^{n} c_{j} x_{j} \qquad (3.4.1)$$
subject to: 
$$\sum_{j=1}^{n} a_{ij} x_{j} = b_{i} \quad i \in [m]$$

$$x_{j} \ge 0 \quad j \in [n],$$

where all coefficients and variables take real values. Let us fix some terminology.

**Definition 3.4.1.** In (3.4.1), the function to be optimized is called the <u>objective function</u>. **Definition 3.4.2.** In (3.4.1), the inequalities and equations to be satisfied are called the constraints.

**Definition 3.4.3.** A feasible solution  $x \in \mathbb{R}^n$  is a point that satisfies all the constraints. **Definition 3.4.4.** The feasible region is the set of all feasible solutions.

**Definition 3.4.5.** An optimal solution of the minimization problem (3.4.1) is a feasible solution  $(\hat{x}_1, ..., \hat{x}_n) \in \mathbb{R}^n$  such that  $\sum_{j=1}^n c_j \hat{x}_j \leq \sum_{j=1}^n c_j x_j$  for any feasible solution  $(x_1, ..., x_n) \in \mathbb{R}^n$ .

We seek a simple way to determine whether a feasible point  $(x_1, ..., x_n) \in \mathbb{R}^n$ of the minimization problem (3.4.1) is optimal. This suggests the idea of finding lower bounds for the minimum. Observe that if we take a real multiple  $y_i$  of each equality  $a_{ij}x_j = b_i$  in (3.4.1) and add these equalities so that the resulting equality  $\sum_{j=1}^n d_j x_j = d_0$ satisfies  $d_j \leq c_j$ , then  $d_0$  is a lower bound for  $\sum_{j=1}^n c_j x_j$ . The best lower bound obtainable in this way is given by

Maximize 
$$\sum_{i=1}^{m} b_i y_i \qquad (3.4.2)$$
  
subject to: 
$$\sum_{i=1}^{m} a_{ij} y_i \le c_j \quad j \in [n].$$

We call (3.4.2), the *dual* of (3.4.1) and refer to (3.4.1) as the *primal problem*. It is a consequence of the way we constructed the dual that every feasible solution  $y \in \mathbb{R}^m$  of the dual (3.4.2) gives a lower bound for the objective value of (3.4.1). This is the content of the weak duality theorem, which we now formally state and prove.

**Theorem 3.4.6** (Weak Duality). If  $x \in \mathbb{R}^n$  is a feasible solution of the primal problem (3.4.1) and  $y \in \mathbb{R}^m$  is a feasible solution of the dual problem (3.4.2), then

$$\sum_{j=1}^n c_j x_j \ge \sum_{i=1}^m b_i y_i.$$

**Proof.** The constraints in (3.4.1) and (3.4.2) yield that

$$\sum_{j=1}^{n} c_j x_j \ge \sum_{j=1}^{n} \left( \sum_{i=1}^{m} a_{ij} y_i \right) x_j = \sum_{i=1}^{m} y_i \left( \sum_{j=1}^{n} a_{ij} x_j \right) = \sum_{i=1}^{m} b_i y_i, \quad (3.4.3)$$

which proves the theorem.

The weak duality theorem has several useful consequences.

**Corollary 3.4.7.** If  $\hat{x} \in \mathbb{R}^n$  is a feasible solution of the primal problem (3.4.1),  $\hat{y} \in \mathbb{R}^m$  is a feasible solution of the dual problem (3.4.2), and  $\sum_{j=1}^n c_j \hat{x}_j = \sum_{i=1}^m b_i \hat{y}_i$ , then  $\hat{x}$  and  $\hat{y}$  are optimal solutions of the primal and dual problems respectively.

**Proof.** Since  $\sum_{j=1}^{n} c_j x_j$  cannot be smaller than  $\sum_{i=1}^{m} b_i \hat{y}_i$  for any feasible  $x \in \mathbb{R}^n$ , and since  $\hat{x}$  achieves this bound,  $\hat{x}$  is optimal. The same argument can be used for  $\hat{y}$ . By rewriting (3.4.3), we obtain the weak complementary slackness theorem.

**Theorem 3.4.8** (Weak Complementary Slackness). If  $\hat{x} \in \mathbb{R}^n$  is a feasible solution of the primal problem (3.4.1),  $\hat{y} \in \mathbb{R}^m$  is a feasible solution of the dual problem (3.4.2), and  $\sum_{j=1}^n c_j \hat{x}_j = \sum_{i=1}^m b_i \hat{y}_i$ , then for each  $j \in [n]$ ,

$$\sum_{i=1}^{m} a_{ij} \hat{y}_i = c_j \text{ or } \hat{x}_j = 0.$$
(3.4.4)

**Proof.** If  $\sum_{j=1}^{n} c_j \hat{x}_j = \sum_{i=1}^{m} b_i \hat{y}_i$ , then equality holds everywhere in (3.4.3). In particular, the first equality is equivalent to

$$\sum_{j=1}^{n} \left( \sum_{i=1}^{m} c_j - a_{ij} \hat{y}_i \right) \hat{x}_j = 0.$$
(3.4.5)

Since all terms in this sum are nonnegative, we must have (3.4.4) holds.

The weak duality theorem, Theorem 3.4.6, gives a sufficient condition for a feasible point  $\hat{x} \in \mathbb{R}^n$  of (3.4.1) to be optimal, namely the existence of  $\hat{y} \in \mathbb{R}^m$  satisfying (3.4.3) with equality. It is much less obvious that this condition is also necessary, which is the content of the strong duality theorem.

**Theorem 3.4.9** (Strong Duality). *If the primal problem* (3.4.1) *has an optimal solution*  $\hat{x} \in \mathbb{R}^n$ , then the dual problem (3.4.2) has an optimal solution  $\hat{y} \in \mathbb{R}^m$  and  $\hat{x}, \hat{y}$  satisfy the relation  $\sum_{j=1}^n c_j \hat{x}_j = \sum_{i=1}^m b_i \hat{y}_i$ .

The strong duality theorem implies the strong complementary slackness theorem.

**Theorem 3.4.10** (Strong Complementary Slackness). *Feasible solutions*  $\hat{x} \in \mathbb{R}^n$  *and*  $\hat{y} \in \mathbb{R}^m$  of the primal and dual problems respectively are optimal if and only if  $\hat{x}_j = 0$  or  $\sum_{i=1}^m a_{ij}\hat{y}_i = c_j$  for  $j \in [n]$ .

All results in this section are due to von Neumann [112] and Gale, Kuhn, and Tucker [61]. The proofs of Theorem 3.4.9 and Theorem 3.4.10 can be found in any good linear programming textbook such as [30, Section 2.5].

### 3.4.2 The Fundamental Theorem of Linear Programming

We stated the duality and complementary slackness theorems in the equality form because these are the results we need in Chapter 6. More generally, a linear program has the form

Maximize 
$$c^T x$$
 (3.4.6)  
subject to:  $Ax \le b$ ,

where  $c, x \in \mathbb{R}^n$ , *A* is an  $n \times m$  matrix with real entries, and  $b \in \mathbb{R}^m$ . Note that (3.4.1) can be put into the form in (3.4.6) in the following way,

Maximize 
$$-c^T x$$
  
subject to:  $\begin{bmatrix} A \\ -A \\ -I \end{bmatrix} x \le \begin{bmatrix} b \\ -b \\ 0 \end{bmatrix}$ ,

where  $c, x \in \mathbb{R}^n$ , *A* is an  $n \times m$  matrix with real entries  $A_{ij} = a_{ij}$ , *I* is the  $m \times m$  identity matrix, and  $b \in \mathbb{R}^m$ .

The Fundamental Theorem of Linear Programming asserts that any linear program is exactly one of three types, and is proved in [30, Theorem 2.12]. We again fix some terminology.

**Definition 3.4.11.** *The linear program* (3.4.6) *is* <u>infeasible</u> *if there does not exist*  $x \in \mathbb{R}^n$  *such that*  $Ax \leq b$ .

**Definition 3.4.12.** *The linear program* (3.4.6) *is <u>unbounded</u> if for each*  $r \in \mathbb{R}$ *, there exists feasible*  $x \in \mathbb{R}^n$  *such that*  $c^T x \ge r$ *.* 

**Theorem 3.4.13** (Fundamental Theorem of Linear Programming). *The linear program* (3.4.6) *is infeasible, unbounded, or has an optimal solution.* 

#### 3.4.3 Integer Programming

In practical applications, we would often like to know if a linear program has an optimal solution  $\hat{x} \in \mathbb{Z}^n$  all of whose coordinates are integers. We describe a method to find such an optimal solution. We begin with some terminology.

**Definition 3.4.14.** A polyhedron is the feasible region of the linear program (3.4.6).

**Definition 3.4.15.** A region  $C \subset \mathbb{R}^n$  is <u>convex</u> if whenever  $x, y \in C$  and  $\lambda \in [0, 1]$ , we also have  $\lambda x + (1 - \lambda)y \in C$ . In other words, C is convex if for any two points  $x, y \in C$ , the line segment joining x and y is also contained in C.

**Definition 3.4.16.** An <u>extreme point</u> of a convex set  $C \subset \mathbb{R}^n$  is a point  $p \in C$  such that there do not exist distinct points  $q, r \in C$  and  $\lambda \in (0,1)$  such that  $p = \lambda q + (1 - \lambda)r$ . In other words, an extreme point of C is a point of C that is not in the interior of any line segment contained in C.

**Definition 3.4.17.** A set  $L \subset \mathbb{R}^n$  is a <u>line</u> if there exist distinct  $x_1, x_2 \in \mathbb{R}^n$  such that  $L = \{\lambda x_1 + (1 - \lambda) x_2 : \lambda \in \mathbb{R}\}.$ 

Polyhedrons are convex, and the following result illustrates why their extreme points are important in linear programming; a proof is in [30, Theorem 2.27].

**Theorem 3.4.18.** If a linear program (3.4.6) has an optimal solution and its feasible region contains no line, then it has an optimal solution that is an extreme point of its feasible region.

Combined with Theorem 3.4.18, the Hoffman-Kruskal theorem [71] yields a way of finding an integral optimal solution.

**Definition 3.4.19.** A matrix M with real entries  $M_{ij}$  is totally unimodular if every square submatrix N of M has determinant 0, 1, or -1.

**Theorem 3.4.20** (Hoffman-Kruskal). If A is a totally unimodular  $n \times m$  matrix and  $b \in \mathbb{Z}^m$ , then every extreme point of the polyhedron  $P = \{x \in \mathbb{R}^n : Ax \leq b\}$  determined by (3.4.6) has integral coordinates.

Suppose we know that the  $n \times m$  matrix A is totally unimodular,  $b \in \mathbb{Z}^m$ , and that the polyhedron  $P = \{x \in \mathbb{R}^n : Ax \le b\}$  determined by (3.4.6) has an optimal solution and does not contain a line. The Hoffman-Kruskal theorem asserts that all extreme points of the polyhedron P have integral coordinates, and Theorem 3.4.18 proves that one of those integral extreme points must be an optimal solution to the linear program (3.4.6).

#### **3.4.4** Flows

We define the notion of a transportation network, and prove the integrality theorem on flows [32] using Hoffman's and Kruskal's Theorem 3.4.20.

**Definition 3.4.21.** A <u>digraph</u> is an ordered pair D = (V,A) comprising a set V of vertices together with a set  $A \subset V \times V$  of ordered pairs of vertices called arcs.

**Definition 3.4.22.** If a = (v, w) is an arc in the digraph D = (V, A), then a is said to be directed from v to w. We call v the <u>tail</u> of a and w the <u>head</u>.

**Definition 3.4.23.** A <u>transportation network</u> is a finite digraph D = (V,A) together with two distinguished vertices called the source s and the sink t, and a capacity function  $k : A \to \mathbb{R}_{\geq 0}$  which associates a nonnegative real number k(a) to each arc  $a \in A$ . The source s must be the tail of all arcs which contain it and the sink t must be the head of all arcs which contain it. We further assume that A does not contain any arcs of the form a = (v, v) for a vertex  $v \in V$ . **Definition 3.4.24.** *Given a transportation network* D = (V,A)*, a <u>flow</u> in D is a function*  $f : A \to \mathbb{R}_{\geq 0}$  which assigns to each arc  $a \in A$  a nonnegative real number f(a) such that

- (i)  $0 \le f(a) \le k(a)$  for all arcs  $a \in A$  and
- (ii) for each vertex  $v \in V \setminus \{s,t\}$  we have

$$\sum_{a \in A: v \text{ is head of } a} f(a) = \sum_{a \in A: v \text{ is tail of } a} f(a).$$

**Definition 3.4.25.** If D = (V,A) is a transportation network and  $f : A \to \mathbb{R}_{\geq 0}$  is a flow on *D* then the <u>strength</u> of the flow, denoted |f|, is defined to be the sum of the values of *f* on the arcs leaving *s*,

$$|f| := \sum_{a \in A: s \in a} f(a).$$

We will be interested in determining the maximum strength of a flow on a transportation network, and we will show how to formulate this problem as a linear program. In the following diagram, we illustrate an example of a transportation network with a maximum flow f of strength |f| = 5.



Figure 3.1: A transportation network with flow and capacity denoted f/c.

To formulate the maximum flow problem as a linear program, we need to define the concept of an incidence matrix of a digraph.

**Definition 3.4.26.** If D = (V,A) is a digraph, then its <u>incidence matrix</u> M is defined to be the  $|V| \times |A|$  matrix with

$$M_{v,a} := \begin{cases} -1 & \text{if } v \text{ is tail of } a \\ 1 & \text{if } v \text{ is head of } a \\ 0 & \text{if } v \notin a. \end{cases}$$

We show that the incidence matrix of a digraph is totally unimodular.

#### Lemma 3.4.27. The incidence matrix M of a digraph D is totally unimodular.

**Proof.** We must show that every square submatrix *N* of *M* satisfies  $det(N) \in \{-1,0,1\}$ . We prove this by induction on the order *l* of *N*. Certainly, the statement is true for l = 1 as the entries of *M* lie in  $\{-1,0,1\}$ . Suppose  $l \ge 2$  and all square submatrices of order less than *l* have determinant 0 or  $\pm 1$ . Observe that every column of *M* has exactly one +1 entry and one -1 entry. Now, if *N* has a zero column, then det(N) = 0. If *N* has a column with exactly one  $\pm 1$  entry, then we can expand its determinant on that column, so  $det(N) = \pm det(N')$ , where *N'* has order l - 1. By the induction hypothesis, we consequently have  $det(N) = \pm det(N') \in \{-1,0,1\}$ . Finally, if every column of *N* has exactly one +1 entry and one -1 entry, then its rows sum to zero, so det(N) = 0. This proves that *M* is totally unimodular.

Finally, we formulate the maximum flow problem as a linear program.

**Lemma 3.4.28** (Maximum Flow Problem). Let D = (V,A) be a transportation network. Let M denote the incidence matrix of D and let  $c^T \in \mathbb{R}^{|A|}$  denote the row in M corresponding to s. Deleting the rows in M corresponding to the source s and the sink t yields  $a(|V|-2) \times |A|$  matrix, which we denote  $\overline{M}$ . Let  $k \in \mathbb{R}^{|A|}$  denote the capacity vector with entries  $k_a = k(a)$ . Finally, let  $I_w$  denote the  $w \times w$  identity matrix and let  $\vec{0}_w \in \mathbb{R}^w$  denote the all zero vector. The maximum strength of a flow on D is the optimal value of the following linear program,

Maximize 
$$-c^{T}x$$
 (3.4.7)  
subject to: 
$$\begin{bmatrix} \overline{M} \\ -\overline{M} \\ I_{|A|} \\ -I_{|A|} \end{bmatrix} x \leq \begin{bmatrix} \vec{0}_{|V|-2} \\ \vec{0}_{|V|-2} \\ k \\ \vec{0}_{|A|} \end{bmatrix}.$$

We now state and prove the integrality theorem on flows [32].

**Theorem 3.4.29** (Dantzig). If D = (V,A) is a transportation network with an integral capacity vector  $k \in \mathbb{Z}^{|A|}$ , then there is a maximum strength flow f on D such that for each  $a \in A$ , the value  $f(a) \in \mathbb{Z}$  is integral.

**Proof.** By Lemma 3.4.27, the incidence matrix M of D is totally unimodular. Hence,  $\overline{M}$ , the matrix formed by removing the rows corresponding to the source s and the sink t, is totally unimodular. A similar argument to the one in Lemma 3.4.27 proves that if a matrix L is totally unimodular, then so are the matrices  $L^T$ , [L|-L], [L|I], and [L|-I]. Hence, the matrix on the left hand side of (3.4.7) is totally unimodular. As  $k \in \mathbb{Z}^{|A|}$  is integral, the vector on the right hand side of (3.4.7) is integral. The Hoffman-Kruskal theorem, Theorem 3.4.20, yields that the extreme points of the polyhedron defined by (3.4.7) all have integral coordinates.

Observe that the linear program in (3.4.7) is not infeasible or unbounded because  $\vec{0}_{|A|}$  is a feasible solution and the value of the objective function is bounded by  $-c^T k$ . Hence, (3.4.7) has an optimal solution by the Fundamental Theorem of Linear Programming, Theorem 3.4.13. Moreover, observe that the polyhedron determined by (3.4.7) does not contain a line since for each  $a \in A$ , we have  $0 \le x_a \le k(a)$ . Hence, by Theorem 3.4.18, one of the integral extreme points must be an optimal solution of the linear program (3.4.7). Consequently, there exists a maximum strength flow f on Dsuch that for each  $a \in A$ , the value  $f(a) \in \mathbb{Z}$  is integral by Lemma 3.4.28.

For more information on flows, see [110, Chapter 7]. We will use the integrality theorem on flows, Theorem 3.4.29, to prove Baranyai's theorem.

#### 3.4.5 Baranyai's Theorem

We prove Baranyai's theorem [10] on decompositions of the complete hypergraph into perfect matchings. This result will be needed in Chapter 5 and Chapter 7.

**Definition 3.4.30.** A <u>partition</u> (respectively <u>m-partition</u>) of a set X is a family (respectively multiset)  $\mathcal{P}$  of subsets of X that satisfies properties (i)-(iii) (respectively (ii)-(iv)):

- (i)  $\emptyset \notin \mathscr{P}$ ,
- (ii)  $P \cap P' = \emptyset$  for all distinct  $P, P' \in \mathscr{P}$ ,
- (*iii*)  $\bigcup_{P \in \mathscr{P}} P = X$ ,
- (*iv*)  $|\mathscr{P}| = m$ .

**Definition 3.4.32.** If k|n, then a <u>perfect matching</u> of the complete hypergraph  $\binom{X}{k}$  is a family  $\mathscr{P} \subset \binom{X}{k}$  that is an (n/k)-partition of X.

Baranyai's theorem asserts that if k|n, then the complete hypergraph  $\binom{X}{k}$  can be partitioned into perfect matchings. It is straightforward to verify that Baranyai's theorem is true for k = 2, but the case k = 3 is much more difficult [95]. We now formally state Baranyai's theorem and give a proof due to A.E. Brouwer and A. Schrijver [22]. All known proofs of Baranyai's theorem use some form or consequence of Theorem 3.4.29.

**Theorem 3.4.33** (Baranyai). If k|n, then the complete hypergraph  $\binom{X}{k}$  can be partitioned into  $\binom{n-1}{k-1}$  perfect matchings.

**Proof.** We prove a seemingly stronger statement. Let m := n/k and let  $M := \binom{n-1}{k-1}$ . We assert that for any nonnegative integer l with  $l \le n$ , there exists a family  $\mathscr{A}_1, \ldots, \mathscr{A}_M$  of m-partitions of [l] such that each subset  $S \subset [l]$  appears exactly  $\binom{n-l}{k-|S|}$  times among the m partitions  $\mathscr{A}_i$ . Observe that the case l = n proves the theorem as then

$$\binom{n-l}{k-|S|} = \binom{0}{k-|S|} = \begin{cases} 1 & \text{if } k = |S| \\ 0 & \text{otherwise.} \end{cases}$$

We proceed by induction on *l*. The base case l = 0 is trivially true since each  $\mathscr{A}_i$  consists of *m* copies of the empty set. Assume that, for some l < n, a family of *m*-partitions  $\mathscr{A}_1, \ldots, \mathscr{A}_M$  with the desired properties exist. We form a transportation network with source *s*, sink *t*, vertices labeled  $\mathscr{A}_i$  for  $i \in [M]$ , and vertices labeled *S* for each subset  $S \subset [l]$ . There is an arc with capacity 1 from the source *s* to each vertex labeled  $\mathscr{A}_i$  to the vertex labeled *S*; if  $S = \emptyset$  we put *j* arcs with capacity 1 from the vertex labeled  $\mathscr{A}_i$  to the vertex labeled  $\emptyset$  if the empty set occurs *j* times in  $\mathscr{A}_i$ . Finally, there is an arc from each vertex labeled *S* to the sink *t* with capacity  $\binom{n-l-1}{k-|S|-1}$ .

We exhibit a flow in this network. Assign a flow value of 1 to the arcs leaving the source *s*. For an arc from a vertex labeled  $\mathcal{A}_i$  to a vertex labeled *S*, assign the flow value (k - |S|)/(n - l). For an arc from a vertex labeled *S* to the sink *t*, assign the flow

value  $\binom{n-l-1}{k-|S|-1}$ . Clearly property (i) of a flow is satisfied so we verify property (ii). The sum of the flow values on arcs leaving a vertex labeled  $\mathscr{A}_i$  is

$$\sum_{S \in \mathscr{A}_i} \frac{k - |S|}{n - l} = \frac{1}{n - l} \left( mk - \sum_{S \in \mathscr{A}_i} |S| \right) = \frac{1}{n - l} (mk - l) = 1.$$

The sum of the flow values on the arcs entering a vertex labeled S is

$$\sum_{i:S\in\mathscr{A}_{i}}\frac{k-|S|}{n-l} = \frac{k-|S|}{n-l}\binom{n-l}{k-|S|} = \binom{n-l-1}{k-|S|-1}.$$

For each arc *a* leaving the sink *s*, the flow value equals the capacity f(a) = k(a), so this is a maximum flow with strength *M*. The same property holds for arcs entering into the sink *t*. Hence, for any maximum flow on this network, the flow value equals the capacity on any arc leaving the source *s* or entering the sink *t*.

Since all arcs have integral capacities, Theorem 3.4.29 yields an *integral*-valued maximum flow  $\hat{f}$ . As  $\hat{f}$  is maximum, for any arc a from s to a vertex labeled  $\mathscr{A}_i$ , we have f(a) = k(a) = 1. As  $\hat{f}$  is integral and all arcs leaving the vertex labeled  $\mathscr{A}_i$  have capacity 1, we see that  $\hat{f}$  sends one unit of flow from the vertex labeled  $\mathscr{A}_i$  to exactly one vertex labeled  $S_i$ , where  $S_i \in \mathscr{A}_i$ . All other arcs leaving the vertex labeled  $\mathscr{A}_i$  will have zero flow. As  $\hat{f}$  is maximum, for any arc a leaving a vertex labeled S to the sink t, we have  $f(a) = k(a) = \binom{n-l-1}{k-|S|-1}$ . Consequently, we see that for each set S, the number of i such that  $S_i = S$  is  $\binom{n-l-1}{k-|S|-1}$ .

We obtain a family of *m*-partitions  $\mathscr{A}'_1, \ldots, \mathscr{A}'_M$  of the set [l+1] by letting  $\mathscr{A}'_i$  be obtained from  $\mathscr{A}_i$  by replacing the distinguished set  $S_i$  by  $S_i \cup \{l+1\}$  for  $i \in [M]$ . We claim that  $T \subset [l+1]$  appears exactly  $\binom{n-(l+1)}{k-|T|}$  times among  $\mathscr{A}'_1, \ldots, \mathscr{A}'_M$ . This is clear if  $T = S \cup \{l+1\}$  as the number of times S is chosen to be  $S_i$  is  $\binom{n-l-1}{k-|S|-1} = \binom{n-(l+1)}{k-|T|}$ . Otherwise,  $T \subset [l]$ . Since T appears  $\binom{n-l}{k-|T|}$  times among  $\mathscr{A}_1, \ldots, \mathscr{A}_M$  and the number of times l+1 is added to T is  $\binom{n-l-1}{k-|T|-1}$ , we see that T appears

$$\binom{n-l}{k-|T|} - \binom{n-l-1}{k-|T|-1} = \binom{n-(l+1)}{k-|T|}$$

times among  $\mathscr{A}'_1, \ldots, \mathscr{A}'_M$ . This completes the induction step.

# Chapter 4

# **Projective Prerequisites**

We saw in Chapter 2 and Chapter 3 that intersecting families and shadows are two core concepts in extremal set theory. We also proved and discussed fundamental results about these concepts, such as the Erdős-Ko-Rado and Kruskal-Katona theorems. By defining suitable notions of "intersecting" and "shadow," one can find remarkable analogs of these theorems for other structures such as vector spaces. A tantalizing feature is that, although results from extremal set theory are often expected to be true for vector spaces, not much is known about analogs because standard techniques do not always apply.

We describe what vector space analogs are in Section 4.1.1 and why this area of research is significant. To explain the analogies between sets and vector spaces, we then introduce a generalization of the binomial coefficients called the *q*-binomial coefficients in Section 4.2. The *q*-binomial coefficients satisfy identities which generalize familiar ones such as Pascal's rule; we explore these in Section 4.3 because we will need them for the analog of Lovász's result, Theorem 2.1.5, in Chapter 5. We prove the Erdős-Ko-Rado theorem for vector spaces and introduce the closely-related *q*-Kneser graphs in Section 4.4. In Chapter 5, we will extend the Erdős-Ko-Rado theorem for vector spaces by proving an analog of Frankl's *r*-wise intersection theorem, Theorem 2.2.7. To do this, we will need the concept of spreads, which are an analog of perfect matchings, so we discuss these in Section 4.5. Finally we end in Section 4.6 by highlighting the difficulties of adapting purely combinatorial techniques to vector spaces and dicuss algebraic methods that have worked in both the set and vector space settings.

## 4.1 Vector Space Analogs

In this section, we describe what vector spaces are and illustrate questions in this area by using the analog of the Erdős-Ko-Rado theorem as an example. We also discuss the significance of this area of study, and give an overview of which methods from Chapter 2 and Chapter 3 carry over to the vector space setting.

## 4.1.1 What are Vector Space Analogs?

In extremal set theory, our underlying set is the *n*-element set X. In vector space analogs, the set X is replaced by an *n*-dimensional vector space V over a finite field  $\mathbb{F}_q$ . The general question in extremal set theory concerns the maximum or minimum size of a family of subsets of X, which is usually k-uniform. In vector space analogs, the questions will usually concern families of k-dimensional subspaces of V.

To visualize the situation, consider the Fano plane in Figure 2.1. Here V is a three-dimensional vector space over the finite field  $\mathbb{F}_2$ . There are seven one-dimensional subspaces represented by points and seven two-dimensional subspaces represented by lines. In the picture, a point lies on a line if the one-dimensional subspace corresponding to the point lies in the two-dimensional subspace corresponding to the line.

We saw in Chapter 2 that the Erdős-Ko-Rado theorem asserts that if X is large enough, then the unique intersecting family  $\mathscr{F} \subset {X \choose k}$  of maximum size consists of the *k*-element subsets containing a fixed point. Let us try to formulate a vector space analog of the Erdős-Ko-Rado theorem. On a first try, we might ask what is the maximum size of a family  $\mathscr{F}$  of *k*-dimensional subspaces of V such that any two members of  $\mathscr{F}$  have nonempty intersection. However, this doesn't quite make sense since the zero subspace is always contained in the intersection of any two subspaces. We instead stipulate that the *dimension* of the intersection of any two members of  $\mathscr{F}$  is nonzero. Now, the vector space analog of Erdős-Ko-Rado asks for the maximum size of a family  $\mathscr{F}$  of *k*-dimensional subspaces of V such that the intersection of any two members of  $\mathscr{F}$  has nonzero dimension. The answer is strikingly similar to the original: if the dimension of V is large enough, then the unique intersecting family  $\mathscr{F} \subset {V \choose k}$  of maximum size consists of all *k*-dimensional subspaces containing a fixed one-dimensional subspace.

## 4.1.2 Significance

Remarkably, questions and results about intersecting families and shadows are valid not only for sets but for a wide range of mathematical objects such as vector spaces, permutations, and graphs. The ultimate objective of this research is to have a unified theory that characterizes the structures for which analogs exist and that proves results simultaneously for broad classes of objects. As we will see in the sections that follow, vector space analogs force us to analyze which methods from Chapter 2 and Chapter 3 work in a more general setting.

Vector space analogs bring new questions and techniques to finite geometry since many of its problems can be reformulated in these terms. They also provide applications for the *q*-analog identities studied by algebraic combinatorialists. Recently, coding theorists such as Vardy are studying vector space analogs because they imply results about projective codes [18, 49, 50]. Since codes are used in communication systems, research in this area may yield practical applications.

### 4.1.3 Methods

In Chapter 2 and Chapter 3, we gave four proofs of the Erdős-Ko-Rado theorem. Which of these approaches carries over to the vector space setting? In Section 4.6, we will discuss the difficulties that arise when we try to mimic the shifting proof or Daykin's proof from Section 2.2.3 and Section 2.2.4 respectively. We will see that we can mimic Katona's proof from Section 2.2.5 to yield the Erdős-Ko-Rado theorem for vector spaces in the special case that k|n; this argument will also be important for the analog of Frankl's *r*-wise intersection theorem in Chapter 5 and for the Manickam-Miklós-Singhi conjecture in Chapter 7. Finally, we will see in Section 4.4 that the eigenvalue proof from Section 3.3.2 fully carries over to the vector space setting. In general, algebraic techniques have been more successful than purely combinatorial ones for vector space analogs. We will discuss some other algebraic successes in Section 4.6. We remark that, in Chapter 5, we will give another proof of the Erdős-Ko-Rado theorem for vector spaces that is surprising because it is purely combinatorial, but does not require any tedious computations.

## 4.2 The *q*-binomial coefficients

Usually, q will denote the order of a finite field, but in this section, we will allow q to be a positive real. We define the q-binomial coefficient, a generalization of the binomial coefficient. When q is the order of the finite field  $\mathbb{F}_q$ , the q-binomial coefficients play the same role in the enumeration of subspaces of V that the binomial coefficients play in the enumeration of subsets of X.

**Definition 4.2.1.** *If*  $a \in \mathbb{R}$ ,  $q \in \mathbb{R}^+$ , and  $k \in \mathbb{Z}^+$ , define the Gaussian binomial coefficient by

$$\begin{bmatrix} a \\ k \end{bmatrix}_q := \prod_{0 \le i < k} \frac{q^{a-i} - 1}{q^{k-i} - 1}$$

When k = 1, we write  $[a]_q := \begin{bmatrix} a \\ 1 \end{bmatrix}_q$ .

**Definition 4.2.2.** *For*  $a \in \mathbb{N}$  *and*  $q \in \mathbb{R}^+$ *, define the q-torial function by* [0]! = 1 *and* 

$$[a]_q! := \prod_{j=1}^a [j]_q \text{ for } a \in \mathbb{Z}^+.$$

Observe that  $\binom{a}{k} = \lim_{q \to 1} \binom{a}{k}_q$  and that, when q = 1 and  $a \in \mathbb{Z}^+$ , we have  $[a]_1 = a$  and  $[a]_1! = a!$ . When  $a \in \mathbb{Z}^+$ , the Gaussian binomial coefficient takes the familiar form

$$\begin{bmatrix} a \\ k \end{bmatrix}_q = \frac{[a]_q!}{[k]_q![a-k]_q!}$$

In this chapter, V always denotes an *n*-dimensional vector space over the finite field  $\mathbb{F}_q$ . We now introduce the notation  $\begin{bmatrix} V \\ k \end{bmatrix}_q$ , which is analogous to  $\begin{pmatrix} X \\ k \end{pmatrix}$ .

**Definition 4.2.3.** If  $q \in \mathbb{Z}^+$  is the order of the finite field  $\mathbb{F}_q$  and  $k \in \mathbb{Z}^+$  is a positive integer, let  $\begin{bmatrix} V \\ k \end{bmatrix}_q := \{S \subset V : \dim(S) = k\}$  denote the family of all k-dimensional subspaces of V.

We know that the size of  $\binom{X}{k}$  equals the binomial coefficient  $\binom{n}{k}$ ; a simple counting argument shows that we similarly have that the size of  $\begin{bmatrix}V\\k\end{bmatrix}_q$  is  $\begin{bmatrix}n\\k\end{bmatrix}_q$ .

**Lemma 4.2.4.** We have  $\left| \begin{bmatrix} V \\ k \end{bmatrix} \right|_q = \begin{bmatrix} n \\ k \end{bmatrix}_q$ .
**Proof.** There are  $(q^n - 1)(q^n - q) \cdots (q^n - q^{k-1})$  ways to choose *k*-tuple independent vectors from *V*. Since a given *k*-space has  $(q^k - 1)(q^k - q) \cdots (q^k - q^{k-1})$  distinct ordered bases, the number of *k*-dimensional subspaces of *V* is

$$\frac{(q^n-1)(q^n-q)\cdots(q^n-q^{k-1})}{(q^k-1)(q^k-q)\cdots(q^k-q^{k-1})} = \frac{(q^n-1)(q^{n-1}-1)\cdots(q^{n-k+1}-1)}{(q^k-1)(q^{k-1}-1)\cdots(q-1)} = \begin{bmatrix} n \\ k \end{bmatrix}_q. \quad \Box$$

## 4.3 The *q*-Pascal Rule

The *q*-binomial coefficients satisfy identities which generalize familiar ones such as Pascal's rule. We discuss the *q*-Pascal rule, which we will need for our proof of the analog of Lovász's result, Theorem 2.1.5, in Chapter 5. The familiar Pascal's identity asserts that

$$\binom{n}{k} = \binom{n-1}{k-1} + \binom{n-1}{k} \text{ for } k \in [n-1].$$

$$(4.3.1)$$

Often, naively changing binomial coefficients in an identity to *q*-binomial coefficients yields a *q*-identity: for example if  $a \in \mathbb{Z}^+$ , then

$$\binom{a}{k} = \binom{a}{a-k}, \ \begin{bmatrix}a\\k\end{bmatrix}_q = \begin{bmatrix}a\\a-k\end{bmatrix}_q \text{ for } k \in [a]. \tag{4.3.2}$$

In the case of Pascal's identity, however, changing binomial coefficients to q-binomial coefficients does not give a q-Pascal identity. When  $q \neq 1$ , we have

$$\begin{bmatrix} 2\\1 \end{bmatrix}_q = 1 + q \neq 2 = \begin{bmatrix} 1\\0 \end{bmatrix}_q + \begin{bmatrix} 1\\1 \end{bmatrix}_q.$$

Interestingly, there are *two q*-Pascal identities as we will show in Lemma 4.3.1. We will first give an arithmetic proof that holds when  $a \in \mathbb{R}$ ,  $q \in \mathbb{R}^+$ , and  $k \in \mathbb{Z}^+$ . Our second proof is more conceptual, but only holds when  $a \in \mathbb{Z}^+$  and q is the order of a finite field.

**Lemma 4.3.1** (The *q*-Pascal Identities). *If*  $a \in \mathbb{R}$ ,  $q \in \mathbb{R}^+$ , and  $k \in \mathbb{Z}^+$ , then

$$\begin{bmatrix} a \\ k \end{bmatrix}_q = \begin{bmatrix} a-1 \\ k-1 \end{bmatrix}_q + q^k \begin{bmatrix} a-1 \\ k \end{bmatrix}_q = q^{a-k} \begin{bmatrix} a-1 \\ k-1 \end{bmatrix}_q + \begin{bmatrix} a-1 \\ k \end{bmatrix}_q.$$

### Proof. We have

$$\begin{split} \begin{bmatrix} a \\ k \end{bmatrix}_{q} &= \left(\frac{q^{a}-1}{q-1}\right) \left(\frac{(q^{a-1}-1)\cdots(q^{a-k+1}-1)}{(q^{2}-1)\cdots(q^{k}-1)}\right) \\ &= \left(\frac{q^{k}-1}{q-1} + q^{k}\frac{q^{a-k}-1}{q-1}\right) \left(\frac{(q^{a-1}-1)\cdots(q^{a-k+1}-1)}{(q^{2}-1)\cdots(q^{k}-1)}\right) \\ &= \frac{(q^{a-1}-1)\cdots(q^{a-k+1}-1)}{(q-1)\cdots(q^{k-1}-1)} + q^{k}\frac{(q^{a-1}-1)\cdots(q^{a-k+1}-1)}{(q^{2}-1)\cdots(q^{k}-1)} \\ &= \begin{bmatrix} a-1 \\ k-1 \end{bmatrix}_{q} + q^{k} \begin{bmatrix} a-1 \\ k \end{bmatrix}_{q} \end{split}$$

The second equality is proved in a similar manner.

From now on, q will be restricted to be the order of the finite field  $\mathbb{F}_q$ , and we will drop the subscript q. Now we give a second proof of Lemma 4.3.1 using Lemma 4.2.4, but note that  $a \in \mathbb{Z}^+$  must be a positive integer.

**Proof.** Since  $a \in \mathbb{Z}^+$ , let n = a and let H be an (a - 1)-dimensional subspace of V. For the second identity, we partition the k-dimensional subspaces of V into k-dimensional subspaces that are contained in H and k-dimensional subspaces that are not contained in H. By Lemma 4.2.4, there are  $\begin{bmatrix} a-1\\k \end{bmatrix}$  k-dimensional spaces that are contained in H. Note that if a k-dimensional subspace of V does not lie in H, then it must intersect H in a (k-1)-dimensional space. By Lemma 4.2.4, there are  $\begin{bmatrix} a-1\\k-1 \end{bmatrix}$  (k-1)-dimensional subspaces in H, each of which is contained in

$$\frac{q^{a}-q^{k-1}}{q^{k}-q^{k-1}} - \frac{q^{a-1}-q^{k-1}}{q^{k}-q^{k-1}} = \begin{bmatrix} a-k+1\\1 \end{bmatrix} - \begin{bmatrix} a-k\\1 \end{bmatrix} = q^{a-k}$$

k-dimensional subspaces of V that are not contained in H. Hence,

$$\begin{bmatrix} a \\ k \end{bmatrix} = \left| \begin{bmatrix} V \\ k \end{bmatrix} \right| = \left| \left\{ S \in \begin{bmatrix} V \\ k \end{bmatrix} : S \not\subset H \right\} \right| + \left| \left\{ S \in \begin{bmatrix} V \\ k \end{bmatrix} : S \subset H \right\}$$
$$= q^{a-k} \begin{bmatrix} a-1 \\ k-1 \end{bmatrix} + \begin{bmatrix} a-1 \\ k \end{bmatrix},$$

which is the second identity. Since  $a \in \mathbb{Z}^+$ , we can use the symmetry property of the binomial coefficients (4.3.2) to yield the first identity.

## 4.4 The *q*-Kneser graphs

We formally state and prove the Erdős-Ko-Rado theorem for vector spaces, and introduce the closely-related *q*-Kneser graphs.

**Definition 4.4.1.** A family  $\mathscr{F} \subset {V \brack k}$  is intersecting if the intersection of any two members of  $\mathscr{F}$  has nonzero dimension; that is, for all  $F, F' \in \mathscr{F}$  we have  $\dim(F \cap F') \neq 0$ .

We discussed the analog of the Erdős-Ko-Rado question in Section 4.1.1 and formally state it now. Hsieh [72] first proved the bound and characterized equality when n > 2k. His proof does not work for all relevant values of n and q, and involves many computations. Later, Frankl and Wilson [60] proved the bound and characterized equality for n > 2k, essentially by computing the eigenvalues of a generalized q-Kneser graph. More recently, Godsil and Newman [66, 94] used Frankl and Wilson's methods to characterize equality in the case n = 2k.

**Theorem 4.4.2** (Hsieh, Frankl-Wilson, Godsil-Newman). Suppose  $\mathscr{F} \subset {V \brack k}$  is intersecting and  $n \ge 2k$ . Then  $|\mathscr{F}| \le {n-1 \brack k-1}$ . Equality holds if and only if

(i) 
$$\mathscr{F} = \left\{ F \in \begin{bmatrix} V \\ k \end{bmatrix} : v \subset F \right\}$$
 for some one-dimensional subspace  $v \subset V$  or

(ii) 
$$n = 2k$$
 and  $\mathscr{F} = \begin{bmatrix} H \\ k \end{bmatrix}$  where H is a  $(2k-1)$ -dimensional subspace of V.

Recall that in the case of sets, there were exponentially many ways to obtain an extremal family in the case n = 2k. Since vector spaces have additional structure, the characterization of equality is stronger and there are only two extremal families when n = 2k. We will see further examples of results which are stronger for vector spaces than for sets in Section 4.6.3.

As in the Erdős-Ko-Rado proof from Section 3.3.2, our proof of Theorem 4.4.2 hinges on computing the eigenvalues of the *q*-Kneser graph, which we define now.

**Definition 4.4.3.** The <u>*q*-Kneser graph</u>, denoted  $qK_{n:k}$ , has vertex set  $\begin{bmatrix} V \\ k \end{bmatrix}$  and edge set  $E = \{\{A, B\} : A, B \in \begin{bmatrix} V \\ k \end{bmatrix}, A \cap B = \{0\}\}.$ 

Many parameters of the q-Kneser graphs are given by expressions that involve q-binomial coefficients and which reduce to those of the Kneser graph when we set

q = 1. The next two lemmas give examples of such parameters. Recall that, as part of the proof from Section 3.3.2, we showed that the Kneser graph  $K_{n:k}$  has  $\binom{n}{k}$  vertices and is  $\binom{n-k}{k}$ -regular. We now prove that the *q*-Kneser graph  $qK_{n:k}$  has  $\binom{n}{k}$  vertices and is  $q^{k^2} \binom{n-k}{k}$ -regular. Observe that, when q = 1, the expressions for the number of vertices and the degree of the *q*-Kneser graph reduce to that of the Kneser graph.

**Lemma 4.4.4.** The q-Kneser graph  $qK_{n:k}$  has  $\begin{bmatrix}n\\k\end{bmatrix}$  vertices and is regular with degree  $q^{k^2} \begin{bmatrix}n-k\\k\end{bmatrix}$ .

**Proof.** Lemma 4.2.4 proves that the *q*-Kneser graph  $qK_{n:k}$  has  $\begin{bmatrix}n\\k\end{bmatrix}$  vertices. Now we show that the *q*-Kneser graph is regular and we determine its degree. Let  $\alpha$  be a vertex of  $qK_{v:k}$ ; it is a *k*-dimensional subspace and contains  $q^k$  elements of  $\mathbb{F}_q^n$ . There are  $(q^n - q^k)(q^n - q^{k+1})\cdots(q^n - q^{2k-1})$  ways to choose *k*-tuple independent vectors in  $\mathbb{F}_q^n$  that are not in  $\alpha$ . Since a given *k*-space has  $(q^k - 1)(q^k - q)\cdots(q^k - q^{k-1})$  distinct ordered bases, there are

$$\frac{(q^n - q^k)(q^n - q^{k+1})\cdots(q^n - q^{2k-1})}{(q^k - 1)(q^k - q)\cdots(q^k - q^{k-1})} = q^{k^2} \begin{bmatrix} n - k \\ k \end{bmatrix}$$

*k*-spaces in  $\mathbb{F}_q^n$  whose intersection with  $\alpha$  is the zero subspace. Hence, the *q*-Kneser graph  $qK_{n:k}$  is regular with degree  $q^{k^2} {n-k \choose k}$ .

Recall that in Theorem 3.3.9, we computed the eigenvalues of the Kneser graph's adjacency matrix and their multiplicities. Delsarte computed the corresponding quantities for the *q*-Kneser graph [36]. Again, when q = 1, the expressions for the eigenvalues and multiplicities reduce to that of the Kneser graph. Here, we interpret  $\begin{bmatrix} n \\ 0 \end{bmatrix} = 1$  and  $\begin{bmatrix} n \\ -1 \end{bmatrix} = 0$ .

**Theorem 4.4.5** (Delsarte). Let  $n \ge 2k$  and let A be the adjacency matrix of the q-Kneser graph  $qK_{n:k}$ . The eigenvalues  $\lambda_i$  of A and their corresponding multiplicities  $m_i$  are

$$\lambda_i = (-1)^i q^{k(k-i)} \begin{bmatrix} n-k-i \\ k-i \end{bmatrix}, \ m_i = \begin{bmatrix} n \\ i \end{bmatrix} - \begin{bmatrix} n \\ i-1 \end{bmatrix}, \ i = 0, \dots, k.$$

Now we prove Theorem 4.4.2 using the eigenvalue method of Section 3.3.2. As in the set case, the independent sets in the *q*-Kneser graph are in bijective correspondence with intersecting families  $\mathscr{F} \subset \begin{bmatrix} V \\ k \end{bmatrix}$ .

**Proof of Theorem 4.4.2.** Suppose  $n \ge 2k$ . Let  $\mathscr{F} \subset \begin{bmatrix} V \\ k \end{bmatrix}$  be a maximum intersecting family. We know  $\mathscr{F}$  is an independent set in the *q*-Kneser graph  $qK_{n:k}$ . By Theorem 4.4.5, the least eigenvalue of the Kneser graph  $qK_{n:k}$  is

$$\tau = \lambda_1 = -q^{k(k-1)} \begin{bmatrix} n-k-1 \\ k-1 \end{bmatrix}.$$

In Lemma 4.4.4, we showed that the *q*-Kneser graph  $qK_{n:k}$  is  $q^{k^2} {n-k \choose k}$ -regular so the ratio bound, Theorem 3.3.7, yields

$$|\mathscr{F}| \leq \frac{{n \brack k}}{1 + \frac{q^{k^2} {n-k \brack k}}{q^{k(k-1)} {n-k-1 \brack k-1}}} = {n-1 \brack k-1},$$

which establishes the bound in Theorem 4.4.2. Using the second claim of Theorem 3.3.7 and the multiplicity  $m_1 = [n] - 1$  of the least eigenvalue  $\lambda_1$ , it is possible to characterize the case of equality when  $n \ge 2k$ ; see [94, Section 5.5] for details.

## 4.5 Spreads

Spreads are the vector space analogs of perfect matchings. We will need them in Section 4.6, where we give a proof of the Erdős-Ko-Rado theorem for vector spaces, Theorem 4.4.2, in the case that k|n. We also use spreads in Chapter 5 to prove an analog of Frankl's *r*-wise intersection theorem, Theorem 2.2.7.

**Definition 4.5.1.** A family  $\mathscr{S} \subset \begin{bmatrix} V \\ t \end{bmatrix}$  of t-dimensional subspaces of V is called a <u>t-spread</u> if every one-dimensional subspace of V is contained in exactly one t-dimensional subspace in  $\mathscr{S}$ .

**Definition 4.5.2.** If  $\mathscr{S} \subset \begin{bmatrix} V \\ t \end{bmatrix}$  is a spread and the elements in  $\mathscr{S}$  that lie in a subspace *U* form a *t*-spread of *U*, then we say that  $\mathscr{S}$  <u>induces</u> a *t*-spread on *U*.

**Definition 4.5.3.** A *t*-spread  $\mathscr{S} \subset \begin{bmatrix} V \\ t \end{bmatrix}$  is called <u>geometric</u> if  $\mathscr{S}$  induces a *t*-spread on each 2*t*-dimensional subspace generated by a pair of elements in  $\mathscr{S}$ .

Baer and Segre [6, 100] proved necessary and sufficient conditions for the existence of geometric *t*-spreads.

**Lemma 4.5.4** (Baer, Segre). A geometric *t*-spread  $\mathscr{S} \subset \begin{bmatrix} V \\ t \end{bmatrix}$  exists if and only if t|n.

**Proof.** If  $\mathscr{S} \subset {V \brack t}$  is a spread, then the size of  $\mathscr{S}$  is

$$|\mathscr{S}| = \frac{\binom{n}{1}}{\binom{t}{1}} = \frac{q^n - 1}{q^t - 1}.$$

Since  $q, |\mathcal{S}| \in \mathbb{Z}^+$  are positive integers, we must have t|n.

Now suppose that n = tl where  $l \in \mathbb{Z}^+$ . We have that *V* is an *n*-dimensional vector space over the field  $\mathbb{F}_q$ , but we can also view *V* as an *l*-dimensional space over the field  $\mathbb{F}_{q^t}$ . Let  $\mathscr{S} \subset {V \brack 1}_{q^t}$  be the family of 1-dimensional subspaces of *V* as a vector space over  $\mathbb{F}_{q^t}$ . We have that

$$|\mathscr{S}| = \begin{bmatrix} l \\ 1 \end{bmatrix}_{q^t} = q^{t(l-1)} + q^{t(l-2)} + \dots + q^t + 1 = \frac{q^{tl} - 1}{q^l - 1} = \frac{q^n - 1}{q^t - 1}.$$

If we think of *V* as an *n*-dimensional vector space over  $\mathbb{F}_q$  again, then the members of  $\mathscr{S}$  are now *t*-dimensional spaces over  $\mathbb{F}_q$  and form a geometric *t*-spread of *V*.  $\Box$ 

We now prove two properties of geometric spreads that we will need for our work in Section 4.6 and Chapter 5. The first proves that a geometric *t*-spread  $\mathscr{S} \subset \begin{bmatrix} V \\ t \end{bmatrix}$  does not just induce *t*-spreads on the 2*t*-dimensional spaces generated by a pair of elements in  $\mathscr{S}$ , but also on any subspace generated by elements of  $\mathscr{S}$ .

**Lemma 4.5.5.** If  $\mathscr{S}$  is a geometric *t*-spread of V, then  $\mathscr{S}$  induces a geometric *t*-spread on any subspace of V that is generated by elements of  $\mathscr{S}$ .

**Proof.** Let  $\{X_1, \ldots, X_m\} \subset \mathscr{S}$ , where  $m \leq n/t$ , and suppose dim $(X_1 \lor \cdots \lor X_m) = mt$ . We will show that  $\mathscr{S}$  induces a *t*-spread on  $X_1 \lor \cdots \lor X_m$ . We proceed by induction on *m*. If  $m \in \{1, 2\}$ , the statement is true by definition of geometric. Suppose the statement is true for m - 1 and let  $\mathscr{B}$  be the spread that  $\mathscr{S}$  induces on  $X_1 \lor \cdots \lor X_{m-1}$ .

Define  $\mathscr{G} := \{X_m \lor A : A \in \mathscr{B}\} \subset \begin{bmatrix} V \\ 2t \end{bmatrix}$ . We claim that every one-dimensional subspace of  $X_1 \lor \cdots \lor X_m$  lies in some  $G \in \mathscr{G}$ . Observe that distinct  $G_1, G_2 \in \mathscr{G}$  satisfy  $G_1 \cap G_2 = X_m$  and that for any  $G \in \mathscr{G}$ , the number of one-dimensional subspaces in

$$G \setminus X_m \text{ is } \begin{bmatrix} 2t\\1 \end{bmatrix} - \begin{bmatrix} t\\1 \end{bmatrix}. \text{ Hence, the number of one-dimensional subspaces in some } G \in \mathscr{G} \text{ is}$$
$$\left| \left\{ W \in \begin{bmatrix} V\\1 \end{bmatrix} : W \in G \in \mathscr{G} \right\} \right| = |\mathscr{G}| \left( \begin{bmatrix} 2t\\1 \end{bmatrix} - \begin{bmatrix} t\\1 \end{bmatrix} \right) + \left| \left\{ W \in \begin{bmatrix} V\\1 \end{bmatrix} : W \in X_m \right\} \right|$$
$$= \left( q^t \begin{bmatrix} m-1\\1 \end{bmatrix} + 1 \right) \left( \begin{bmatrix} 2t\\1 \end{bmatrix} - \begin{bmatrix} t\\1 \end{bmatrix} \right) + \begin{bmatrix} t\\1 \end{bmatrix} = \begin{bmatrix} mt\\1 \end{bmatrix}$$
$$= \left| \left\{ W \in \begin{bmatrix} V\\1 \end{bmatrix} : W \in X_1 \lor \cdots \lor X_m \right\} \right|,$$

which proves that every one-dimensional subspace of  $X_1 \vee \cdots \vee X_m$  lies in some  $G \in \mathscr{G}$ .

Consequently, if  $X \in \mathscr{S}$  and  $X \cap (X_1 \vee \cdots \vee X_m) \neq \{0\}$ , then  $X \cap G \neq \{0\}$  for some  $G \in \mathscr{G}$ . Since  $\mathscr{S}$  is geometric, this implies  $X \subset G \subset X_1 \vee \cdots \vee X_m$ . Hence,  $\mathscr{S}$ induces a *t*-spread on  $X_1 \vee \cdots \vee X_m$ .

The second lemma proves that an invertible linear transformation of V maps a geometric *t*-spread to another geometric *t*-spread.

**Definition 4.5.6.** The set of all invertible linear transformations of V is denoted

 $GL(V) := \{ \pi : V \to V : \pi \text{ is an invertible linear transformation} \}.$ 

**Lemma 4.5.7.** If  $\mathscr{S}$  is a geometric *t*-spread of *V*, then for any invertible linear transformation  $\pi \in GL(V)$ , the family  $\pi(\mathscr{S}) := {\pi(S) : S \in \mathscr{S}}$  is also a geometric *t*-spread of *V*.

**Proof.** First we check that  $\pi(\mathscr{S})$  is a spread. For  $S \in \mathscr{S}$ , we have  $\pi(S)$  is a *t*-dimensional subspace because  $\pi$  is an isomorphism of *V*. If  $v \in \begin{bmatrix} V \\ 1 \end{bmatrix}$ , then  $\pi^{-1}(v) \in S$  for some  $S \in \mathscr{S}$  as  $\mathscr{S}$  is a spread. Hence,  $v = \pi(\pi^{-1}(v)) \in \pi(S)$ . Also, we cannot have *v* in  $\pi(S) \cap \pi(T)$  for distinct  $S, T \in \mathscr{S}$  as otherwise  $\pi^{-1}(v) \in S \cap T$ , which contradicts  $\mathscr{S}$  being a spread. We have shown that every one-dimensional subspace of *V* lies in exactly one *t*-dimensional subspace of  $\pi(\mathscr{S})$  and so  $\pi(\mathscr{S})$  is a *t*-spread. Given  $S, T \in \mathscr{S}$ , if  $\mathscr{S}'$  is the spread that  $\mathscr{S}$  induces on  $S \vee T$ , then it is easy to check that  $\pi(\mathscr{S}')$  is the spread that  $\pi(S) \vee \pi(T)$ .

## 4.6 Which Tools Work?

On a first glance, it might seem as if theorems in extremal set theory should easily extend to the vector space setting by changing identities involving binomial coefficients to their corresponding *q*-analogs. On closer inspection, however, we will see in Section 4.6.1 and Section 4.6.2 that adapting combinatorial techniques to vector spaces is often not straightforward. Generally, algebraic methods have been more successful than combinatorial ones for vector space analogs, as we saw in Section 4.4. We will discuss further results that have been proved using algebraic techniques in Section 4.6.3.

### 4.6.1 Shifting and Shadows

Adapting combinatorial techniques to vector spaces is often tricky. We cite two flawed attempts to extend the shifting technique to vector spaces. We discuss the difficulties that surface when trying to adapt Daykin's proof in Section 2.2.4 to vector spaces. We also show that no analog of the colex order exists for shadows in vector spaces.

### 4.6.1.1 Shifting

There have been two published attempts [31, 39] to extend the shifting proof from Section 2.2.3 to vector spaces, and both of these attempts are acknowledged to be flawed by their respective authors.

### 4.6.1.2 Daykin's Proof

For a simple example of the kinds of issues that occur when attempting to extend combinatorial proofs to vector spaces, consider Daykin's proof of the Erdős-Ko-Rado theorem in Section 2.2.4. The definition of a family's shadow extends naturally to vector spaces: if  $\mathscr{F} \subset \begin{bmatrix} V \\ k \end{bmatrix}$  is a family of *k*-dimensional subspaces of *V*, then its shadow consists of the (k-1)-dimensional subspaces of *V* contained in at least one member of  $\mathscr{F}$ . In Chapter 5, we will prove an analog of Lovász's result, Theorem 2.1.5, for vector spaces. We cannot, however, mimic its application in Daykin's proof.

Let  $\mathscr{F} \subset \begin{bmatrix} V \\ k \end{bmatrix}$  be intersecting, and suppose we try to mimic Daykin's proof to show that  $|\mathscr{F}| \leq \begin{bmatrix} n-1 \\ k-1 \end{bmatrix}$ . It is not clear what the analog of  $\mathscr{G}$  should be as the set complement  $V \setminus F$  of the subspace  $F \in \mathscr{F}$  with respect to V is not itself a subspace. If we try to define  $\mathscr{G} = \{F^{\perp} : F \in \mathscr{F}\} \subset \begin{bmatrix} V \\ n-k \end{bmatrix}$  to be the family of orthogonal complements of  $\mathscr{F}$ , then it is possible that  $F \subset G$  for some  $G \in \mathscr{G}$ . More importantly, the use of the Pascal identity in Daykin's proof does not work with the q-Pascal identity.

#### 4.6.1.3 No Analog of the Colex Order

A crucial difference between sets and vector spaces is that no analog of the colex order exists for shadows in vector spaces. Recall that we defined the colex order in Definition 2.1.2. We say that a set system  $\mathscr{H} \subset {\binom{X}{k}}$  is a solution to the shadow minimization problem with parameters *k* and  $|\mathscr{H}|$  if the shadow of  $\mathscr{H}$  is minimum over all set-systems  $\mathscr{F} \subset {\binom{X}{k}}$  with the same cardinality,

$$|\partial \mathscr{H}| = \min_{|\mathscr{F}| = |\mathscr{H}|} |\partial \mathscr{F}|.$$

We are especially interested in the case where there are nested solutions  $\{\mathscr{H}_m\}$ , i.e., such that  $|\mathscr{H}_m| = m$ ,  $\mathscr{H}_m$  is a solution to the shadow minimization problem with parameters k and m, and  $\mathscr{H}_{m-1} \subset \mathscr{H}_m$  for  $1 \le m \le {n \choose k}$ .

The Kruskal Katona theorem, Theorem 2.1.4, shows that the colex order on  $\binom{X}{k}$  satisfies the following properties:

- 1. For  $1 \le m \le {\binom{n}{k}}$ , we have  $\{\mathscr{C}_m^k\}$  is a family of nested solutions to the shadow minimization problem with parameters *k* and *m*.
- 2. The shadow of  $\mathscr{C}_m^k$  is an initial segment of the colex order on  $\binom{X}{k-1}$ .

No analog of the colex order exists for vector spaces. For example, Bezrukov and Blokhuis [12] showed that there is no total order of the subspaces in  $\mathbb{F}_2^n$  satisfying properties (i) and (ii) when  $n \ge 4$ . Even more remarkably, Harper [69] and Ure [109] showed that nested solutions to the shadow minimization problem for vector spaces do not always exist.

**Theorem 4.6.1** (Harper, Ure). Let  $V = \mathbb{F}_8^3$ . If  $\mathscr{F} \subset {V \choose 2}$  has size  $|\mathscr{F}| = 24$  and minimum shadow over all families with the same cardinality, then there does not exist  $\mathscr{F}' \subset \mathscr{F}$  such that  $|\mathscr{F}'| = 22$  and  $\mathscr{F}'$  has minimum shadow over all families with the same cardinality.

## 4.6.2 Katona's Cyclic Permutation Method

Like the combinatorial methods discussed previously, Katona's cyclic permutation technique in Section 2.2.5 does not readily adapt to give a full proof of the Erdős-Ko-Rado theorem for vector spaces, Theorem 4.4.2. A similar argument, however, does yield the theorem in the special case that k|n. The proof uses spreads and is due to Greene and Kleitman [67]. This argument will be important for the analog of Frankl's *r*-wise intersection theorem in Chapter 5 and for the Manickam-Miklós-Singhi conjecture in Chapter 7. We first prove two lemmas; the first computes the size of GL(V). Note that, when q = 1, the expression for the size of GL(V) reduces to that of the size of  $S_X$ , the group of permutations on X.

**Lemma 4.6.2.** We have  $|GL(V)| = q^{n(n-1)/2}(q-1)^n[n]!$ 

**Proof.** Since *V* is an *n*-dimensional vector space over  $\mathbb{F}_q$ , there are

$$(q^{n}-1)(q^{n}-q)(q^{n}-q^{2})\cdots(q^{n}-q^{n-1}) = q^{n(n-1)/2}(q-1)^{n}[n]!$$

ways to choose a basis of *V*. The number of invertible linear transformations of *V* equals the number of distinct bases of *V* so  $|GL(V)| = q^{n(n-1)/2}(q-1)^n[n]!$ .

The second lemma shows that if  $A, B \in {V \brack k}$  are *k*-dimensional subspaces of *V*, then there are  $q^{n(n-1)/2}(q-1)^n[k]![n-k]!$  invertible linear transformations  $\pi \in GL(V)$  such that  $\pi(A) = B$ . Observe that, when q = 1, this expression reduces to the number of permutations of *X* that send a given *k*-subset to another.

**Lemma 4.6.3.** If  $A, B \in {V \brack k}$  are k-dimensional subspaces of V, then the number of invertible linear transformations such that  $\pi(A) = B$  is  $q^{n(n-1)/2}(q-1)^n[k]![n-k]!$ 

**Proof.** Let  $v_1, \ldots, v_k$  be a basis of A. Extend this basis to a basis  $v_1, \ldots, v_k, v_{k+1}, \ldots, v_n$ of V. Since  $\pi(A) = B$ , there are  $q^k - 1$  choices for  $\pi(v_1)$ . For  $i \in [k] \setminus \{1\}$ , we must pick  $\pi(v_i)$  such that  $\pi(v_i) \in B$  but is not a linear combination of  $\pi(v_1), \ldots, \pi(v_{i-1})$ . Hence, there are  $q^k - q^{i-1}$  choices for  $\pi(v_i)$  when  $i \in [k]$ . We must pick  $\pi(v_{k+1}) \in V \setminus B$  so there are  $q^n - q^k$  choices for  $\pi(v_{k+1})$ . For each  $j \in [n] \setminus [k+1]$ , we must choose  $\pi(v_j)$  in  $V \setminus B$ so that  $\pi(v_j)$  is not a linear combination of  $\pi(v_{k+1}), \ldots, \pi(v_{j-1})$ . Consequently, there are  $q^n - q^{j-1}$  choices for  $\pi(v_j)$  when  $j \in [n] \setminus [k]$ . We see that the number of invertible linear transformations such that  $\pi(A) = B$  is

$$(q^{k}-1)\cdots(q^{k}-q^{k-1})(q^{n}-q^{k})\cdots(q^{n}-q^{n-1}) = q^{n(n-1)/2}(q-1)^{n}[k]![n-k]! \quad \Box$$

Now, we give Greene and Kleitman's simple proof [67] of the Erdős-Ko-Rado theorem for vector spaces, Theorem 4.4.2, in the case that k|n.

**Proof.** Let  $\mathscr{F} \subset \begin{bmatrix} V \\ k \end{bmatrix}$  be an intersecting family and assume that k|n. Let  $\mathscr{S} \subset \begin{bmatrix} V \\ k \end{bmatrix}$  be a spread of *V* and let  $\pi \in GL(V)$  be an invertible linear transformation. The proof of Lemma 4.5.7 shows that  $\pi(\mathscr{S})$  is also a spread. Since  $\mathscr{F} \subset \begin{bmatrix} V \\ k \end{bmatrix}$  is intersecting, for any  $\pi \in GL(V)$ , we have  $|\mathscr{F} \cap \pi(\mathscr{S})| \leq 1$ . Consequently, by Lemma 4.6.2, we have

$$\sum_{\pi \in GL(V)} |\mathscr{F} \cap \pi(\mathscr{S})| \le q^{n(n-1)/2} (q-1)^n [n]!$$

Now given  $S \in \mathscr{S}$  and  $F \in \mathscr{F}$ , there are  $q^{n(n-1)/2}(q-1)^n[k]![n-k]!$  invertible linear transformations  $\pi \in GL(V)$  such that  $\pi(S) = F$  by Lemma 4.6.3. Hence,

$$\begin{split} \left(\frac{q^n-1}{q^k-1}\right)|\mathscr{F}|q^{n(n-1)/2}(q-1)^n[k]![n-k]! &= |\mathscr{S}||\mathscr{F}||\{\pi \in GL(V) : \pi(S) = F\}|\\ &= \sum_{\pi \in GL(V)} |\mathscr{F} \cap \pi(\mathscr{S})|\\ &\leq q^{n(n-1)/2}(q-1)^n[n]! \end{split}$$

We consequently have that

$$|\mathscr{F}| \le \left(\frac{q^k - 1}{q^n - 1}\right) \frac{[n]!}{[k]![n - k]!} = \frac{[k]}{[n]} \frac{[n]!}{[k]![n - k]!} = {n - 1 \choose k - 1}.$$

## 4.6.3 Algebraic Successes

In this section, we mention some other important intersection theorems for vector spaces. Common to all these results is that their proofs use algebraic techniques.

First, we discuss Frankl and Graham's analog of the modular Frankl-Wilson theorem from Section 3.2. As in the Erdős-Ko-Rado theorem for vector spaces, this result provides another example of a theorem that is stronger for vector spaces than for

sets. Frankl and Graham's proof makes use of the algebraic machinery of higher incidence matrices, and many variants of Fisher's Inequality, Theorem 2.3.9, were originally proved in this way as well.

Vector space analogs do not only provide applications of algebraic methods, but also inspire new ones. Lovász, for example, introduced his wedge product technique specifically to prove the vector space analog of Bollobás's Two Families Theorem. We discuss this result in Section 4.6.3.2.

Finally, in Section 4.6.3.3, we state some old and new results concerning the analog of the *t*-intersection theorem in Section 2.2.8 to vector spaces.

### 4.6.3.1 Modular Frankl-Wilson Theorem

Recall that the Frankl-Wilson theorem, Theorem 3.2.2, asserts that if  $\mathscr{F} \subset 2^X$  is an *L*-intersecting family, where |L| = s, then  $|\mathscr{F}| \leq \sum_{i=0}^{s} {n \choose i}$ . If  $\mathscr{F} \subset {X \choose k}$  is a *uniform L*-intersecting family, then Ray-Chaudhuri and Wilson [97] proved that the upper bound on  $|\mathscr{F}|$  can be strengthened.

**Theorem 4.6.4** (Ray-Chaudhuri – Wilson). Suppose  $L \subset \mathbb{N}$  has size |L| = s. If  $\mathscr{F} \subset {\binom{X}{k}}$  is a uniform *L*-intersecting family, then  $|\mathscr{F}| \leq {\binom{n}{s}}$ .

Frankl and Wilson [59] showed that the conclusion of Theorem 4.6.4 holds under the considerably weaker condition that the intersection sizes in L belong to at most sresidue classes modulo a prime p.

**Theorem 4.6.5** (Frankl-Wilson). Suppose  $p \in \mathbb{Z}^+$  is a prime number and that  $L \subset \mathbb{N}$  has size  $|L| = s \leq p - 1$ . Also suppose that  $k \in \mathbb{Z}^+$  satisfies  $k \notin L \pmod{p}$ . If  $\mathscr{F} \subset \binom{X}{k}$  is a uniform family such that  $|F \cap F'| \in L \pmod{p}$  for all distinct  $F, F' \in \mathscr{F}$ , then  $|\mathscr{F}| \leq \binom{n}{s}$ .

The restriction that p be prime in Theorem 4.6.5 is crucial. If the prime number p is replaced by p = 6, for example, or even  $q = p^2$ , where  $p \ge 7$ , then Theorem 4.6.5 will be false [5, Section 5.9]!

Remarkably, Frankl and Graham [56] showed that an analog of Theorem 4.6.5 holds for vector spaces but *without* the restriction that p be prime. As in the Erdős-Ko-Rado theorem for vector spaces, Theorem 4.4.2, we obtain a stronger analog for vector spaces because of the additional structure they have.

**Theorem 4.6.6** (Frankl-Graham). Suppose  $b \in \mathbb{Z}^+$  is a positive integer and that  $L \subset \mathbb{N}$ has size  $|L| = s \leq b - 1$ . Also suppose that  $k \in \mathbb{Z}^+$  satisfies  $k \notin L \pmod{b}$ . If  $\mathscr{F} \subset {V \choose k}$ is a uniform family such that  $\dim(F \cap F') \in L \pmod{b}$  for all distinct  $F, F' \in \mathscr{F}$ , then

$$|\mathscr{F}| \leq \begin{bmatrix} n \\ s \end{bmatrix},$$

*except possibly for*  $q = 2, b = 6, s \in \{3, 4\}$ .

### 4.6.3.2 Bollobás's Two Families Theorem

Bollobás's Two Families Theorem [16] is an important combinatorial result with many consequences [2, 44, 96]. An elegant proof of the theorem using the polynomial method is given in [5, Section 5.1].

**Theorem 4.6.7** (Bollobas). Suppose that  $A_1, \ldots, A_m \in {\binom{X}{r}}$  are r-element sets and that  $B_1, \ldots, B_m \in {\binom{X}{s}}$  are s-element sets such that

- (*i*)  $A_i$  and  $B_i$  are disjoint for  $i \in [m]$ ,
- (ii)  $A_i$  and  $B_j$  have nonempty intersection whenever  $i \neq j$ .

Then  $m \leq \binom{r+s}{r}$ .

We have seen in Section 4.6.1 that adapting combinatorial techniques to vector spaces can be challenging. In Section 4.4 and Section 4.6.3.1, we also saw examples of algebraic proofs of extremal set theory results that generalize nicely to their vector space analogs. In the case of Bollobás's Two Families Theorem, neither the known combinatorial or algebraic proofs seemed to adapt to the case of vector spaces. As a result, Lovász [82] introduced the method of wedge products specifically to tackle the analog of Theorem 4.6.7.

**Theorem 4.6.8** (Lovász). Let  $U_1, \ldots, U_m \in {V \brack r}$  be *r*-dimensional subspaces of *V* and let  $W_1, \ldots, W_m \in {V \brack s}$  be s-dimensional subspaces of *V* such that

- (*i*)  $U_i \cap W_i = \{0\}$  for  $i \in [m]$ ,
- (ii)  $U_i \cap W_j \neq \{0\}$  whenever  $i \neq j$ .

Then  $m \leq \binom{r+s}{r}$ .

### 4.6.3.3 On *t*-intersecting Families

In this section, we state some old and new results on *t*-intersecting families of vector spaces.

**Definition 4.6.9.** For a positive integer  $t \in \mathbb{Z}^+$ , a family  $\mathscr{F} \subset \begin{bmatrix} V \\ k \end{bmatrix}$  of k-dimensional spaces is called <u>t-intersecting</u> if any two elements in  $\mathscr{F}$  have intersection of dimension at least t; that is, for all  $F, F' \in \mathscr{F}$  we have  $\dim(F \cap F') \ge t$ .

Wilson's proof [116] of the *t*-intersection theorem, Theorem 2.2.12, generalizes nicely to vector spaces [60].

**Theorem 4.6.10** (Frankl-Wilson). If  $\mathscr{F} \subset {V \brack k}$  is *t*-intersecting and  $n \ge 2k - t$ , then

$$|\mathscr{F}| \le \max\left\{ {n-t \brack k-t}, {2k-t \brack k} \right\}.$$
(4.6.1)

We also have that

- (i) if n > 2k, then equality holds in (4.6.1) if and only if  $\mathscr{F} = \left\{ F \in \begin{bmatrix} V \\ k \end{bmatrix} : S \subset F \right\}$  for some t-dimensional subspace  $S \in \begin{bmatrix} V \\ t \end{bmatrix}$ ;
- (ii) if  $2k t \le n < 2k$ , then equality holds in (4.6.1) if and only if  $\mathscr{F} = \begin{bmatrix} Y \\ k \end{bmatrix}$  for some (2k t)-dimensional subspace  $Y \in \begin{bmatrix} V \\ 2k t \end{bmatrix}$ .

Note that when n = 2k, we have  $\binom{n-t}{k-t} = \binom{2k-t}{k}$ . Hence, the two non-isomorphic families in Theorem 4.6.10 have the same cardinality and they are conjectured to be the only extremal families when n = 2k. This has only been proved, however, for t = 1 by Godsil and Newman [66, 94].

Very recently, Tokushige [106] used Wilson's methods along with new results by Ellis, Friedgut, and Pilpel [46] to generalize Theorem 4.6.10 to cross-intersecting families. Another recent result is Frankl and Tokushige's proof [57] of an analog of Katona's *t*-intersection theorem using the method of higher incidence matrices discussed in Section 4.6.3.1.

# Chapter 5

# Shadows and Intersections in Vector Spaces

We [27] prove a vector space analog of Lovász's version of the Kruskal-Katona theorem, Theorem 2.1.5. We apply this result to extend Frankl's theorem on r-wise intersecting families, Theorem 2.2.7, to vector spaces. In particular, we obtain a short new proof of the Erdős-Ko-Rado theorem for vector spaces, Theorem 4.4.2. In Section 5.4, we cite our other results in this area, namely obtaining a vector space analog of the Hilton-Milner theorem [14, 24] and determining the chromatic number of the q-Kneser graph [14, 26]. Finally, we end the chapter in Section 5.5 with some open problems.

Before we can give precise statements of our results, we first formally define the concepts of shadow and *r*-wise intersection for vector spaces. We see that they are natural analogs of their set counterparts.

**Definition 5.0.11.** For a family  $\mathscr{F} \subset \begin{bmatrix} V \\ k \end{bmatrix}$ , we define the shadow of  $\mathscr{F}$ , denoted  $\partial \mathscr{F}$ , to consist of those (k-1)-dimensional subspaces of V contained in at least one member of  $\mathscr{F}$ ,

$$\partial \mathscr{F} := \left\{ E \in \begin{bmatrix} V \\ k-1 \end{bmatrix} : E \subset F \in \mathscr{F} \right\}.$$

As in Section 4.6.1.3, we can ask for families  $\mathscr{H} \subset \begin{bmatrix} V \\ k \end{bmatrix}$  whose shadow is minimum over all families  $\mathscr{F} \subset \begin{bmatrix} V \\ k \end{bmatrix}$  with the same cardinality,

$$|\partial \mathscr{H}| = \min_{|\mathscr{F}| = |\mathscr{H}|} |\partial \mathscr{F}|.$$

In general, this is a difficult question because of the reasons discussed in Section 4.6.1.3. Surprisingly, we can still prove an analog of Lovász's version of the Kruskal-Katona theorem. Recall that, in Lovász's result, given a family  $\mathscr{F} \subset {X \choose k}$ , we could find a real number  $y \ge k$  such that  $|\mathscr{F}| = {y \choose k}$ . The same is true for *q*-binomial coefficients. If *k* and *q* are fixed, then  $\begin{bmatrix} a \\ k \end{bmatrix}$  is a continuous function of *a* which is positive and strictly increasing when  $a \ge k$ ; hence, by the intermediate value theorem, if  $r \ge 1$  is a real number, then there exists a unique real number  $a_r \ge k$  such that  $r = \begin{bmatrix} a_r \\ k \end{bmatrix}$ . Consequently, we can formulate an analog of Lovász's version of the Kruskal-Katona theorem, and this is our first result.

**Theorem 5.0.12** (Chowdhury-Patkós). Let  $\mathscr{F} \subset \begin{bmatrix} V \\ k \end{bmatrix}$  and let  $y \ge k$  be the real number defined by  $|\mathscr{F}| = \begin{bmatrix} y \\ k \end{bmatrix}$ . Then

$$|\partial \mathscr{F}| \ge \begin{bmatrix} y\\ k-1 \end{bmatrix}.$$

If equality holds, then  $y \in \mathbb{Z}^+$  and  $\mathscr{F} = \begin{bmatrix} Y \\ k \end{bmatrix}$ , where Y is a y-dimensional subspace of V.

As in Section 2.2.6, we can apply our Lovász analog, Theorem 5.0.12, to yield an analog of Frankl's *r*-wise intersection theorem, Theorem 2.2.7. Frankl's proof from Section 2.2.6, however, does not generalize to vector spaces for the reasons discussed in Section 4.6.1. Hence, we will need new proof techniques.

**Definition 5.0.13.** A family  $\mathscr{F} \subset \begin{bmatrix} V \\ k \end{bmatrix}$  *r*-wise intersecting if the intersection of any *r* members of  $\mathscr{F}$  has nonzero dimension; that is,  $\dim(\bigcap_{i=1}^{r} F_i) \neq 0$  for all  $F_1, \ldots, F_r \in \mathscr{F}$ .

**Theorem 5.0.14** (Chowdhury-Patkós). Suppose that  $\mathscr{F} \subset \begin{bmatrix} V \\ k \end{bmatrix}$  is *r*-wise intersecting and that  $(r-1)n \ge rk$ . Then

$$|\mathscr{F}| \le {n-1 \brack k-1}.$$

Equality holds if and only if  $\mathscr{F} = \left\{ F \in \begin{bmatrix} V \\ k \end{bmatrix} : v \subset F \right\}$  for some one-dimensional subspace  $v \subset V$ , unless r = 2 and n = 2k.

Theorem 5.0.12 establish shadows as a viable method for proving vector space analogs such as Theorem 5.0.14, and more applications are expected. Recently, Wang [113] used Theorem 5.0.12 to prove a conjecture of Erdős, Faigle, and Kern [48]. The method of proof in Theorem 5.0.14 has also been used to prove results on *r*-cross intersecting families of sets [58]; Patkós and I are currently working with Frankl and Tokushige to extend these results to vector spaces.

## 5.1 Shadows and an Analog of Lovász's Theorem

Recall that, in Section 2.1.6, we gave Keevash's [77] recent and elegant proof of Lovász's version of the Kruskal-Katona theorem. In this section, we adapt his argument to prove Theorem 5.0.12. We first generalize all the definitions in Section 2.1.6 to their vector space analogs.

**Definition 5.1.1.** *For*  $\mathscr{F} \subset \begin{bmatrix} V \\ k \end{bmatrix}$  *and*  $v \in \begin{bmatrix} V \\ 1 \end{bmatrix}$ *, define* 

$$K_{k+1}^{k}(\mathscr{F}) := \left\{ T \in \begin{bmatrix} V \\ k+1 \end{bmatrix} : \begin{bmatrix} T \\ k \end{bmatrix} \subset \mathscr{F} \right\}$$

to be the family of (k+1)-dimensional spaces of V all of whose k-dimensional subspaces lie in  $\mathscr{F}$  and

$$K_{k+1}^k(v) := \{T \in K_{k+1}^k(\mathscr{F}) : v \subset T\}$$

to be the family of (k+1)-dimensional spaces in  $K_{k+1}^k(\mathscr{F})$  that contain v.

**Definition 5.1.2.** For  $v \in \begin{bmatrix} V \\ 1 \end{bmatrix}$ , define the <u>degree</u> of v, which is denoted by deg(v), to be the number of elements of  $\mathscr{F}$  that contain v,

$$\deg(v) := |\{F \in \mathscr{F} : v \subset F\}|.$$

**Definition 5.1.3.** If  $v \in {V \brack 1}$  and  $U \subset V$  is an (n-1)-dimensional subspace that does not contain v, then define the link of v in U to be the family of (k-1)-dimensional spaces in U whose linear span with v is an element of  $\mathscr{F}$ ,

$$L_U(v) := \left\{ A \in \begin{bmatrix} U \\ k-1 \end{bmatrix} : A \lor v \in \mathscr{F} \right\} \subset \begin{bmatrix} V \\ k-1 \end{bmatrix}.$$

As in Section 2.1.6, we first establish an upper bound on  $|K_{k+1}^k(\mathscr{F})|$  in terms of  $|\mathscr{F}|$ ; we will see that Theorem 5.0.12 follows as a simple corollary.

**Theorem 5.1.4** (Chowdhury-Patkós). Let  $\mathscr{F} \subset \begin{bmatrix} V \\ k \end{bmatrix}$  and let  $y \ge k$  be the real number defined by  $|\mathscr{F}| = \begin{bmatrix} y \\ k \end{bmatrix}$ . Then

$$|K_{k+1}^k(\mathscr{F})| \le \begin{bmatrix} y\\k+1 \end{bmatrix}.$$

Equality holds if and only if  $y \in \mathbb{Z}^+$  and  $\mathscr{F} = \begin{bmatrix} Y \\ k \end{bmatrix}$  for some y-dimensional subspace  $Y \subset V$ .

*Proof.* We argue by induction on k. The base case k = 1 is easy: Suppose  $\mathscr{F} \subset \begin{bmatrix} V \\ 1 \end{bmatrix}$  and  $|\mathscr{F}| = [y]$ . Since there are q + 1 one-dimensional spaces in a two-dimensional space,  $|K_2^1(v)| \le (1/q)([y] - 1)$  if  $v \in \mathscr{F}$  and  $|K_2^1(v)| = 0$  otherwise. Observe that

$$(q+1)|K_2^1(\mathscr{F})| = \sum_{v \in [1]^V} |K_2^1(v)| \le \frac{|y|(|y|-1)}{q},$$
(5.1.1)

which implies that  $|K_2^1(\mathscr{F})| \leq {y \choose 2}$ .

Now assume the statement is true for k-1. We first show that if  $v \in {V \brack 1}$ , then  $|K_{k+1}^k(v)| \le ([y-k]/[k]) \deg(v)$ ; we will then sum this inequality over all  $v \in {V \brack 1}$  and double count to obtain the desired upper bound on  $|K_{k+1}^k(\mathscr{F})|$ . If  $\deg(v) = 0$ , then clearly  $|K_{k+1}^k(v)| \le ([y-k]/[k]) \deg(v)$ , so we will assume that  $\deg(v) \ne 0$ . We will need to consider the cases where  $\deg(v)$  is large and where  $\deg(v)$  is small separately.

First, let's consider the case when deg $(v) \ge {v-1 \choose k-1}$ . If  $T \in K_{k+1}^k(v)$ , then observe that the  $q^k$  k-dimensional subspaces in T that do not contain v are elements of  $\mathscr{F}$  that do not contain v. By counting pairs (S,T) where  $v \not\subset S \in {V \choose k}$  and  $S \subset T \in K_{k+1}^k(v)$ , the previous observation yields

$$q^{k}|K_{k+1}^{k}(v)| = \left|\left\{S \in \begin{bmatrix} V\\k \end{bmatrix} : v \not\subset S \subset T \in K_{k+1}^{k}(v)\right\}\right| \le |\mathscr{F}| - d(v).$$
(5.1.2)

By rearranging (5.1.2) and applying the *q*-Pascal identity, Lemma 4.3.1, we obtain

$$\begin{aligned} K_{k+1}^{k}(x) &| \leq \frac{|\mathscr{F}| - \deg(x)}{q^{k}} \leq \frac{{[\binom{y}{k}]} - {[\binom{y-1}{k-1}]}}{q^{k}} = \frac{q^{k} {[\binom{y-1}{k}]}}{q^{k}} \\ &= \frac{[y-k]}{[k]} {[\binom{y-1}{k-1}]} \leq \frac{[y-k]}{[k]} \deg(v). \end{aligned}$$
(5.1.3)

We have equality in (5.1.3) if and only if  $\deg(v) = \begin{bmatrix} v-1 \\ k-1 \end{bmatrix}$  since  $\deg(v) \neq 0$ .

Now suppose deg $(v) \leq {\binom{y-1}{k-1}}$ . Let  $U \subset V$  be an (n-1)-dimensional subspace that does not contain v, and observe that  $|L_U(v)| = \deg(v)$ . Moreover, if  $T_1, T_2$  are distinct elements of  $K_{k+1}^k(v)$  then  $T_1 \cap U$  and  $T_2 \cap U$  are distinct k-dimensional spaces in  $K_k^{k-1}(L_U(v))$ . Consequently,  $|K_{k+1}^k(v)| \le |K_k^{k-1}(L_U(v))|$ . Define the real number  $y_v \ge k$ by  $\deg(v) = \begin{bmatrix} y_v - 1 \\ k-1 \end{bmatrix}$ . As  $L_U(v) \subset \begin{bmatrix} V \\ k-1 \end{bmatrix}$  and  $|L_U(v)| = \deg(v) = \begin{bmatrix} y_v - 1 \\ k-1 \end{bmatrix}$ , the induction hypothesis yields that

$$|K_{k+1}^{k}(v)| \le |K_{k-1}^{k}(L(x))| \le {y_{\nu}-1 \choose k} = \frac{[y_{\nu}-k]}{[k]} {y_{\nu}-1 \choose k-1} \le \frac{[y-k]}{[k]} \deg(v), \quad (5.1.4)$$

where the last inequality follows because  $y_v \le y$ , by our assumption on deg(v). We have equality in (5.1.4) if and only if deg(v) =  $\begin{bmatrix} y-1 \\ k-1 \end{bmatrix}$  by our assumption that deg(v)  $\ne 0$ .

To finish the proof, we sum the inequality  $|K_{k+1}^k(v)| \le ([y-k]/[k]) \deg(v)$  over all  $v \in {V \brack 1}$  and double-count to obtain the desired inequality on  $|K_{k+1}^k(\mathscr{F})|$ . We have

$$\begin{split} [k+1]|K_{k+1}^{k}(\mathscr{F})| &= \sum_{v \in \begin{bmatrix} V\\1 \end{bmatrix}} |K_{k+1}^{k}(v)| \le \frac{[y-k]}{[k]} \sum_{v \in \begin{bmatrix} V\\1 \end{bmatrix}} d(v) = \frac{[y-k]}{[k]} [k]|\mathscr{F}| \qquad (5.1.5) \\ &= [y-k] \begin{bmatrix} y\\k \end{bmatrix} = [k+1] \begin{bmatrix} y\\k+1 \end{bmatrix}. \end{split}$$

Therefore,  $|K_{k+1}^k(\mathscr{F})| \leq {y \choose k+1}$ , and equality holds only when all one-dimensional subspaces *v* with nonzero degree satisfy  $d(v) = {y-1 \choose k-1}$ .

We now characterize the case of equality. Again the proof proceeds by induction on *k*. The base case k = 1 is easy: Suppose  $\mathscr{F} \subset \begin{bmatrix} V \\ 1 \end{bmatrix}$ ,  $|\mathscr{F}| = [y]$ , and  $|K_2^1(\mathscr{F})| = \begin{bmatrix} y \\ 2 \end{bmatrix}$ . Then (5.1.1) implies that  $|K_2^1(v)| = (1/q)([y] - 1)$  for all  $v \in \mathscr{F}$ . Hence, if *v*, *w* are distinct elements of  $\mathscr{F}$ , then every one-dimensional space in the two-dimensional space spanned by *v* and *w* lies in  $\mathscr{F}$ . It is easy to see by induction that if *A* is a subspace of dimension  $1 \le d < [y]$  such that  $\begin{bmatrix} A \\ 1 \end{bmatrix} \subset \mathscr{F}$ , then there exists a subspace *B* of dimension d+1 that contains *A* and for which  $\begin{bmatrix} B \\ 1 \end{bmatrix} \subset \mathscr{F}$ . In particular, this proves that  $y \in \mathbb{Z}^+$  and  $\mathscr{F} = \begin{bmatrix} Y \\ 1 \end{bmatrix}$  for some *y*-dimensional subspace *Y*.

Now suppose  $\mathscr{F} \subset {V \brack k}$ ,  $|\mathscr{F}| = {y \brack k}$ , and  $|K_{k+1}^k(\mathscr{F})| = {y \brack k+1}$ . Choose  $v \in {V \brack 1}$  for which  $d(v) \neq 0$ . Since  $|K_{k+1}^k(\mathscr{F})| = {y \brack k+1}$ , we have  $d(v) = {y-1 \brack k-1}$  and  $|K_{k+1}^k(v)| = {y-1 \brack k}$ . If  $U \in {V \brack k-1}$  does not contain v, then  $|L_U(v)| = d(v) = {y-1 \brack k-1}$ . Hence,

$$\begin{bmatrix} y-1\\k \end{bmatrix} = |K_{k+1}^k(v)| \le |K_k^{k-1}(L_U(v))| \le \begin{bmatrix} y-1\\k \end{bmatrix},$$

which implies that  $|K_k^{k-1}(L_U(v))| = {y-1 \brack k}$ . Since  $L_U(v) \subset {V \brack k-1}$ , by the induction hypothesis,  $L_U(v) = {W \brack k-1}$  for some (y-1)-dimensional space W, which implies  $y \in \mathbb{Z}^+$ .

Moreover, for every *k*-dimensional subspace *A* in  $K_k^{k-1}(L_U(v)) = \begin{bmatrix} W \\ k \end{bmatrix}$ , we have that  $A \lor v$  is an element of  $K_{k+1}^k(v)$ . Hence all *k*-dimensional subspaces in  $Y := W \lor v$  lie in  $\mathscr{F}$ . Since  $|\mathscr{F}| = \begin{bmatrix} y \\ k \end{bmatrix}$  and dim(Y) = y, we must have  $\mathscr{F} = \begin{bmatrix} Y \\ k \end{bmatrix}$ .

## 5.2 Analog of Frankl's *r*-wise Intersection Theorem

In this section, we prove the bound in Theorem 5.0.14 and characterize equality when (r-1)n > rk. The proof proceeds by induction on  $(r-1)n - rk \in \mathbb{N}$ . For the base case (r-1)n - rk = 0, we generalize Greene and Kleitman's argument in Section 4.6.2.

**Definition 5.2.1.** A family  $\mathscr{F} \subset \begin{bmatrix} V \\ k \end{bmatrix}$  is <u>*r*-wise co-intersecting</u> if any *r* elements of  $\mathscr{F}$  are contained in a common (n-1)-dimensional space.

Suppose  $r, n, k \in \mathbb{Z}^+$  satisfy (r-1)n - rk = 0 and let  $\mathscr{F} \subset \begin{bmatrix} V \\ k \end{bmatrix}$  be an *r*-wise intersecting family. Endow *V* with the usual inner product, and consider the family

$$\mathscr{F}^{\perp} := \{F^{\perp} : F \in \mathscr{F}\} \subset \begin{bmatrix} V\\ n-k \end{bmatrix}.$$

Let  $\mathscr{B}$  be a geometric (n-k)-spread of V. We want to determine the maximum number of elements of  $\mathscr{B}$  that lie in  $\mathscr{F}^{\perp}$ . Since  $\mathscr{F}$  is *r*-wise intersecting, we have that  $\mathscr{F}^{\perp}$  is *r*-wise co-intersecting. If r = 2 and n = 2k, the family  $\mathscr{F}^{\perp}$  is both intersecting and co-intersecting; hence only one element of the spread  $\mathscr{B}$  can lie in  $\mathscr{F}^{\perp}$  in this case. Lemma 5.2.2 determines the maximum number of elements of  $\mathscr{B}$  that lie in  $\mathscr{F}^{\perp}$ whenever  $r, n, k \in \mathbb{Z}^+$  satisfy (r-1)n - rk = 0.

**Lemma 5.2.2.** Let  $r, n, k \in \mathbb{Z}^+$  satisfy (r-1)n - rk = 0. Suppose that  $\mathscr{B}$  is a geometric (n-k)-spread of V. If  $\mathscr{B}' \subset \mathscr{B}$  is a r-wise co-intersecting subfamily, then

$$|\mathscr{B}'| \leq \frac{q^{(r-1)(n-k)}-1}{q^{n-k}-1}$$

If equality holds,  $\mathscr{B}'$  is a (n-k)-spread of a (r-1)(n-k)-dimensional space.

*Proof.* Let  $B_1, \ldots, B_m$  be a maximum subfamily of  $\mathscr{B}'$  with dim  $(\bigvee_{i=1}^m B_i) = m(n-k)$ . Hence, if  $B \in \mathscr{B}'$  then  $B \cap \bigvee_{i=1}^m B_i \neq \{0\}$ . Since  $\mathscr{B}$  is geometric,  $\mathscr{B}$  induces a spread on  $\bigvee_{i=1}^{m} B_i$  by Lemma 4.5.5. As  $B \cap \bigvee_{i=1}^{m} B_i \neq \{0\}$  for every B in  $\mathscr{B}'$ , all elements in  $\mathscr{B}'$  lie in  $\bigvee_{i=1}^{m} B_i$ . Since  $\mathscr{B}'$  is *r*-wise co-intersecting, we must have  $m \leq r-1$ . Therefore,

$$|\mathscr{B}'| \leq \frac{q^{(r-1)(n-k)} - 1}{q^{n-k} - 1},$$

which is the number of elements in a (n-k)-spread of a (r-1)(n-k)-dimensional space. Also, if equality holds,  $\mathscr{B}'$  is a (n-k)-spread of a (r-1)(n-k)-dimensional space.

Now we prove the base case of Theorem 5.0.14; the case r = 2 of Lemma 5.2.3 is the result of Greene and Kleitman that was presented in Section 4.6.2.

**Lemma 5.2.3.** Suppose  $r, n, k \in \mathbb{Z}^+$  satisfy (r-1)n - rk = 0. If  $\mathscr{F} \subset \begin{bmatrix} V \\ k \end{bmatrix}$  is r-wise intersecting, then  $|\mathscr{F}| \leq \begin{bmatrix} n-1 \\ k-1 \end{bmatrix}$ .

*Proof.* Let  $\mathscr{B}$  be a geometric (n-k)-spread of V and let  $\pi \in GL(V)$  be an isomorphism. By Lemma 4.5.7, the spread  $\pi(\mathscr{B})$  is also geometric. Consider the family  $\mathscr{F}^{\perp} \subset \begin{bmatrix} V \\ n-k \end{bmatrix}$ . Since  $\mathscr{F}$  is *r*-wise intersecting,  $\mathscr{F}^{\perp}$  is *r*-wise co-intersecting. By Lemma 5.2.2,

$$|\mathscr{F}^{\perp} \cap \pi(\mathscr{B})| \le \frac{q^{(r-1)(n-k)} - 1}{q^{n-k} - 1} = \frac{q^k - 1}{q^{n-k} - 1}$$
(5.2.1)

because  $\mathscr{F}^{\perp} \cap \pi(\mathscr{B})$  is a *r*-wise co-intersecting subfamily of  $\pi(\mathscr{B})$  and because we have k = (r-1)(n-k) when r, n, k satisfy (r-1)n - rk = 0.

By Lemma 4.6.2, we have  $|GL(V)| = q^{n(n-1)/2}(q-1)^n[n]!$ , so

$$\sum_{\pi \in GL(V)} |\mathscr{F}^{\perp} \cap \pi(\mathscr{B})| \leq \frac{q^k - 1}{q^{n-k} - 1} \cdot q^{n(n-1)/2} (q-1)^n [n]!.$$

Now, given  $F^{\perp} \in \mathscr{F}^{\perp}$  and  $B \in \mathscr{B}$  there are  $q^{n(n-1)/2}(q-1)^n[n-k]![k]!$  isomorphisms  $\pi \in GL(V)$  such that  $\pi(B) = F^{\perp}$  by Lemma 4.6.3. Consequently,

$$\begin{split} & \left(\frac{q^n - 1}{q^{n-k} - 1}\right) |\mathscr{F}^{\perp}| \ q^{n(n-1)/2} (q-1)^n [n-k]! [k]! \\ = & |\mathscr{B}| \ |\mathscr{F}^{\perp}| \ |\{\pi \in GL(V) : \pi(B) = F^{\perp}\}| \\ = & \sum_{\pi \in GL(V)} |\mathscr{F}^{\perp} \cap \pi(\mathscr{B})| \\ \leq & \frac{q^k - 1}{q^{n-k} - 1} \cdot \ q^{n(n-1)/2} (q-1)^n [n]!. \end{split}$$

Since  $|\mathscr{F}| = |\mathscr{F}^{\perp}|$ , we have

$$|\mathscr{F}| \le \left(\frac{q^{n(n-1)/2}(q-1)^n [n]!}{q^{n(n-1)/2}(q-1)^n [n-k]! [k]!}\right) \left(\frac{q^{n-k}-1}{q^n-1}\right) \left(\frac{q^k-1}{q^{n-k}-1}\right) = {n-1 \brack k-1}. \quad \Box$$

Now we prove the bound in Theorem 5.0.14 and characterize equality when (r-1)n > rk.

**Proof of Theorem 5.0.14.** The proof proceeds by induction on  $(r-1)n - rk \in \mathbb{N}$ . The base case (r-1)n - rk = 0 was proved in Lemma 5.2.3. Suppose Theorem 5.0.14 holds when r, n, k satisfy (r-1)n - rk = p for  $p \ge 0$ . We will prove Theorem 5.0.14 holds when r, n, k satisfy (r-1)n - rk = p + 1. Let  $\mathscr{F} \subset \begin{bmatrix} V \\ k \end{bmatrix}$  be a maximum size r-wise intersecting family. Now the family  $\mathscr{P} := \{P \in \begin{bmatrix} V \\ k \end{bmatrix} : v \subset P\}$ , where  $v \subset V$  is some one-dimensional subspace, is r-wise intersecting so  $|\mathscr{F}| \ge |\mathscr{P}| = \begin{bmatrix} n-1 \\ k-1 \end{bmatrix}$ . Let W be an (n+1)-dimensional space over  $\mathbb{F}_q$  that contains V. Define the family

$$\mathscr{A} := \left\{ A \in \begin{bmatrix} W \\ k+1 \end{bmatrix} : \exists F \in \mathscr{F} \text{ with } F \subset A \right\}$$

to be the family of all (k+1)-dimensional spaces in W that contain some  $F \in \mathscr{F}$ . We will partition  $\mathscr{A}$  into the following subfamilies:

$$\mathscr{A}_1 := \{ A \in \mathscr{A} : A \not\subset V \}, \quad \mathscr{A}_2 := \mathscr{A} \setminus \mathscr{A}_1.$$

First let us compute the size of  $\mathscr{A}_1$ . Observe that if  $A \in \begin{bmatrix} W \\ k+1 \end{bmatrix}$  and A does not lie in V, then A intersects V in exactly a k-dimensional space. Therefore, A cannot contain two distinct k-dimensional spaces in  $\mathscr{F}$ . Note that any  $F \in \mathscr{F}$  can be extended to a (k+1)-dimensional space in  $\mathscr{A}_1$  in  $q^{n-k}$  ways. Therefore,  $|\mathscr{A}_1| = q^{n-k} |\mathscr{F}| \ge q^{n-k} {n-1 \brack k-1}$ .

Now we will compute the size of  $\mathscr{A}_2$ . By duality, we have  $F \subset A \in \mathscr{A}_2$  for some  $F \in \mathscr{F}$  if and only if  $F^{\perp} \supset A^{\perp} \in {V \brack n-k-1}$ . Therefore,  $|\mathscr{A}_2| = |\partial \mathscr{F}^{\perp}|$ . Since

$$|\mathscr{F}^{\perp}| = |\mathscr{F}| \ge {\binom{n-1}{k-1}} = {\binom{n-1}{n-k}}, \tag{5.2.2}$$

Theorem 5.0.12 yields

$$|\mathscr{A}_{2}| = \left|\partial\mathscr{F}^{\perp}\right| \ge {n-1 \choose n-k-1} = {n-1 \choose k}.$$
(5.2.3)

As  $\mathscr{A} = \mathscr{A}_1 \dot{\cup} \mathscr{A}_2$ , we have by Lemma 4.3.1 that

$$|\mathscr{A}| = |\mathscr{A}_1| + |\mathscr{A}_2| \ge q^{n-k} {n-1 \choose k-1} + {n-1 \choose k} = {n \choose k}.$$
(5.2.4)

Since  $\mathscr{F}$  is *r*-wise intersecting,  $\mathscr{A}$  is an *r*-wise intersecting family of (k+1)-dimensional spaces in *W*. Observe that r, n+1, k+1 satisfy

$$(r-1)(n+1) - r(k+1) = (r-1)n - rk - 1 = (p+1) - 1 = p.$$

By the induction hypothesis  $|\mathscr{A}| \leq {n \choose k}$ , which implies equality everywhere in (5.2.2), (5.2.3), and (5.2.4). Hence,  $q^{n-k}|\mathscr{F}| = |\mathscr{A}_1| = q^{n-k} {n-1 \choose k-1}$  and  $|\mathscr{F}| = {n-1 \choose k-1}$ . Moreover,  $|\mathscr{F}^{\perp}| = {n-1 \choose n-k}$  and  $|\partial \mathscr{F}^{\perp}| = |\mathscr{A}_2| = {n-1 \choose n-k-1}$ . Therefore  $\mathscr{F}^{\perp}$  satisfies equality in Theorem 5.0.12, which implies that  $\mathscr{F}^{\perp} = {Y \choose n-k}$  for some (n-1)-dimensional subspace  $V \subset V$ . By duality,  $\mathscr{F} = \{F \in {V \choose k} : v \subset F\}$  for some one-dimensional subspace  $v \subset V$ .

## **5.3** Characterizing Equality in the Base Case

We characterize equality in Theorem 5.0.14 when (r-1)n - rk = 0. Recall from Section 4.4 that Godsil and Newman characterized equality in the Erdős-Ko-Rado theorem for vector spaces using the methods of [60]. In particular, they showed

**Theorem 5.3.1** (Godsil and Newman). If n = 2k and  $\mathscr{F} \subset \begin{bmatrix} V \\ k \end{bmatrix}$  is an intersecting family of maximum size, then  $\mathscr{F} = \{F \in \begin{bmatrix} V \\ k \end{bmatrix} : v \subset F\}$  for some one-dimensional subspace  $v \subset V$  or  $\mathscr{F} = \begin{bmatrix} U \\ k \end{bmatrix}$  for some (2k-1)-dimensional subspace  $U \subset V$ .

We use their result to characterize equality in Theorem 5.0.14 when  $r \ge 3$  and (r-1)n - rk = 0. The proof proceeds by induction on r; the base case r = 2 and n = 2k is Theorem 5.3.1. Let  $\mathscr{F} \subset \begin{bmatrix} V \\ k \end{bmatrix}$  be a maximum size r-wise intersecting family. In this section, it will be more natural to state results in terms of  $\mathscr{F}^{\perp} \subset \begin{bmatrix} V \\ n-k \end{bmatrix}$  so we make the following simple observation.

**Lemma 5.3.2.** We have  $\mathscr{F} \subset \begin{bmatrix} V \\ k \end{bmatrix}$  is a maximum size *r*-wise intersecting family if and only if  $\mathscr{F}^{\perp} \subset \begin{bmatrix} V \\ n-k \end{bmatrix}$  is a maximum size *r*-wise co-intersecting family.  $\Box$ 

Lemma 5.3.5 allows us to use induction. We first state two simple corollaries of Lemma 5.2.3 that will be used in the proof of Lemma 5.3.5. Since r,n,k satisfy (r-1)n-rk = 0, note that V is r(n-k)-dimensional.

**Corollary 5.3.3.** Suppose r, n, k satisfy (r-1)n - rk = 0 and that  $\mathscr{F} \subset \begin{bmatrix} V \\ k \end{bmatrix}$  is an r-wise intersecting family. If there is a geometric (n-k)-spread  $\mathscr{B}$  of V such that equality holds in (5.2.1) for all  $\pi \in GL(V)$ , then  $\mathscr{F}$  has maximum size.  $\Box$ 

**Corollary 5.3.4.** Suppose r, n, k satisfy (r-1)n - rk = 0. If  $\mathscr{F} \subset \begin{bmatrix} V \\ k \end{bmatrix}$  is a maximum size *r*-wise intersecting family, then equality holds in (5.2.1) for every  $\pi \in GL(V)$  and every geometric (n-k)-spread  $\mathscr{B}$  of V.

**Lemma 5.3.5.** Let  $\mathscr{F} \subset {V \brack k}$  be a maximum size *r*-wise intersecting family. Fix  $F^{\perp}$  in  $\mathscr{F}^{\perp}$  and let  $U \subset V$  be an (r-1)(n-k)-dimensional space that intersects  $F^{\perp}$  trivially; that is  $F^{\perp} \cap U = \{0\}$ . Then

$$\mathscr{F}^{\perp}|_U := \{ E \in \mathscr{F}^{\perp} : E \subset U \}$$

is a maximum size (r-1)-wise co-intersecting family in  $\begin{bmatrix} U\\ n-k \end{bmatrix}$ .

Proof. Let  $\mathscr{S}$  be a geometric (n-k)-spread of V. Choose  $S_1, \ldots, S_r$  in  $\mathscr{S}$  such that  $\bigvee_{i=1}^r S_i = V$ . Since  $F^{\perp} \cap U = \{0\}$ , there exists an isomorphism  $\rho \in GL(V)$  such that  $\rho(S_1) = F^{\perp}$  and  $\rho(\bigvee_{i=2}^r S_i) = U$ . The (n-k)-spread  $\mathscr{B} := \rho(\mathscr{S})$  is geometric by Lemma 4.5.7, and  $F^{\perp} \in \mathscr{B}$ . Moreover  $U = \bigvee_{i=2}^r \rho(S_i)$  so, by Lemma 4.5.5, we have that  $\mathscr{B}$  induces a geometric (n-k)-spread  $\mathscr{B}'$  on U.

Observe that  $\mathscr{F}^{\perp}|_U$  is (r-1)-wise co-intersecting since  $F^{\perp} \cap U = \{0\}$ . To prove that  $\mathscr{F}^{\perp}|_U \subset \begin{bmatrix} U\\ n-k \end{bmatrix}$  is a maximum size (r-1)-wise co-intersecting family, we will apply Lemma 5.3.2 and Corollary 5.3.3. That is, we will show that if  $\alpha \in GL(U)$  then equality holds in (5.2.1):

$$\left|\mathscr{F}^{\perp}|_{U}\cap \alpha(\mathscr{B}')\right|=rac{q^{(r-2)(n-k)}-1}{q^{n-k}-1}.$$

Let  $\pi \in GL(V)$  be an invertible linear transformation such that  $\pi(F^{\perp}) = F^{\perp}$ ,  $\pi(U) = U$ , and  $\pi|_U = \alpha$ . Since  $\mathscr{F}^{\perp}$  is a maximum size *r*-wise co-intersecting family,  $\mathscr{F}^{\perp} \cap \pi(\mathscr{B})$  is an (n-k)-spread of an (r-1)(n-k)-dimensional space  $W_{\pi}$  by Lemma 5.2.2 and Corollary 5.3.4. We have that  $F^{\perp}$  is contained in  $W_{\pi}$  and intersects U trivially so dim $(W_{\pi} \cap U) = (r-2)(n-k)$ .

The spread  $\pi(\mathscr{B})$  induces the spread  $\mathscr{F}^{\perp} \cap \pi(\mathscr{B})$  on  $W_{\pi}$  and induces the spread  $\alpha(\mathscr{B}')$  on U. Consider the elements of  $\alpha(\mathscr{B}')$  that intersect  $W_{\pi} \cap U$  nontrivially; as these elements are in  $\pi(\mathscr{B})$  and intersect  $W_{\pi}$ , they must lie in  $W_{\pi}$  and hence in  $W_{\pi} \cap U$ . Hence, the elements of  $\alpha(\mathscr{B}')$  that intersect  $W_{\pi} \cap U$  nontrivially form a spread of  $W_{\pi} \cap U$ . Moreover, these elements lie in  $\mathscr{F}^{\perp} \cap \pi(\mathscr{B})$  so

$$\mathscr{F}^{\perp}|_U \cap \alpha(\mathscr{B}') = (\mathscr{F}^{\perp} \cap \pi(\mathscr{B})) \cap \alpha(\mathscr{B}')$$

is the spread  $\pi(\mathscr{B})$  induces on  $W_{\pi} \cap U$ . Since  $W_{\pi} \cap U$  is (r-2)(n-k)-dimensional,  $|\mathscr{F}^{\perp}|_U \cap \alpha(\mathscr{B}')|$  satisfies (5.2.1) with equality. By Lemma 5.3.2 and Corollary 5.3.3,  $\mathscr{F}^{\perp}|_U$  is a maximum size (r-1)-wise co-intersecting family in  $\begin{bmatrix} U\\n-k \end{bmatrix}$ .

Characterizing Equality in Theorem 5.0.14 when (r-1)n - rk = 0 and  $r \ge 3$ : We now characterize equality in Theorem 5.0.14 when (r-1)n - rk = 0 and  $r \ge 3$ . The proof proceeds by induction on *r*; the base case r = 2 and n = 2k is Theorem 5.3.1.

Let  $r \ge 3$  and suppose the statement is proved for any  $2 \le r' < r$ . Let  $\mathscr{F} \subset \begin{bmatrix} V \\ k \end{bmatrix}$  be a maximum size *r*-wise intersecting family and observe that  $\mathscr{F}^{\perp} \subset \begin{bmatrix} V \\ n-k \end{bmatrix}$  is a maximum size *r*-wise co-intersecting family. Our objective is to show that  $\mathscr{F}^{\perp} = \begin{bmatrix} H \\ n-k \end{bmatrix}$ , where *H* is a (n-1)-dimensional space of *V*. By duality, this implies that  $\mathscr{F} = \{F \in \begin{bmatrix} V \\ k \end{bmatrix} : v \subset F\}$ for some one-dimensional subspace  $v \subset V$ , which is the desired conclusion.

Fix some  $F^{\perp} \in \mathscr{F}^{\perp}$ . By Lemma 5.3.5, if *U* is a (r-1)(n-k)-dimensional subspace that intersects  $F^{\perp}$  trivially, then  $\mathscr{F}^{\perp}|_{U}$  is a maximum size (r-1)-wise co-intersecting family in  $\begin{bmatrix} U\\n-k \end{bmatrix}$ . When r = 3, then dim U = 2(n-k) and  $\mathscr{F}^{\perp}|_{U}$  is a maximum size intersecting and co-intersecting family in  $\begin{bmatrix} U\\n-k \end{bmatrix}$ ; hence by Theorem 5.3.1

- 1.  $\mathscr{F}^{\perp}|_U = \{E \in \begin{bmatrix} U \\ n-k \end{bmatrix} : u \subset E\}$  for some one-dimensional subspace  $u \subset U$  or
- 2.  $\mathscr{F}^{\perp}|_U = \begin{bmatrix} U'\\ n-k \end{bmatrix}$  for some (2(n-k)-1)-dimensional subspace  $U' \subset U$ .

If r > 3 then, by the induction hypothesis and duality,  $\mathscr{F}^{\perp}|_U = \begin{bmatrix} U'\\n-k \end{bmatrix}$ , where  $U' \subset U$  is some ((r-1)(n-k)-1)-dimensional subspace.

Our first task is to eliminate the possibility that  $\mathscr{F}^{\perp}|_U = \{E \in \begin{bmatrix} U\\ n-k \end{bmatrix} : u \subset E\}$  for some one-dimensional subspace  $u \subset U$  in the case r = 3. We now show that if

 $\mathscr{F}^{\perp}|_{U} = \{E \in \begin{bmatrix} U\\ n-k \end{bmatrix} : u \subset E\}$  for some one-dimensional subspace  $u \subset U$ , then every element of  $\mathscr{F}^{\perp}$  must intersect  $F^{\perp} \lor u$  nontrivially.

**Claim 5.3.6.** If  $\mathscr{F}^{\perp}|_U = \{E \in \begin{bmatrix} U \\ n-k \end{bmatrix} : u \subset E\}$  for some one-dimensional subspace  $u \subset U$ , then for all  $G \in \mathscr{F}^{\perp}$  we have  $G \cap (F^{\perp} \lor u) \neq \{0\}$ .

*Proof.* Suppose, for a contradiction, that there exists  $G \in \mathscr{F}^{\perp}$  such that G intersects  $F^{\perp} \lor u$  trivially. We have  $\dim((F^{\perp} \lor G) \cap U) = n - k$  because  $F^{\perp}$  intersects both G and U trivially. Since u does not lie in  $F^{\perp} \lor G$  and  $\mathscr{F}^{\perp}|_U = \{E \in \begin{bmatrix} U \\ n-k \end{bmatrix} : u \subset E\}$ , we can find  $E' \in \mathscr{F}^{\perp}|_U$  that intersects  $F^{\perp} \lor G$  trivially. Hence  $F^{\perp} \lor G \lor E' = V$ , which contradicts the fact that  $\mathscr{F}^{\perp}$  is 3-wise co-intersecting.

We now show that if  $\mathscr{F}^{\perp}|_U = \{E \in \begin{bmatrix} U\\ n-k \end{bmatrix} : u \subset E\}$  for some one-dimensional subspace  $u \subset U$ , then any (n-k)-dimensional space that meets  $F^{\perp}$  trivially but meets  $F^{\perp} \lor u$  nontrivially must lie in  $\mathscr{F}^{\perp}$ .

**Claim 5.3.7.** Suppose  $\mathscr{F}^{\perp}|_{U} = \{E \in \begin{bmatrix} U \\ n-k \end{bmatrix} : u \subset E\}$  for some one-dimensional subspace  $u \subset U$ . If  $G \in \begin{bmatrix} V \\ n-k \end{bmatrix}$ ,  $G \cap F^{\perp} = \{0\}$ , and  $G \cap (F^{\perp} \lor u) \neq \{0\}$ , then  $G \in \mathscr{F}^{\perp}$ .

Proof. There exists a geometric (n-k)-spread  $\mathscr{B}$  of V that contains both G and  $F^{\perp}$  because  $G \cap F^{\perp} = \{0\}$ . As  $\mathscr{B}$  is a spread, all subspaces in  $(\mathscr{F}^{\perp} \cap \mathscr{B}) \setminus \{F^{\perp}\}$  meet  $F^{\perp} \lor u$  in a one-dimensional subspace that does not lie in  $F^{\perp}$  by Claim 5.3.6. Lemma 5.2.2 and Corollary 5.3.4 imply that  $\mathscr{F}^{\perp} \cap \mathscr{B}$  is a spread of a 2(n-k)-dimensional space so  $|(\mathscr{F}^{\perp} \cap \mathscr{B}) \setminus \{F^{\perp}\}| = q^{n-k}$ . There are  $q^{n-k}$  one-dimensional subspaces in  $F^{\perp} \lor u$  that do not lie in  $F^{\perp}$ . Hence, each one-dimensional subspace in  $(F^{\perp} \lor u) \setminus F^{\perp}$  meets a unique subspace in  $(\mathscr{F}^{\perp} \cap \mathscr{B}) \setminus \{F^{\perp}\}$ . Since G meets  $F^{\perp} \lor u$  in a one-dimensional subspace that does not lie in  $F^{\perp}$  and  $G \in \mathscr{B}$ , we must have  $G \in \mathscr{F}^{\perp} \cap \mathscr{B} \subset \mathscr{F}^{\perp}$ .

We now eliminate the possibility that  $\mathscr{F}^{\perp}|_{U} = \{E \in \begin{bmatrix} U\\ n-k \end{bmatrix} : u \subset E\}$  for some onedimensional subspace  $u \subset U$ . We will construct three (n-k)-dimensional subspaces that together span V, and intersect  $F^{\perp} \lor u$  in a one-dimensional subspace not lying in  $F^{\perp}$ . By Claim 5.3.7, these three spaces lie in  $\mathscr{F}^{\perp}$ , which contradicts  $\mathscr{F}^{\perp}$  being 3-wise co-intersecting. To build these three subspaces, we first choose three one-dimensional subspaces  $v_1^1, v_2^1, v_3^1$  in  $(F^{\perp} \lor u) \setminus F^{\perp}$  such that  $v_3^1 \not\subset v_1^1 \lor v_2^1$ . These one-dimensional subspaces exist because  $\dim(F^{\perp} \lor u) = (n-k) + 1 \ge 3$  so, after picking  $v_1^1$  and  $v_2^1$ , any one-dimensional subspace of  $F^{\perp} \lor u$  not in  $F^{\perp} \cup (v_1^1 \lor v_2^1)$  will do. Since the number of one-dimensional subspaces in  $(F^{\perp} \lor u) \setminus (F^{\perp} \cup (v_1^1 \lor v_2^1))$  is  $q^{n-k} - q > 0$ , we can indeed choose  $v_3^1$ .

We construct a family of one-dimensional subspaces

$$\{v_i^j: i \in \{1, 2, 3\}, j \in \{1, \dots, n-k\}\}$$

such that, for each  $i \in \{1,2,3\}$ , the subspace  $V_i = \bigvee_{j=1}^{n-k} v_i^j$  intersects  $F^{\perp} \lor u$  in the onedimensional subspace  $v_i^1 \not\subset F^{\perp}$  and  $\bigvee_{i=1}^3 V_i = V$ . The subspaces  $V_1, V_2, V_3$  are the desired three (n-k)-dimensional subspaces. We pick the one-dimensional subspaces one after the other; we have to show that at each step there is a possible one-dimensional subspace to pick. When picking the last one-dimensional subspace  $v_3^{n-k}$  we must choose a onedimensional subspace from V that is not in  $V_1 \lor V_2 \lor \bigvee_{j=1}^{n-k-1} v_j^j$  nor in  $F^{\perp} \lor \bigvee_{j=1}^{n-k-1} v_j^j$ . By inclusion-exclusion, there are  $q^{3(n-k)-1} - q^{2(n-k)-2} > 0$  one-dimensional subspaces in V that do not lie in  $V_1 \lor V_2 \lor \bigvee_{j=1}^{n-k-1} v_j^j$  nor in  $F^{\perp} \lor \bigvee_{j=1}^{n-k-1} v_3^j$ ; thus it is indeed possible to construct the desired three (n-k)-dimensional subspaces. Therefore, we have eliminated the possibility that  $\mathscr{F}^{\perp}|_U = \{E \in [\bigcup_{n-k}^{U}] : u \subset E\}$  for some one-dimensional subspace  $u \subset U$  in the case r = 3.

We may now assume that  $r \ge 3$  and that if U is a (r-1)(n-k)-dimensional space that intersects  $F^{\perp}$  trivially then  $\mathscr{F}^{\perp}|_{U} = \begin{bmatrix} U'\\ n-k \end{bmatrix}$  for some ((r-1)(n-k)-1)-dimensional subspace  $U' \subset U$ . Our ultimate goal is to prove that  $\mathscr{F}^{\perp} = \begin{bmatrix} F^{\perp} \lor U'\\ n-k \end{bmatrix}$ . Naturally, we first show that if  $U_1, U_2$  are two (r-1)(n-k)-dimensional subspaces that intersect  $F^{\perp}$ trivially, then  $F^{\perp} \lor U'_1 = F^{\perp} \lor U'_2$ .

**Claim 5.3.8.** Let  $U_1, U_2$  be two (r-1)(n-k)-dimensional subspaces of V that intersect  $F^{\perp}$  trivially. Let  $U'_1, U'_2$  be the ((r-1)(n-k)-1)-dimensional subspaces of  $U_1$  and  $U_2$  such that  $\mathscr{F}^{\perp}|_{U_1} = \begin{bmatrix} U'_1 \\ n-k \end{bmatrix}$  and  $\mathscr{F}^{\perp}|_{U_2} = \begin{bmatrix} U'_2 \\ n-k \end{bmatrix}$ . Then  $F^{\perp} \vee U'_1 = F^{\perp} \vee U'_2$ .

*Proof.* Suppose, for a contradiction, that  $F^{\perp} \vee U'_1 \neq F^{\perp} \vee U'_2$ . We choose subspaces  $W_1, \ldots, W_{r-2}$  in  $\begin{bmatrix} U'_1 \\ n-k \end{bmatrix}$  such that dim  $\left(\bigvee_{i=1}^{r-2} W_i\right) = (r-2)(n-k)$  and  $W_1$  is not contained in  $F^{\perp} \vee U'_2$ .

The subspace  $F^{\perp} \vee \bigvee_{i=1}^{r-2} W_i$  is (r-1)(n-k)-dimensional because  $U_1$  intersects  $F^{\perp}$  trivially. The subspace  $U'_2$  is ((r-1)(n-k)-1)-dimensional and intersects  $F^{\perp}$ 

trivially so

$$(r-2)(n-k)-1 \leq \dim\left(U_2' \cap \left(F^{\perp} \lor \bigvee_{i=1}^{r-2} W_i\right)\right) \leq (r-2)(n-k).$$

Suppose that dim  $(U'_2 \cap (F^{\perp} \vee \bigvee_{i=1}^{r-2} W_i)) = (r-2)(n-k)$  for a contradiction. By definition of  $W_1$ , we can choose a one-dimensional subspace  $w \subset W_1$  that does not lie in  $F^{\perp} \vee U'_2$ . The subspace  $F^{\perp} \vee w$  is (n-k+1)-dimensional. The subspace  $F^{\perp} \vee \bigvee_{i=1}^{r-2} W_i$  is (r-1)(n-k)-dimensional and contains  $F^{\perp} \vee w$ . Note that  $F^{\perp} \vee w$  must intersect  $U'_2$  nontrivially if dim  $(U'_2 \cap (F^{\perp} \vee \bigvee_{i=1}^{r-2} W_i)) = (r-2)(n-k)$ . This is a contradiction because w does not lie in  $F^{\perp} \vee U'_2$  by construction. We therefore conclude that dim  $(U'_2 \cap (F^{\perp} \vee \bigvee_{i=1}^{r-2} W_i)) = (r-2)(n-k) - 1$ .

Since  $U'_2$  is ((r-1)(n-k)-1)-dimensional, this implies that there exists a subspace Z in  $\begin{bmatrix} U'_2 \\ n-k \end{bmatrix}$  that intersects  $F^{\perp} \lor \bigvee_{i=1}^{r-2} W_i$  trivially. Now  $F^{\perp}, W_1, \dots, W_{r-2}, Z$  lie in  $\mathscr{F}^{\perp}$  since  $\mathscr{F}^{\perp}|_{U_1} = \begin{bmatrix} U'_1 \\ n-k \end{bmatrix}$  and  $\mathscr{F}^{\perp}|_{U_2} = \begin{bmatrix} U'_2 \\ n-k \end{bmatrix}$ . By construction,  $F^{\perp} \lor \bigvee_{i=1}^{r-2} W_i \lor Z = V$ , which contradicts  $\mathscr{F}^{\perp}$  being *r*-wise co-intersecting. This proves  $F^{\perp} \lor U'_1 = F^{\perp} \lor U'_2$ .  $\Box$ 

Now we show that any (n-k)-dimensional subspace in  $F^{\perp} \vee U'$  that intersects  $F^{\perp}$  trivially must lie in  $\mathscr{F}^{\perp}$ .

**Claim 5.3.9.** If 
$$G \in \begin{bmatrix} F^{\perp} \lor U' \\ n-k \end{bmatrix}$$
 and  $G \cap F^{\perp} = \{0\}$ , then  $G \in \mathscr{F}^{\perp}$ .

*Proof.* Since  $G \cap F^{\perp} = \{0\}$ , there exists a (r-1)(n-k)-dimensional subspace U(G) that contains G and intersects  $F^{\perp}$  trivially. Let U(G)' be the ((r-1)(n-k)-1)-dimensional subspace of U(G) such that  $\mathscr{F}^{\perp}|_{U(G)} = \begin{bmatrix} U(G)'\\ n-k \end{bmatrix}$ . By Claim 5.3.8,

$$G \subset (F^{\perp} \lor U') \cap U(G) = (F^{\perp} \lor U(G)') \cap U(G) = U(G)'.$$

Hence  $G \in \begin{bmatrix} U(G)' \\ n-k \end{bmatrix} \subset \mathscr{F}^{\perp}$ .

Now we are ready to prove  $\mathscr{F}^{\perp} = \begin{bmatrix} F^{\perp} \lor U' \\ n-k \end{bmatrix}$ . Suppose, for a contradiction, that there exists a subspace  $H \in \mathscr{F}^{\perp}$  that is not in  $\begin{bmatrix} F^{\perp} \lor U' \\ n-k \end{bmatrix}$ . We will construct r-1 subspaces in  $\begin{bmatrix} F^{\perp} \lor U' \\ n-k \end{bmatrix}$  that each intersect  $F^{\perp}$  trivially and that together with H span V. By Claim 5.3.9, these r-1 subspaces lie in  $\mathscr{F}^{\perp}$  which contradicts  $\mathscr{F}^{\perp}$  being r-wise co-intersecting.

To build these r - 1 subspaces, we construct a family of one-dimensional subspaces

$$\{v_i^j : i \in \{1, \dots, r-1\}, j \in \{1, \dots, n-k\}\}$$

such that for each  $i \in \{1, ..., r-1\}$ , the subspace  $G_i = \bigvee_{j=1}^{n-k} v_i^j$  lies in  $F^{\perp} \lor U'$ , intersects  $F^{\perp}$  trivially, and  $\bigvee_{i=1}^{r-1} G_i \lor H = V$ . The subspaces  $G_1, ..., G_{r-1}$  are the desired r-1 subspaces. We pick the one-dimensional subspaces one after the other; we have to show that at each step there is a possible one-dimensional subspace to pick. When picking the last one-dimensional subspace  $v_{r-1}^{n-k}$  we must pick a one-dimensional subspace from  $F^{\perp} \lor U'$  that is not in  $H \lor \bigvee_{i=1}^{r-2} G_i \lor \bigvee_{j=1}^{n-k-1} v_{r-1}^j$  nor in  $F^{\perp} \lor \bigvee_{j=1}^{n-k-1} v_{r-1}^j$ . Since H is not contained in  $F^{\perp} \lor U'$ , we have

$$\dim\left(\left(F^{\perp} \vee U'\right) \cap \left(H \vee \bigvee_{i=1}^{r-2} G_i \vee \bigvee_{j=1}^{n-k-1} v_{r-1}^j\right)\right) = r(n-k) - 2.$$

Hence, there are at least

$$q^{r(n-k)-2} - (q^{2(n-k)-2} + q^{2(n-k)-3} + \dots + 1) > 0$$

one-dimensional subspaces of  $F^{\perp} \vee U'$  that do not lie in  $H \vee \bigvee_{i=1}^{r-2} G_i \vee \bigvee_{j=1}^{n-k-1} v_{r-1}^j$  nor in  $F^{\perp} \vee \bigvee_{j=1}^{n-k-1} v_{r-1}^j$ ; thus it is indeed possible to construct the desired r-1 subspaces. This proves that  $\mathscr{F}^{\perp} \subseteq \begin{bmatrix} F^{\perp} \vee U' \\ n-k \end{bmatrix}$ , and since  $|\mathscr{F}^{\perp}| = \begin{bmatrix} n-1 \\ k-1 \end{bmatrix}$  we have  $\mathscr{F}^{\perp} = \begin{bmatrix} F^{\perp} \vee U' \\ n-k \end{bmatrix}$ . The subspace  $F^{\perp} \vee U'$  is (n-1)-dimensional; by duality,  $\mathscr{F} = \{F \in \begin{bmatrix} V \\ k \end{bmatrix} : v \subset F\}$  for some one-dimensional subspace  $v \subset V$ , which is the desired conclusion.

## 5.4 Other Results of the Author

In this section, we discuss the author's other results in this area, namely obtaining an analog of the Hilton-Milner theorem and determining the chromatic number of the q-Kneser graphs. The results will be stated but not proved.

## 5.4.1 Analog of the Hilton-Milner Theorem

Recall that Hilton's and Milner's Theorem 2.2.13 is an extension of the Erdős-Ko-Rado theorem, Theorem 2.2.2, that gives the size of the largest nontrivial intersecting family  $\mathscr{F} \subset {X \choose k}$ . The Hilton-Milner theorem was first proved by Hilton and Milner [70] using lengthy combinatorial arguments; later Frankl and Füredi produced an elegant proof using the shifting technique. Due to the issues discussed in Section 4.6.1, none of the known proofs of the Hilton-Milner theorem readily generalize to vector spaces.

We will see that the similarity between the Hilton-Milner theorem and its vector space analog, Theorem 5.4.1, is remarkable. We stress, however, that the naive analog of the extremal family (i) in Theorem 2.2.13 is *not* maximally intersecting: If  $E \in {V \brack 1}$  is a one-dimensional subspace and  $U \in {V \brack k}$  is a *k*-dimensional subspace that does not contain *E*, then the family

$$\{U\} \cup \left\{W \in \begin{bmatrix} V\\k \end{bmatrix} : E \subset W, \ \dim(W \cap U) \ge 1\right\}$$
(5.4.1)

is not maximally intersecting, as we can add all subspaces in  $\begin{bmatrix} E \lor U \\ k \end{bmatrix}$  that are not in (5.4.1). We will say that  $\mathscr{F}$  is an *HM*-type family if

$$\mathscr{F} = \left\{ W \in \begin{bmatrix} V \\ k \end{bmatrix} : E \subset W, \ \dim(W \cap U) \ge 1 \right\} \cup \begin{bmatrix} E \lor U \\ k \end{bmatrix}$$

for some  $E \in {V \brack 1}$  and  $U \in {V \brack k}$  with  $E \not\subset U$ .

**Theorem 5.4.1** (Blokhuis-Brouwer-Chowdhury-Frankl-Mussche-Patkós-Szőnyi). For  $k \ge 3$ , suppose  $q \ge 3$ ,  $n \ge 2k + 1$  or q = 2,  $n \ge 2k + 2$ . If  $\mathscr{F} \subset \begin{bmatrix} V \\ k \end{bmatrix}$  is an intersecting family and there does not exist  $v \in \begin{bmatrix} V \\ 1 \end{bmatrix}$  such that  $\mathscr{F} \subset \{F \in \begin{bmatrix} V \\ k \end{bmatrix} : v \subset F\}$ , then

$$|\mathscr{F}| \leq {n-1 \choose k-1} - q^{k(k-1)} {n-k-1 \choose k-1} + q^k.$$

Equality holds if and only if

(i)  $\mathscr{F}$  is an HM-type family,

(ii) 
$$\mathscr{F} = \{F \in \begin{bmatrix} V \\ k \end{bmatrix} : \dim(S \cap F) \ge 2\}$$
 for some  $S \in \begin{bmatrix} V \\ 2 \end{bmatrix}$  if  $k = 3$ .

Variants of Theorem 5.4.1 are used to establish our next result on coloring the q-Kneser graphs.

## **5.4.2** Coloring the *q*-Kneser Graph

We discuss the question of coloring the q-Kneser graphs. First, we define what is meant by a graph coloring.

**Definition 5.4.2.** Given a graph G = (V, E) a proper coloring of G with l colors is a map from V to a set of colors with cardinality l such that no two adjacent vertices receive the same color.

If V is finite, then clearly G can be properly colored with |V| colors. We will be interested in the least number of colors required to properly color G.

**Definition 5.4.3.** Given a graph G = (V, E), the <u>chromatic number</u> of G, denoted  $\chi(G)$ , is the least integer  $l \in \mathbb{Z}^+$  such that G has a proper coloring with l colors.

Observe that a set of vertices that receives the same color in a proper coloring must be an independent set. Hence, the chromatic number of a graph is the minimum number of independent sets needed to partition its vertex set V.

Determining the chromatic number of a graph can be very difficult. In 1955, Kneser [78] conjectured that, when  $n \ge 2k$ , the chromatic number of the Kneser graph  $K_{n:k}$  is  $\chi(qK_{n:k}) = n - 2k + 2$ . The problem remained open for twenty three years until Lovász [83] and Bárány [9] found proofs that surprisingly use algebraic topology.

**Theorem 5.4.4** (Bárány, Lovász). *If*  $n \ge 2k$ , the chromatic number of the Kneser graph  $K_{n:k}$  is  $\chi(K_{n:k}) = n - 2k + 2$ .

We can easily see that n - 2k + 2 colors suffice to properly color the Kneser graph  $K_{n:k}$ . If  $\alpha$  is a *k*-subset of  $K_{n:k}$  and its largest element is greater than 2k, define this element to be the color of  $\alpha$ . Thus, the *k*-subsets not contained in  $\{1, \ldots, 2k\}$  can be colored with n - 2k colors. The *k*-subsets not already colored induce a copy of  $K_{2k:k}$ , which is bipartite, so the remaining *k*-subsets can be colored with two colors. This proves that  $\chi(K_{n:k}) \leq n - 2k + 2$ .

Kneser's conjecture motivates the question of coloring the q-Kneser graphs. Not surprisingly, the known proofs of Theorem 5.4.4 do not generalize to the vector space setting. We have seen in Section 4.4 that many parameters of the q-Kneser graph reduce

to the corresponding parameters of the Kneser graph by setting q = 1. Interestingly, we will see that the chromatic number is *not* one of these parameters. The author, Godsil, and Royle [26] determined the chromatic number of the *q*-Kneser graphs  $qK_{n:2}$  when  $n \ge 4$ , and characterized the minimum colorings.

**Theorem 5.4.5** (Chowdhury-Godsil-Royle). *The chromatic number of the q-Kneser* graph  $qK_{4:2}$  is  $\chi(qK_{4:2}) = q^2 + q$ . If n > 4, then the chromatic number of the q-Kneser graph  $qK_{n:2}$  is  $\chi(qK_{n:2}) = [n-1]$ .

We see that the *q*-Kneser graph  $qK_{5:2}$  has chromatic number  $\chi(qK_{5:2}) = [4]$  by Theorem 5.4.5. The Kneser graph,  $K_{5:2}$ , also known as the Petersen graph, has chromatic number  $\chi(K_{5:2}) = 3$ , however. The relationship between the chromatic numbers of the Kneser graph and the *q*-Kneser graph is thus more complex than setting q = 1.

Using variants of Theorem 5.4.1, we can determine the chromatic number of the *q*-Kneser graphs  $qK_{n:k}$  for  $k \ge 3$ , and characterize their minimum colorings.

**Theorem 5.4.6** (Blokhuis-Brouwer-Chowdhury-Frankl-Mussche-Patkós-Szőnyi). Suppose that  $k \ge 3$ , and that either  $q \ge 3$  and  $n \ge 2k + 1$ , or q = 2 and  $n \ge 2k + 2$ . Then the chromatic number of the q-Kneser graph  $qK_{n:k}$  is  $\chi(qK_{n:k}) = [n - k + 1]$ .

Some of the remaining cases have been settled; see [15].

## 5.5 **Open Problems**

As extremal set theory questions usually have natural vector space analogs, many open problems in this area remain. We discuss three of our favorites here: the analog of Mubayi's and Verstraëte's Theorem 2.2.10, the analog of Baranyai's Theorem 3.4.33, and the question of determining the largest clique in the *q*-Kneser graph. The theorems in Chapter 4 and Chapter 5 seem to suggest that the obvious analogs of extremal set theory results should be true in vector spaces, even if their proofs don't readily generalize. The last question we discuss, that of determining the largest clique in the *q*-Kneser graph, shows that this is not always true!

### 5.5.1 Forbidding Triangles

Recall that Theorem 2.2.10 asserted that if  $\mathscr{F} \subset {X \choose k}$  contains no triangles and  $n \ge 3k/2$ , then  $|\mathscr{F}| \le {n-1 \choose k-1}$  with equality if and only if there exists an  $x \in X$  for which  $\mathscr{F} = \{F \in \mathscr{F} : x \in F\}$ . We saw that if  $3k/2 \le n < 2k$ , then a family  $\mathscr{F} \subset {X \choose k}$  without triangles is necessarily 3-wise intersecting, and so the problem is solved by Frankl's *r*-wise intersection theorem, Theorem 2.2.7, in this case.

We can naturally ask for the largest family  $\mathscr{F} \subset \begin{bmatrix} V \\ k \end{bmatrix}$  of *k*-dimensional subspaces of *V* that does not contain a triangle.

**Definition 5.5.1.** A triangle  $\mathscr{T} = \{A, B, C\} \subset \begin{bmatrix} V \\ k \end{bmatrix}$  is a family consisting of three sets A, B, C such that  $A \cap \overline{B}, A \cap \overline{C}, B \cap \overline{C} \neq \{0\}$  but  $A \cap B \cap \overline{C} = \{0\}$ .

**Question 5.5.2.** What is the maximum size of a family  $\mathscr{F} \subset \begin{bmatrix} V \\ k \end{bmatrix}$  that does not contain a triangle? What are the extremal families?

As in Section 2.2.7, this question is uninteresting unless  $n \ge 3k/2$ . Also, if  $3k/2 \le n < 2k$ , then a family  $\mathscr{F} \subset {V \brack k}$  that contains no triangle must be 3-wise intersecting. Hence, the author's and Patkós's analog of Frankl's theorem, Theorem 5.0.14, proves that, in this case,  $|\mathscr{F}| \le {n-1 \brack k-1}$  with equality if and only if  $\mathscr{F} = \{F \in {V \brack k} : v \subset F\}$  for some  $v \in {V \brack 1}$ . The proof of Theorem 2.2.10 does not generalize to vector spaces because of the issues discussed in Section 4.6.1.2.

## 5.5.2 Baranyai's Theorem

Recall that Baranyai's Theorem 3.4.33, proved that if k|n, then the complete hypergraph  $\binom{X}{k}$  can be partitioned into  $\binom{n-1}{k-1}$  perfect matchings. Is the same true for vector spaces? We saw that the vector space analogs of perfect matchings are spreads. As the number of k-dimensional spaces in  $\binom{V}{k}$  and in a spread are  $\binom{n}{k}$  and  $\lfloor n \rfloor / \lfloor k \rfloor$  respectively, the number of spreads needed to partition  $\binom{V}{k}$  is  $\binom{n-1}{k-1}$ .

**Question 5.5.3.** If k|n, can  $\begin{bmatrix} V \\ k \end{bmatrix}$  be partitioned into  $\begin{bmatrix} n-1 \\ k-1 \end{bmatrix}$  spreads?

This question is surprisingly difficult, even in the case k = 2. We remark that a partition of  $\begin{bmatrix} V \\ k \end{bmatrix}$  into *k*-dimensional subspaces is called a *parallelism* by finite geometers.

Beutelspacher [11] has constructed parallelisms when k = 2 and  $n = 2^{l}$  is a power of two. Baker [7, 8] answered Question 5.5.3 affirmatively when k = 2 and V is a vector space over  $\mathbb{F}_2$ . No constructions of parallelisms for k > 2 were known until very recently [107].

## 5.5.3 Cliques in the *q*-Kneser Graph

We saw in Section 3.3.1 that an independent set in a graph is a set of vertices S such that no two vertices in S are adjacent. In this section, we will be concerned with cliques, which are the exact opposites of independent sets.

**Definition 5.5.4.** *Given a graph* G = (V, E), a <u>clique</u> is a set of vertices C such that any two distinct vertices in C are adjacent.

In Section 3.3.2 and Section 4.4, we were interested in determining the largest independent sets of the Kneser and *q*-Kneser graphs respectively. A clique in the Kneser graph  $K_{n:k}$  corresponds to a family of *k*-subsets of *X* that are pairwise disjoint. Hence, the largest clique in the Kneser graph  $K_{n:k}$  has size  $\lfloor n/k \rfloor$ . The corresponding quantity for the *q*-Kneser graph is not as easily determined.

**Question 5.5.5.** What is the size and structure of the largest clique in the q-Kneser graph  $qK_{n:k}$ ?

A clique in the *q*-Kneser graph  $qK_{n:k}$  is a family of *k*-dimensional spaces of *V* that pairwise intersect in the zero subspace. Clearly, when k|n, a spread is the largest clique in the *q*-Kneser graph  $qK_{n:k}$  and has size [n]/[k]. Hence, for k = 2, all that remains to solve Question 5.5.5 is the case of odd *n*.

**Theorem 5.5.6** (Beutelspacher). If *n* is odd, the largest clique in the *q*-Kneser graph  $qK_{n:2}$  has size

$$\frac{q^n-q}{q^2-1}-(q-1).$$

Eisfeld and Storme [42] conjectured that if n = ck + r, where  $1 \le r \le k - 1$ , then the largest clique in the *q*-Kneser graph  $qK_{n:k}$  has size

$$\frac{q^n - q^r}{q^k - 1} - (q^r - 1). \tag{5.5.1}$$

Note that Eisfeld and Storme's conjecture (5.5.1) reduces to Theorem 5.5.6 in the case k = 2. Very recently, El-Zanati et al. [43] disproved Eisfeld's and Storme's conjecture. They showed that when q = 2, n = 3c + 2, and  $n \ge 8$ , the largest clique in the *q*-Kneser graph  $qK_{n:3}$  has size

$$\frac{2^n-2^2}{2^3-1}-2>\frac{2^n-2^2}{2^3-1}-(2^2-1),$$

which is larger than the conjectured maximum in (5.5.1).

Chapter 5, in part, is a reprint of the material as it appears in "Shadows and intersections in vector spaces," 2010. Chowdhury, Ameera; Patkós, Balázs. *J. Combin. Theory Ser. A*, 117(8):1095–1106, 2010. The dissertation author was the primary investigator and author of this paper.

# **Chapter 6**

# **On a Conjecture of Frankl and Füredi**

Recall that Fisher's Inequality, Theorem 2.3.9, states that a  $\lambda$ -intersecting family  $\mathscr{F} \subset 2^X$  of size *m* has a 1-shadow of size at least *m*. Frankl and Füredi [55] conjectured a similar inequality, Conjecture 2.3.10, for the 2-shadows of nontrivial  $\lambda$ -intersecting families. Their conjecture asserts that if  $\mathscr{F} \subset 2^X$  is a nontrivial  $\lambda$ -intersecting family of size *m*, then the number of pairs  $\{x, y\} \in {X \choose 2}$  that are contained in some  $F \in \mathscr{F}$  is at least  ${m \choose 2}$ . Conjecture 2.3.10 generalizes Fisher's Inequality and easily implies it since  ${\binom{|\partial^1 \mathscr{F}|}{2}} \ge |\partial^2 \mathscr{F}| \ge {m \choose 2}$  proves  $|\partial^1 \mathscr{F}| \ge m$ .

One important difference between Fisher's Inequality and the Frankl-Füredi conjecture is that the latter has a nontriviality restriction. Unfortunately, this condition is necessary, and sunflowers are an example of a trivial family for which the Frankl-Füredi conjecture is not valid.

**Definition 6.0.7.** A family  $\mathscr{F} \subset 2^X$  is a sunflower if deg $(x) \in \{1, |\mathscr{F}|\}$  for all  $x \in X$ .

If  $\mathscr{F} \subset {\binom{X}{k}}$  is a  $\lambda$ -intersecting sunflower of size *m*, then  $|\partial^2 \mathscr{F}| \leq m{\binom{k}{2}} < {\binom{m}{2}}$  when m > k(k-1)+1. We note that Conjecture 2.3.10 is equivalent to the seemingly stronger statement that if  $\mathscr{F} \subset 2^X$  is a  $\lambda$ -intersecting family of size *m* that is not a sunflower, then  $|\partial^2 \mathscr{F}| \geq {\binom{m}{2}}$ . Fisher's inequality and its variants are usually proved by the linear independence arguments discussed in Chapter 3. One difficulty in proving Conjecture 2.3.10 in this way is understanding how to interpret the nontriviality restriction in a linear algebra setting.
## 6.1 Results

Frankl and Füredi [55] verified Conjecture 2.3.10 when  $\lambda = 1$ . The author's work [25] appears to be the first to consider Conjecture 2.3.10 since [55] was published over twenty years ago. Several special cases of Conjecture 2.3.10 had already been proved, however, before [55] was published. For example, Ryser [99], Woodall [117], and Babai [3] showed Conjecture 2.3.10 is true when m = n. Majindar [86] proved Conjecture 2.3.10 for regular  $\lambda$ -intersecting families.

The author's main results verify the Frankl-Füredi conjecture in some special cases. We first show that their conjecture holds for nontrivial  $\lambda$ -intersecting families that satisfy a reasonable extra condition and characterize the extremal families. We then apply this result to verify the Frankl-Füredi conjecture when  $\mathscr{F}$  is additionally required to be uniform and  $\lambda$  is small. More precisely, we prove the following theorems.

**Theorem 6.1.1** (Chowdhury). Let  $\mathscr{F} \subset 2^X$  be a  $\lambda$ -intersecting family of size m. If  $\mathscr{F}$  satisfies

$$\sum_{F \in \mathscr{F}} \binom{|F|}{2} \ge \sum_{x \in X} \binom{\deg(x)}{2} = \lambda \binom{m}{2}, \tag{6.1.1}$$

then  $|\partial^2 \mathscr{F}| \ge {\binom{m}{2}}$ . Moreover, if  $\lambda \ge 2$  and  $\mathscr{F} \subset {\binom{X}{k}}$  is also k-uniform, then we have  $|\partial^2 \mathscr{F}| = {\binom{m}{2}}$  if and only if  $\mathscr{F}$  is a symmetric design.

Note that if  $\mathscr{F} \subset {X \choose k}$  is a  $\lambda$ -intersecting family of size *m*, then (6.1.1) is equivalent to

$$m \le \frac{k(k-1)}{\lambda} + 1. \tag{6.1.2}$$

**Theorem 6.1.2** (Chowdhury). Let  $\mathscr{F} \subset {\binom{X}{k}}$  be a nontrivial  $\lambda$ -intersecting family of size *m*.

- (i) If  $\lambda = 2$ , then  $|\partial^2 \mathscr{F}| \ge {m \choose 2}$  and equality holds if and only if  $\mathscr{F}$  is a symmetric design.
- (ii) If  $\lambda = 3$  and  $k \notin \{8, 11\}$ , then  $|\partial^2 \mathscr{F}| \ge {m \choose 2}$  and equality holds if and only if  $\mathscr{F}$  is a symmetric design.

#### 6.2 Old and New Conjectures

In light of (6.1.2), it is interesting to note that Stanton and Mullin [104] once conjectured that if  $\mathscr{F} \subset {X \choose k}$  is a nontrivial  $\lambda$ -intersecting family of size *m*, then (6.1.2) holds. Had this conjecture been true, Theorem 6.1.1 would have implied that Conjecture 2.3.10 is true for uniform families as well as characterized the case of equality. Unfortunately, Hall [68] proved that Stanton and Mullin's conjecture is true only for  $\lambda \in \{1,2\}$  and produced counterexamples for every  $\lambda \geq 3$ .

Since (6.1.1) and (6.1.2) are equivalent for uniform families, Hall's proof of Stanton and Mullin's conjecture for  $\lambda = 2$  shows that (6.1.1) is true for uniform, non-trivial, 2-intersecting families. Combined with Theorem 6.1.1, Hall's result proves Theorem 6.1.2 (i). If (6.1.1) were true for every nontrivial 2-intersecting family, then Theorem 6.1.1 would imply that Conjecture 2.3.10 is true for  $\lambda = 2$ . We exhibit one nontrivial 2-intersecting family that does not satisfy (6.1.1), but feel that this may be the only counterexample. Recall the unique 2-design  $\hat{\mathscr{F}}$  from Section 3.1.3, which was discovered by Ryser [99]. It is easily seen that  $\sum_{F \in \hat{\mathscr{F}}} {|F| \choose 2} = 39$  while  $2{m \choose 2} = 42$ . We conjecture that  $\hat{\mathscr{F}}$  is the only nontrivial 2-intersecting family for which (6.1.1) does not hold.

**Conjecture 6.2.1.** If  $\mathscr{F} \subset 2^X$  is a nontrivial 2-intersecting family of size *m* that is not the unique 2-design, then (6.1.1) holds.

Frankl and Füredi [55] showed (6.1.1) holds for all nontrivial 1-intersecting families, and we gave their argument in Section 2.3.4.

Theorem 6.1.1 implies that a uniform counterexample to Conjecture 2.3.10 is also a counterexample to Stanton and Mullin's conjecture. It is not difficult to see that Hall's counterexamples to Stanton and Mullin's conjecture do not give counterexamples to Conjecture 2.3.10; for definitions see [68]. Hence, we can view Conjecture 2.3.10 as a weakening of Stanton and Mullin's conjecture.

A further weakening of Stanton and Mullin's conjecture is Conjecture 6.2.2, which is due to Hall [68]. Together with Theorem 6.1.1, we see that Conjecture 6.2.2 would imply that Conjecture 2.3.10 is true if  $\mathscr{F}$  is additionally required to be *k*-uniform and *k* is sufficiently large. Deza [38] showed that  $k_1 = 2$ ; Hall [68] showed that  $k_2 = 3$ ; our proof of Theorem 6.1.2 shows that  $k_3 \leq 12$ .

**Conjecture 6.2.2** (Hall, 1977). For each  $\lambda \in \mathbb{Z}^+$ , there exists a  $k_{\lambda} \in \mathbb{Z}^+$  such that if  $k \ge k_{\lambda}$  and  $\mathscr{F} \subset {X \choose k}$  is a nontrivial  $\lambda$ -intersecting family of size m, then (6.1.2) holds.

It is natural to wonder if the obvious analog of Conjecture 2.3.10 for higher shadows holds. By considering  $\lambda$ -blowups of projective planes of order q when q is large enough, we have infinitely many nontrivial  $\lambda$ -intersecting families  $\mathscr{F}$  satisfying  $|\partial^i \mathscr{F}| < {m \choose i}$  for each  $i \ge 3$  and each  $\lambda \in \mathbb{Z}^+$ .

## 6.3 **Proof of Theorem 6.1.1**

We use the linear programming techniques discussed in Section 3.4 to prove Theorem 6.1.1. We will use Theorem 6.1.1 to prove Theorem 6.1.2 in Section 6.4. Though we used codegrees in the proof of Theorem 3.1.4, we formally define them now.

**Definition 6.3.1.** For a subset  $S \subset X$  and a family  $\mathscr{F} \subset 2^X$ , we define the <u>co-degree</u> of *S*, denoted codeg(*S*), to be the number of sets in  $\mathscr{F}$  that contain *S*,

$$\operatorname{codeg}(S) := |\{F \in \mathscr{F} : S \subset F\}|$$

**Proof of Theorem 6.1.1**. When  $\lambda = 1$ , the proof of Theorem 6.1.1 is trivial because  $|\partial^2 \mathscr{F}|$  equals the left hand side of (6.1.1). We therefore assume that  $\lambda \ge 2$ . Let  $\mathscr{F} \subset 2^X$  be a  $\lambda$ -intersecting family of size *m*. Let  $a_i$  denote the number of pairs  $\{x, y\} \in {X \choose 2}$  with codeg  $(\{x, y\}) = i$ , and observe that the following identities hold

$$\sum_{i\geq 1} ia_i = \sum_{F\in\mathscr{F}} \binom{|F|}{2}, \quad \sum_{i\geq 1} \binom{i}{2} a_i = \binom{\lambda}{2} \binom{m}{2}$$

The first counts pairs  $(\{x, y\}, F)$  where  $\{x, y\} \in {X \choose 2}$ ,  $F \in \mathscr{F}$ , and  $\{x, y\} \subset F$ . The second counts pairs  $(\{x, y\}, \{F_1, F_2\})$  where  $\{x, y\} \in {X \choose 2}$ ,  $\{F_1, F_2\} \subset \mathscr{F}$ , and  $\{x, y\} \subset F_1 \cap F_2$ . Consequently,  $(a_1, \ldots, a_m)$  is a feasible solution to the following linear program with

(6.3.1)

objective value  $|\partial^2 \mathscr{F}|$ :

Minimize

subject to:

$$\sum_{i\geq 1}^{Z_i} iz_i = \sum_{F\in\mathscr{F}} \binom{|F|}{2}$$
$$\sum_{i\geq 1} \binom{i}{2} z_i = \binom{\lambda}{2} \binom{m}{2}$$
$$z_i \geq 0, \quad i \in \{1, \dots, m\}.$$

 $\sum_{m}^{m}$ 

The dual of this linear program is:

Maximize 
$$\binom{\lambda}{2}\binom{m}{2}x + \left(\sum_{F \in \mathscr{F}}\binom{|F|}{2}\right)y$$
 (6.3.2)  
subject to:  $\binom{i}{2}x + iy \le 1, \quad i \in \{1, \dots, m\}.$ 

The feasible region of the dual linear program (6.3.2) has extreme points given by

$$\left(-\frac{1}{\binom{j+1}{2}},\frac{2}{j+1}\right), \ j \in \{1,\dots,m-1\}.$$
 (6.3.3)

If  $\mathscr{F}$  satisfies (6.1.1), then setting  $j = \lambda - 1$  in (6.3.3) and applying weak duality, Theorem 3.4.6, yields

$$|\partial^2 \mathscr{F}| \ge {\binom{\lambda}{2}} {\binom{m}{2}} \left(-\frac{1}{\binom{\lambda}{2}}\right) + \left(\sum_{F \in \mathscr{F}} {\binom{|F|}{2}}\right) \left(\frac{2}{\lambda}\right) \ge {\binom{m}{2}}, \tag{6.3.4}$$

as desired. Finally, note that equality in (6.1.1) follows from counting pairs  $(x, \{F_1, F_2\})$  such that  $\{F_1, F_2\} \subset \mathscr{F}$  and  $x \in F_1 \cap F_2$ .

We now assume that  $\lambda \ge 2$  and that  $\mathscr{F} \subset {X \choose k}$  is also *k*-uniform. We prove that  $|\partial^2 \mathscr{F}| = {m \choose 2}$  if and only if  $\mathscr{F}$  is a symmetric design. Ryser [99], Woodall [117], and Babai [3] showed that if  $\mathscr{F} \subset 2^X$  is a  $\lambda$ -intersecting family of size  $m = |\partial^1 \mathscr{F}| = n$ , then  $|\partial^2 \mathscr{F}| = {m \choose 2}$ . Conversely, suppose  $|\partial^2 \mathscr{F}| = {m \choose 2}$  and let  $a_i$  denote the number of pairs  $\{x, y\} \in {X \choose 2}$  with codeg  $(\{x, y\}) = i$ . We will show that  $\mathscr{F}$  is *k*-regular, which immediately implies that  $\mathscr{F}$  is a symmetric design. By (6.3.4), we see that equality holds in (6.1.2),  $(a_1, \ldots, a_m)$  is an optimal solution to the primal linear program (6.3.1),

and (6.3.3) with  $j = \lambda - 1$  is an optimal solution to the dual linear program (6.3.2). By weak complementary slackness, Theorem 3.4.8, this implies

$$a_i = 0$$
 or  $\binom{i}{2} \left(\frac{-1}{\binom{\lambda}{2}}\right) + i\left(\frac{2}{\lambda}\right) = 1.$ 

Hence  $a_i = 0$  for  $i \notin \{\lambda - 1, \lambda\}$  for  $i \in \{1, \dots, m\}$ . The constraints in the primal linear program (6.3.1) imply  $a_{\lambda-1} + a_{\lambda} = {m \choose 2}$  and  ${\binom{\lambda-1}{2}}a_{\lambda-1} + {\binom{\lambda}{2}}a_{\lambda} = {\binom{\lambda}{2}}{\binom{m}{2}}$ , so  $a_{\lambda-1} = 0$  and  $a_{\lambda} = {\binom{m}{2}}$ .

Let  $x \in X$  and count pairs (y, F) such that  $\{x, y\} \subset F$ . Since  $a_{\lambda} = {m \choose 2}$ , we have

$$\lambda |\partial^1 L(x)| = (k-1) \deg(x) \text{ so } |\partial^1 L(x)| = \frac{k-1}{\lambda} \deg(x).$$
(6.3.5)

We will give a lower bound on  $|\partial^1 L(x)|$  in terms of deg(x) that will allow us to prove that  $\mathscr{F}$  is k-regular. For  $x \in X$ , let  $b_{x,i}$  denote the number of vertices  $y \in X$  such that  $\operatorname{codeg}(\{x,y\}) = i$  and observe that the following identities hold

$$\sum_{i\geq 1} ib_{x,i} = (k-1)\deg(x), \quad \sum_{i\geq 1} \binom{i}{2}b_{x,i} = (\lambda-1)\binom{\deg(x)}{2}.$$

The first follows from counting pairs (y, F) where  $\{x, y\} \in {X \choose 2}$ ,  $F \in \mathscr{F}$ , and  $\{x, y\} \subset F$ . The second follows from counting pairs  $(y, \{F_1, F_2\})$  where  $\{x, y\} \in {X \choose 2}$ ,  $\{F_1, F_2\} \subset \mathscr{F}$ , and  $\{x, y\} \subset F_1 \cap F_2$ . Consequently,  $(b_{x,1}, \ldots, b_{x,m})$  is a feasible solution to the following linear program with objective value  $|\partial^1 L(x)|$ :



The dual of this linear program is:

Maximize 
$$(\lambda - 1) {deg(x) \choose 2} y + (k - 1) deg(x) z$$
  
subject to:  ${i \choose 2} y + iz \le 1, \quad i \in \{1, \dots, m\}.$ 

Since (6.3.3) with  $j = \lambda - 1$  is a feasible solution, using (6.3.5) yields

$$\frac{k-1}{\lambda}\deg(x) = |\partial^1 L(x)| \ge \deg(x)\left(\frac{2(k-1)}{\lambda} - (\lambda-1)\frac{\deg(x)-1}{2}\binom{\lambda}{2}^{-1}\right).$$

Hence, deg $(x) \ge k$  for each  $x \in X$ . On the other hand, let  $F \in \mathscr{F}$  and count pairs (x, F') such that  $F \ne F' \in \mathscr{F}$  and  $x \in F \cap F'$ . Since equality holds in (6.1.2), we have

$$k^2 \le \sum_{x \in F} \deg(x) = \lambda (m-1) + k = k^2$$

so deg(x) = k for all  $x \in X$ . Hence  $\mathscr{F}$  is k-regular and is thus a symmetric design.  $\Box$ 

## 6.4 Proof of Theorem 6.1.2

In light of (6.1.1) and (6.1.2), we are interested in upper bounds on the sizes of nontrivial  $\lambda$ -intersecting families  $\mathscr{F}$  that depend only on the sizes of the sets in  $\mathscr{F}$ . One of the first results of this kind is Deza's theorem [38], which bounds the size of  $\lambda$ -intersecting families that are not sunflowers. In the case when  $\mathscr{F} \subset {\binom{X}{k}}$  is *k*-uniform, the upper bound on *m* in (6.4.1) is bigger than the upper bound on *m* in (6.1.2) by a factor of roughly  $\lambda$ .

**Theorem 6.4.1** (Deza, 1974). Let  $\mathscr{F} \subset 2^X$  be a  $\lambda$ -intersecting family of size m that is not a sunflower. Define  $K := \max_{F \in \mathscr{F}} |F|$ . Then

$$m \le \max\{\lambda(\lambda+1) + 1, (K-\lambda)((K-\lambda) + 1) + 1\}.$$
(6.4.1)

Since nontriviality is a stronger restriction on  $\mathscr{F}$  than not being a sunflower, it is plausible that (6.4.1) could be improved for nontrivial  $\mathscr{F}$ . Frankl and Füredi [55] did exactly this when they showed that (6.1.1) holds for all nontrivial 1-intersecting families. We mentioned in the introduction that Stanton and Mullin [104] conjectured that (6.4.1) could be improved to (6.1.2) if  $\mathscr{F}$  is nontrivial and *k*-uniform; Theorem 6.4.1 verifies Stanton and Mullin's conjecture for  $\lambda = 1$  and Hall proved Stanton and Mullin's conjecture when  $\lambda = 2$ . **Theorem 6.4.2** (Hall, 1977). If  $\mathscr{F} \subset {\binom{X}{k}}$  is a nontrivial 2-intersecting family of size *m*, *then* 

$$m \leq \binom{k}{2} + 1.$$

We adapt Hall's proof of Theorem 6.4.2 to prove Theorem 6.1.2. (For the reader's convenience, we first reproduce Hall's proof of Theorem 6.4.2.) In our proof, we will use the fact that if  $\mathscr{F} \subset {X \choose k}$  is a  $\lambda$ -intersecting family, then deg(x) does not lie in a certain interval. Deza [37] showed that if  $\mathscr{F} \subset {X \choose k}$  is a  $\lambda$ -intersecting family of size *m* then, for all  $x \in X$ ,

$$\deg(x)(m+1-\deg(x)) \le \max\{\lambda, k-\lambda\}(m+1). \tag{6.4.2}$$

McCarthy and Vanstone [92] adapted an argument of Connor [29], and improved this bound; they gave the following restriction on deg(x).

**Theorem 6.4.3** (McCarthy-Vanstone, 1979). Let  $\mathscr{F} \subset {\binom{X}{k}}$  be a  $\lambda$ -intersecting family of size *m*.

(*i*) If  $x \in X$  then,

$$\deg(x)((k-\lambda)+\lambda(m-\deg(x))) \le (k-\lambda)((k-\lambda)+\lambda m).$$
(6.4.3)

(*ii*) Let  $\{x, y\} \subset {X \choose 2}$  and define

(a) 
$$a_{11} = (k - \lambda)((k - \lambda) + \lambda m) - \deg(x)((k - \lambda) + \lambda(m - \deg(x))),$$
  
(b)  $a_{12} = a_{21} = \lambda \deg(x) \deg(y) - ((k - \lambda) + \lambda m) \operatorname{codeg}(\{x, y\}),$   
(c)  $a_{22} = (k - \lambda)((k - \lambda) + \lambda m) - \deg(y)((k - \lambda) + \lambda(m - \deg(y))).$ 

The following determinant is non-negative:

$$\det \left(\begin{array}{cc} a_{11} & a_{12} \\ a_{21} & a_{22} \end{array}\right) \ge 0. \tag{6.4.4}$$

We now reproduce Hall's proof of Theorem 6.4.2. Note that Hall had originally used (6.4.2) in his proof, but we will use (6.4.3) instead since it makes the argument cleaner.

Hall's Proof of Theorem 6.4.2 Suppose, for a contradiction, that there exists a nontrivial 2-intersecting  $\mathscr{F} \subset {X \choose k}$  of size  $m > {k \choose 2} + 1$ . Write

$$m = \binom{k}{2} + 1 + \varepsilon, \ \varepsilon \in \mathbb{Z}^+.$$
(6.4.5)

Note that the left hand side of (6.4.3) is quadratic in deg(x) with roots deg(x) = 0 and deg(x) = m - 1 + k/2. If there exists an  $x \in X$  with  $k \le deg(x) \le (m - 1 + k/2) - k$ , then (6.4.3) is true for deg(x) = k; together with (6.4.5), this implies that  $\varepsilon \le 0$ , which is impossible. Hence, for all  $x \in X$ , either

$$\deg(x) \le k - 1 \text{ or } \deg(x) \ge m - \left\lceil \frac{k}{2} \right\rceil.$$
(6.4.6)

We say a vertex  $x \in X$  with deg $(x) \le k - 1$  is *light* and is heavy if deg $(x) \ge m - \lceil k/2 \rceil$ .

By (6.4.5), for any  $F \in \mathscr{F}$ , we have

$$\sum_{x \in F} \deg(x) = 2(m-1) + k = k^2 + 2\varepsilon.$$
(6.4.7)

Since the average degree of a vertex in  $F \in \mathscr{F}$  is greater than k, every set  $F \in \mathscr{F}$  contains a heavy vertex. As  $\mathscr{F}$  is nontrivial, there are at least two heavy vertices  $x_1, x_2$ . Define

$$s := |\{F \in \mathscr{F} : \{x_1, x_2\} \subset \mathscr{F}\}|, \qquad t := |\{F \in \mathscr{F} : x_1 \in F, x_2 \notin F\}|$$
$$u := |\{F \in \mathscr{F} : x_1 \notin F, x_2 \in F\}|, \qquad v := |\{F \in \mathscr{F} : x_1, x_2 \notin F\}|$$

We have  $s \le k-1$  because  $\lambda = 2$  and  $\mathscr{F}$  is nontrivial. Since u + v and t + v count the number of sets  $F \in \mathscr{F}$  not on  $x_1, x_2$  respectively, (6.4.6) yields  $t + v, u + v \le \lceil k/2 \rceil$ . Consequently (6.4.5) implies,

$$\binom{k}{2}+1+\varepsilon=m=s+t+u+v\leq s+(t+v)+(u+v)\leq (k-1)+2\left\lceil\frac{k}{2}\right\rceil\leq 2k.$$

As  $\varepsilon \in \mathbb{Z}^+$ , we have a contradiction for  $k \ge 5$ . For k = 4, Theorem 6.4.1 yields  $m \le 7$ , so we have a contradiction in this case too. We have shown that if  $\mathscr{F} \subset {X \choose k}$  is a nontrivial 2-intersecting family of size *m* then  $m \le {k \choose 2} + 1$ .

For larger  $\lambda$ , if we knew that a nontrivial  $\lambda$ -intersecting  $\mathscr{F} \subset {X \choose k}$  that does not satisfy (6.1.2) has at least  $\lambda$  heavy vertices, then Hall's argument would yield a proof of Conjecture 6.2.2. Unfortunately, Hall's averaging argument only shows that

any nontrivial  $\lambda$ -intersecting  $\mathscr{F} \subset {X \choose k}$  that does not satisfy (6.1.2) has at least two heavy vertices. In the proof of Theorem 6.1.2, we expend a lot of effort to eliminate the possibility that there are exactly two heavy vertices when  $\lambda = 3$ ; the key difficulty is getting a good bound on the number of sets  $F \in \mathscr{F}$  that contain both the heavy vertices.

**Proof of Theorem 6.1.2**. We observe that Theorem 6.4.2 together with Theorem 6.1.1 yields Theorem 6.1.2 (i).

For the rest of the proof, we assume that  $\lambda = 3$ . We will show that if  $\mathscr{F} \subset {\binom{X}{k}}$  is a nontrivial 3-intersecting family, where  $k \notin \{8, 11\}$ , then (6.1.2) holds. Theorem 6.1.1 then implies that  $|\partial^2 \mathscr{F}| \ge {\binom{m}{2}}$  and that equality holds if and only if  $\mathscr{F}$  is a symmetric design. First suppose k < 6. It is not difficult to see that if  $\mathscr{F}$  is a nontrivial *k*-uniform 3-intersecting family of size *m*, where  $k \in \{4, 5\}$ , then  $m \le 5$ ; for proofs of these results in a more general setting see [62], [63], and [111]. Hence, (6.1.2) holds when k < 6.

Suppose, for a contradiction, that  $k \ge 12$  and that  $\mathscr{F} \subset {X \choose k}$  is a nontrivial 3-intersecting family of size *m* for which (6.1.2) does not hold. Write

$$m = \frac{k(k-1)}{3} + 1 + \varepsilon, \ \varepsilon > 0.$$
 (6.4.8)

Note that the left hand side of (6.4.3) is quadratic in deg(x) with roots deg(x) = 0 and deg(x) = m - 1 + k/3. If there exists an  $x \in X$  with  $k \le deg(x) \le (m - 1 + k/3) - k$ , then (6.4.3) is true for deg(x) = k; together with (6.4.8), this implies that  $\varepsilon \le 0$ , which is impossible. Hence, for all  $x \in X$ , either

$$\deg(x) \le k-1 \text{ or } \deg(x) \ge m - \left\lceil \frac{2k}{3} \right\rceil \ge m - \frac{2k+2}{3}.$$
 (6.4.9)

Following Hall [68], we say a vertex  $x \in X$  is *light* if deg $(x) \le k - 1$  and is heavy if deg $(x) \ge m - \lfloor 2k/3 \rfloor$ .

By (6.4.8), for any  $F \in \mathscr{F}$ , we have

$$\sum_{x \in F} \deg(x) = 3(m-1) + k = k^2 + 3\varepsilon.$$
(6.4.10)

Since the average degree of a vertex in  $F \in \mathscr{F}$  is greater than k, every set  $F \in \mathscr{F}$  contains a heavy vertex. As  $\mathscr{F}$  is nontrivial, there are at least two heavy vertices. We consider two cases, according to whether there are exactly two or greater than two heavy vertices. *Case 1: There are exactly two heavy vertices.* Let  $x_1, x_2$  be the heavy vertices. Since  $\mathscr{F}$  is nontrivial, there exists a set  $F_1 \in \mathscr{F}$  which contains  $x_1$  but not  $x_2$ , and there exists a set  $F_2 \in \mathscr{F}$  which contains  $x_2$  but not  $x_1$ . Let  $F_1 \cap F_2 := \{y_1, y_2, y_3\}$ . Define

$$s := |\{F \in \mathscr{F} : \{x_1, x_2\} \subset \mathscr{F}\}|, \qquad t := |\{F \in \mathscr{F} : x_1 \in F, x_2 \notin F\}|,$$
$$u := |\{F \in \mathscr{F} : x_1 \notin F, x_2 \in F\}|,$$

and observe that m = s + t + u since every  $F \in \mathscr{F}$  contains a heavy vertex. By (6.4.9), we have  $t, u \leq \lfloor 2k/3 \rfloor \leq (2k+2)/3$ .

We now show how to obtain an upper bound on *s* in terms of *k*. Observe that any  $F \in \mathscr{F}$  that contains  $\{x_1, x_2\}$  intersects  $F_1 \setminus \{x_1\}$  in a subset of size two. Consequently,

$$2s = \sum_{\{x_1, x_2\} \subset F} |F \cap F_1 \setminus \{x_1\}| = \sum_{w \in F_1 \setminus \{x_1\}} \operatorname{codeg}(x_1, x_2, w).$$
(6.4.11)

We claim that if  $w \in X \setminus \{x_1, x_2\}$  and there exists an  $F \in \mathscr{F}$  such that

$$\{x_1, x_2\} \not\subset F, \ w \notin F, \tag{6.4.12}$$

then  $\operatorname{codeg}(\{x_1, x_2, w\}) \leq (k-1)/2$ . Suppose  $F', \hat{F} \in \mathscr{F}$  are distinct sets in  $\mathscr{F}$  that both contain  $\{x_1, x_2, w\}$ . Since  $\lambda = 3$ , the intersections of F' and  $\hat{F}$  with  $F \setminus \{x_1, x_2\}$  must be disjoint subsets of size two. Hence,  $\operatorname{codeg}(\{x_1, x_2, w\}) \leq (k-1)/2$ . Observe that if  $w \in F_1 \setminus \{x_1, y_1, y_2, y_3\}$ , then  $F_2$  is a set in  $\mathscr{F}$  that satisfies (6.4.12). We will consider two subcases according to whether for each  $y_i \in F_1 \cap F_2$ , there exists an  $F \in \mathscr{F}$  that satisfies (6.4.12) for  $w = y_i$ .

Subcase 1: For each  $y_i \in F_1 \cap F_2$ , there exists an  $F \in \mathscr{F}$  that satisfies (6.4.12) for  $w = y_i$ .

Applying (6.4.9) and (6.4.11) yields

$$\frac{k(k-1)}{3} + 1 + \varepsilon = m = s + t + u \le \frac{(k-1)^2}{4} + 2\left\lceil \frac{2k}{3} \right\rceil \le \frac{3k^2 + 10k + 19}{12}.$$
 (6.4.13)

This implies that  $k^2 - 14k - 7 + 12\varepsilon \le 0$ , which is a contradiction for  $k \ge 15$  since  $\varepsilon \ge 1/3$ . For the remaining values of *k*, we refer the reader to the appendix.

Subcase 2: There exists a  $y_i \in F_1 \cap F_2$  for which no  $F \in \mathscr{F}$  satisfies (6.4.12) for  $w = y_i$ .

Note that if  $y_i$  is in every  $F \in \mathscr{F}$  that does not contain  $\{x_1, x_2\}$  then, by (6.4.9), codeg  $(\{x_1, x_2, y_i\}) \le k - 1 - (t + u)$ . Suppose that every  $F \in \mathscr{F}$ , not containing  $\{x_1, x_2\}$ , contains *j* of the elements  $\{y_1, y_2, y_3\}$  where  $j \in \{1, 2, 3\}$ . Applying (6.4.11) yields

$$\frac{k(k-1)}{3} + 1 + \varepsilon = m = s + t + u$$

$$\leq \frac{1}{2} \left( \sum_{w \in F_1 \setminus \{x_1\}} \operatorname{codeg}(x_1, x_2, w) \right) + t + u \\
\leq \frac{1}{2} \left( j(k-1-(t+u)) + \frac{(k-1-j)(k-1)}{2} \right) + t + u \\
= \frac{j}{2}(k-1) + \frac{(k-1-j)(k-1)}{4} + (2-j)\frac{t+u}{2} \\
\leq \frac{j}{2}(k-1) + \frac{(k-1-j)(k-1)}{4} + (2-j)\frac{2k+2}{3} \\
= -\left(\frac{k}{6} + \frac{7}{6}\right)j + \frac{(k-1-j)(k-1)}{4} + \frac{4k+4}{3} \\
\leq \frac{3k^2 + 5k + 8}{12},$$
(6.4.14)

since the penultimate expression in (6.4.14) is maximized when j = 1. This implies that  $(k-1)(k-8) + 12(\varepsilon - 1/3) \le 0$ , which is a contradiction for  $k \ge 9$ .

If k = 8 then  $\varepsilon = 1/3$ . Observe that  $\operatorname{codeg}(x_1, x_2, w) \le 3$  if w is not one of the j special vertices in  $\{y_1, y_2, y_3\}$ ; in the bound for s in (6.4.14), we use the weaker bound  $\operatorname{codeg}(x_1, x_2, w) \le 7/2$  for vertices w that are not one of the j special vertices in  $\{y_1, y_2, y_3\}$ . If we replace the weaker bound on  $\operatorname{codeg}(x_1, x_2, w)$  by the tighter bound, then we get a contradiction for k = 8 as well. Finally, if  $k \in \{6,7\}$ , then  $\varepsilon \in \mathbb{Z}^+$  so we also get a contradiction in this case.

*Case 2: There are greater than two heavy vertices.* Let  $x_1, x_2, x_3$  be three heavy vertices. Define

$$\begin{split} s &:= |\{F \in \mathscr{F} : \{x_1, x_2, x_3\} \subset \mathscr{F}\}|, \qquad t := |\{F \in \mathscr{F} : x_1 \in F, x_2, x_3 \notin F\}|, \\ u &:= |\{F \in \mathscr{F} : x_2 \in F, x_1, x_3 \notin F\}|, \qquad v := |\{F \in \mathscr{F} : x_3 \in F, x_1, x_2 \notin F\}|, \\ w &:= |\{F \in \mathscr{F} : x_1, x_2 \in F, x_3 \notin F\}|, \qquad x := |\{F \in \mathscr{F} : x_1, x_3 \in F, x_2 \notin F\}|, \\ y &:= |\{F \in \mathscr{F} : x_2, x_3 \in F, x_1 \notin F\}|, \qquad z := |\{F \in \mathscr{F} : x_1, x_2, x_3 \notin F\}|. \end{split}$$

By counting the number of sets not containing  $x_1, x_2$ , or  $x_3$  respectively we have

$$u+v+y+z, t+v+x+z, t+u+w+z \le \left\lceil \frac{2k}{3} \right\rceil \le \frac{2k+2}{3},$$
 (6.4.15)

by (6.4.9). As  $\lambda = 3$  and  $\mathscr{F}$  is nontrivial, we have  $s \le k - 2$ . Therefore (6.4.15) implies,

$$\frac{k(k-1)}{3} + 1 + \varepsilon = m = s + t + u + v + w + x + y + z$$
  

$$\leq s + (u + v + y + z) + (t + v + x + z) + (t + u + w + z)$$
  

$$\leq (k-2) + 3 \left\lceil \frac{2k}{3} \right\rceil \leq (k-2) + (2k+2) = 3k.$$
(6.4.16)

This implies  $k^2 - 10k + 3 + 3\varepsilon \le 0$ , so we have a contradiction for  $k \ge 10$  since  $\varepsilon \ge 1/3$ . For the remaining values of *k*, we refer the reader to the appendix.

We have shown that if  $\mathscr{F} \subset {X \choose k}$  is a nontrivial 3-intersecting family of size *m* and  $k \notin \{8, 11\}$ , then  $\mathscr{F}$  satisfies (6.1.2). By Theorem 6.1.1, this implies that if  $\mathscr{F}$  satisfies the hypotheses of Theorem 6.1.2 (ii), then  $|\partial^2 \mathscr{F}| \ge {m \choose 2}$  and equality holds if and only if  $\mathscr{F}$  is a symmetric design.

## 6.5 Appendix

Here, we collect some computations that are needed to verify Theorem 6.1.2 for small values of k. We regret that we could not prove Theorem 6.1.2 for all values of k. The missing cases are k = 11, m = 40 in Case 1, Subcase 1 and k = 8, m = 20 in Case 2.

Case Analysis for Case 1, Subcase 1:

If k = 14, then codeg  $(x_1, x_2, w) \le 6$  for  $w \in F_1 \setminus \{x_1\}$  so (6.4.11) yields  $s \le 39$ . Using this value for *s* in (6.4.13) yields  $\varepsilon < 0$ , which contradicts (6.4.8).

If k = 13, then the penultimate inequality in (6.4.13) yields that m = 54, s = 36, and t = u = 9. Using these values in (6.4.4) yields a contradiction.

If k = 12, then codeg  $(x_1, x_2, w) \le 5$  for  $w \in F_1 \setminus \{x_1\}$  so (6.4.11) yields  $s \le 27$ . Using this value for *s* in (6.4.13) yields  $\varepsilon < 0$ , which contradicts (6.4.8). If k = 11, then (6.4.13) yields  $m \in \{38, 39, 40, 41\}$ . If we add the constraint  $z_s = 1$  to (6.3.1), then the dual linear program becomes

Maximize 
$$3\binom{m}{2}w + \binom{k}{2}mx + y \qquad (6.5.1)$$
  
subject to: 
$$\binom{i}{2}w + ix \le 1, \quad i \in \{1, \dots, m\} \setminus \{s\}$$
  
$$\binom{s}{2}w + sx + y \le 1.$$

and any feasible solution to (6.5.1) is a lower bound on  $|\partial^2 \mathscr{F}|$ . If m = 38, then (6.4.13) implies  $s \ge 22$  because  $t, u \le 8$ . Observe that  $\binom{p}{2} \left(-\frac{1}{3}\right) + p\left(\frac{2}{3}\right) \le \binom{q}{2} \left(-\frac{1}{3}\right) + q\left(\frac{2}{3}\right)$  if  $p > q \ge 2$ . We have  $\binom{22}{2} \left(-\frac{1}{3}\right) + 22 \left(\frac{2}{3}\right) + 63\frac{1}{3} = 1$ , so the previous inequality implies that  $(-1/3, 2/3, 63\frac{1}{3})$  is always a feasible solution to (6.5.1) for m = 38 and k = 11. Hence,  $|\partial^2 \mathscr{F}| > \binom{m}{2}$ , which contradicts our initial assumption. A similar argument eliminates the case m = 39. If m = 41, then s = 25 and t = u = 8 so deg $(x_1) = 33$ ; this contradicts Theorem 6.4.3.

If k = 10, then the penultimate inequality in (6.4.13) yields m = 32 and s = 18. Since  $\binom{18}{2} \binom{-1}{3} + 18 \binom{2}{3} + 40 = 1$ , we have (-1/3, 2/3, 40) is a feasible solution to (6.5.1) for m = 32 and k = 10. Consequently,  $|\partial^2 \mathscr{F}| > \binom{m}{2}$ , which contradicts our initial assumption.

If k = 9, then (6.4.13) yields  $\varepsilon \in \{1, 2, 3\}$  and  $s \le 16$ . We consider the cases  $\varepsilon = 1$  and  $\varepsilon \in \{2, 3\}$  separately.

<u> $\varepsilon = 1$ </u>: By (6.4.9), we have  $s \ge 14$  and  $t, u \le 6$ . Note that  $\binom{15}{2} (-\frac{1}{3}) + 15(\frac{2}{3}) + 26 = 1$  so  $(-\frac{1}{3}, \frac{2}{3}, 26)$  is a feasible solution to (6.5.1) for m = 26, k = 9, and  $s \ge 15$ . If s = 14, then t = u = 6 so deg $(x_1) = deg(x_2) = 20$ . Observe that (6.4.9) and (6.4.10) imply

$$84 = \sum_{x \in F_1} \deg(x) = \deg(x_1) + \sum_{x_1 \neq x \in F_1} \deg(x) \le 20 + 8(8) = 84;$$

hence, if w is a light vertex in a set in  $\mathscr{F}$  that doesn't contain both  $x_1, x_2$ , then deg(w) = 8. Now suppose that z is a light vertex that is only contained in sets that contain both  $x_1, x_2$ ; that is deg $(z) = \operatorname{codeg}(x_1, x_2, z)$ . Since  $F_1$  satisfies (6.4.12) for w = z, we see deg $(z) = \operatorname{codeg}(x_1, x_2, z) \le 4$ . Now let  $F' \in \mathscr{F}$  be a set that contains both  $x_1, x_2$  and observe that

$$84 = \sum_{x \in F'} \deg(x) = \deg(x_1) + \deg(x_2) + \sum_{z \in F' \setminus \{x_1, x_2\}} \deg(z) = 20 + 20 + \sum_{z \in F' \setminus \{x_1, x_2\}} \deg(z).$$

For  $z \in F' \setminus \{x_1, x_2\}$ , we have  $\deg(z) \le 4$  or  $\deg(z) = 8$  so either F' contains four vertices of degree eight and three vertices of degree four or five vertices of degree eight and two vertices whose degrees sum to four. Hence for  $x \in X$ , we have  $\deg(x) \in \{1, 2, 3, 4, 8, 20\}$ . Let  $n_i$  denote the number of vertices of degree *i*. Also define

$$m_1 := |\{F \in \mathscr{F} : \{x_1, x_2\} \subset \mathscr{F}, \exists w, z \in F \text{ with } \deg(w) = 1, \deg(z) = 3\}|,$$
  

$$m_2 := |\{F \in \mathscr{F} : \{x_1, x_2\} \subset \mathscr{F}, \exists w, z \in F \text{ with } \deg(w) = \deg(z) = 2\}|,$$
  

$$m_3 := |\{F \in \mathscr{F} : \{x_1, x_2\} \subset \mathscr{F}, \exists w \in F \text{ with } \deg(w) = 4\}|,$$

and observe that  $m_1 + m_2 + m_3 = s = 14$ . Note that  $n_{20} = 2$ ,  $3n_3 = n_1 = m_1$ ,  $n_2 = m_2$ , and  $4n_4 = 3m_3$ . In particular,  $m_3$  is even so  $n_1 + n_2 = m_1 + m_2$  is also even. Observe that

$$234 = 9 \cdot 26 = km = \sum_{x \in X} \deg(x) = 20n_{20} + 8n_8 + 4n_4 + 3n_3 + 2n_2 + n_1$$
$$= 20n_{20} + 8n_8 + 4n_4 + n_1 + 2n_2 + n_1 = 20n_{20} + 8n_8 + 4n_4 + 2(n_1 + n_2). \quad (6.5.2)$$

Since  $n_1 + n_2$  is even, (6.5.2) implies that 4|234, which is a contradiction.

 $\varepsilon \in \{2,3\}$ : Without loss of generality, we will assume deg $(x_1) \le deg(x_2)$  or equivalently that  $u \ge t$ . Observe that (6.4.9) and (6.4.10) imply

$$\deg(x_1) = \sum_{x \in F_1} \deg(x) - \sum_{z \in F_1 \setminus \{x_1\}} \deg(z) \ge \sum_{x \in F_1} \deg(x) - (k-1)^2$$
$$= k^2 + 3\varepsilon - (k-1)^2 = 2k - 1 + 3\varepsilon.$$
(6.5.3)

If  $\varepsilon \in \{2,3\}$ , then  $s \le 16$  implies u = 6. When  $\varepsilon = 2$ , we have deg $(x_1) = 21$ , which contradicts (6.5.3). When  $\varepsilon = 3$ , we have deg $(x_1) = 22$ , which again contradicts (6.5.3).

The case  $k \in \{6,7,8\}$  can be eliminated with an argument similar to the one for  $k = 9, \varepsilon \in \{2,3\}$ ; we omit the details.

Case Analysis for Case 2:

If k = 9, then we arrive at a contradiction by using the third to last expression in (6.4.16).

If k = 8, then  $m \in \{20, 21, 22, 23, 24\}$  by (6.4.16). For  $m \in \{21, 22, 23, 24, 25\}$ , any heavy vertex x satisfies deg $(x) \ge m - 5$  by (6.4.3). Hence, the upper bound in (6.4.15) is improved and implies that deg $(x_1) = s + t + w + x \le 16$ . This gives a contradiction for  $m \in \{22, 23, 24\}$ . For m = 21, we have s = 6, w = x = y = 5, and t = u = v = z = 0. Hence codeg $(\{x_i, x_j\}) = 11$  for  $\{i, j\} \in \{1, 2, 3\}$ . Adding the constraint  $z_{11} \ge 3$  to (6.3.1) yields that  $|\partial^2 \mathscr{F}| \ge {m \choose 2}$ , which contradicts our initial assumption; we omit the details since the computation is similar to that in (6.5.1).

If k = 7, then  $m \in \{16, 17, 18, 19, 20\}$  by (6.4.16) and (6.4.3) shows that any heavy vertex x satisfies deg $(x) \ge m - 4$ . Hence, the upper bound in (6.4.15) is improved and implies that deg $(x_1) = s + t + w + x \le 13$ . This gives a contradiction for  $m \in \{18, 19, 20\}$ .

If k = 7 and m = 17, then we conclude that s = 5, w = x = y = 4, and that t = u = v = z = 0. Note that if  $x \in X$  is heavy then deg $(x) \ge 13$  by (6.4.9). If there is a fourth heavy vertex  $x_4$ , it can be in at most one of the five sets on  $x_1, x_2, x_3$ ; moreover since heavy vertices have degree at least thirteen,  $x_4$  is in each of the four sets on  $\{x_1, x_2\}$ ,  $\{x_1, x_3\}$ , and  $\{x_2, x_3\}$ . As  $\lambda = 3$ , this argument shows that there are at most four heavy vertices. If there are exactly four heavy vertices  $x_1, x_2, x_3, x_4$ , then  $codeg(\{x_i, x_i\}) = 9$ for  $\{i, j\} \subset \{1, 2, 3, 4\}$ . Adding the constraint  $z_9 \ge 6$  to (6.3.1) yields that  $|\partial^2 \mathscr{F}| \ge {m \choose 2}$ , which contradicts our initial assumption; we omit the details since the computation is similar to the one in (6.5.1). Consequently, there are exactly three heavy vertices and  $deg(x_1) = deg(x_2) = deg(x_3) = 13$ . Moreover, any set in  $\mathscr{F}$  contains either exactly two or exactly three heavy vertices. Let F' be a set that contains exactly two heavy vertices. Equations (6.4.9) and (6.4.10) yield that F' contains four vertices of degree six and one of degree five. Hence, if w is a light vertex and w is contained in a set of  $\mathscr{F}$  with two heavy vertices, then deg(w)  $\in$  {5,6}; otherwise deg(w) = codeg(x<sub>1</sub>, x<sub>2</sub>, x<sub>3</sub>, w) = 1. Now let  $\hat{F}$  be a set in  $\mathscr{F}$  that contains exactly three heavy vertices. Since deg $(w) \in \{1, 5, 6\}$ for  $w \in X \setminus \{x_1, x_2, x_3\}$ , (6.4.10) yields that  $\hat{F}$  contains three vertices of degree five and one vertex of degree one. As s = 5, there are fifteen vertices of degree five and five of degree one. Let  $n_i$  denote the number of vertices of degree *i*. We have

$$119 = km = \sum_{x \in X} \deg(x) = 13n_{13} + 6n_6 + 5n_5 + n_1 = 39 + 6n_6 + 75 + 5,$$

which implies that  $n_6 = 0$ , a contradiction.

If k = 7 and m = 16 then (6.4.16) implies that  $s \in \{4, 5\}$  and  $t + u + v + 2z \le 1$ . We conclude that z = 0 and at most one of t, u, v equals one. We first show that the situation where exactly one of t, u, v is one is impossible. Without loss of generality, assume for a contradiction that t = 1 and u = v = 0. If s = 4, then (6.4.15) implies that

$$16 = m = s + t + u + v + w + x + y + z \le 4 + 1 + 0 + 0 + 0 + 3 + 3 + 4 = 15,$$

which is a contradiction. If s = 5, then we can conclude via a similar argument that w = x = 3 and y = 4. Consider the unique  $\hat{F} \in \mathscr{F}$  with  $x_1 \in F$  and  $x_2, x_3 \notin F$ . Since light vertices have degree at most six,  $\hat{F}$  must contain another heavy vertex  $x_4$  by (6.4.10). Now deg $(x_4) \ge 12$  and since  $x_4$  can only be in one of the five sets on  $x_1, x_2, x_3$ , we have that  $x_4$  is in each of the remaining 11 sets. As a result, for  $i, j \in \{1, 2, 3, 4\}$ , we have codeg  $(\{x_i, x_j\}) = 8$  if  $\{i, j\} \ne \{1, 3\}$  and we have codeg  $(\{x_1, x_3\}) = 9$ . Adding the constraints  $z_9 \ge 1$  and  $z_8 \ge 5$  to (6.3.1) yields that  $|\partial^2 \mathscr{F}| \ge \binom{m}{2}$ , which contradicts our initial assumption; we omit the details since the computation is similar to the one in (6.5.1). Hence, we can assume t = u = v = 0. If s = 5, then two of w, x, y equal four and the other equals three. Hence, two of the pairs  $\{x_1, x_2\}, \{x_1, x_3\}, \{x_2, x_3\}$  have codegree nine and one has codegree eight. Adding the constraints  $z_9 \ge 2$  and  $z_8 \ge 1$  to (6.3.1) yields that  $|\partial^2 \mathscr{F}| \ge \binom{m}{2}$ , which contradicts our initial assumption; we omit the details since the computation; we omit the details since the constraints  $z_9 \ge 2$  and  $z_8 \ge 1$  to (6.3.1) yields that  $|\partial^2 \mathscr{F}| \ge \binom{m}{2}$ , which contradicts our initial assumption; we omit the details since the constraints  $z_9 \ge 2$  and  $z_8 \ge 1$  to (6.3.1) yields that  $|\partial^2 \mathscr{F}| \ge \binom{m}{2}$ , which contradicts our initial assumption; we omit the details since the computation is similar to the one in (6.5.1). If s = 4, then w = x = y = 4 so deg $(x_1) = \deg(x_2) = \deg(x_3) = 12$  and codeg  $(\{x_i, x_j\}) = 8$  for  $\{i, j\} \subset \{1, 2, 3\}$ . If we add the constraint  $z_8 \ge 3$  to (6.3.1), then the corresponding dual linear program is

Maximize 
$$360w + 336x + y$$
 (6.5.4)  
subject to:  
 $\binom{i}{2}w + ix \le 1, \quad i \in [16] \setminus \{8\}$   
 $\binom{8}{2}w + 8x + y \le 1.$   
 $y \ge 0.$ 

It follows that (-1/3, 2/3, 5) is a feasible solution to (6.5.4) so  $|\partial^2 \mathscr{F}| \ge 119$ . Since  $\binom{16}{2} = 120$ , we obtain a contradiction unless (-1/3, 2/3, 5) is an optimal solution to (6.5.4). If (-1/3, 2/3, 5) is an optimal solution to (6.5.4), then by complementary slackness,  $\operatorname{codeg}(\{x, y\}) \in \{0, 2, 3, 8\}$  for  $\{x, y\} \in \binom{X}{2}$ ; moreover  $z_8 = 3$ . Observe that

t = u = v = z = 0 implies that if  $w \in X \setminus \{x_1, x_2, x_3\}$ , then

$$\deg(w) = \sum_{\{i,j\}\subset\{1,2,3\}} \operatorname{codeg}\left(\{x_i, x_j, w\}\right) - 2\operatorname{codeg}\left(\{x_1, x_2, x_3, w\}\right) \le 9,$$

so (6.4.9) shows that  $x_1, x_2, x_3$  are the only heavy vertices. Moreover, every  $F \in \mathscr{F}$  contains either exactly two or exactly three heavy vertices as t = u = v = z = 0. If  $F' \in \mathscr{F}$  is a set with exactly two heavy vertices, then (6.4.9) and (6.4.10) yield that either F' contains three vertices of degree six and two vertices of degree five or four vertices of degree six and one of degree four. Now, every light vertex w must be contained in a set with exactly two heavy vertices; otherwise  $\deg(w) = \operatorname{codeg}(x_1, x_2, x_3, w) = 1$ , which contradicts the fact that  $\operatorname{codeg}(\{x, y\}) \in \{0, 2, 3, 8\}$ . Hence,  $\operatorname{deg}(w) \in \{4, 5, 6\}$  for  $w \in X \setminus \{x_1, x_2, x_3\}$ . As a result, if  $\hat{F}$  is a set with three heavy vertices, then (6.4.10) yields that  $\hat{F}$  contains four vertices of degree four. We conclude that  $\operatorname{deg}(w) \in \{4, 6\}$  for  $w \in X \setminus \{x_1, x_2, x_3\}$ . Since s = 4, there are sixteen vertices of degree four. Let  $n_i$  denote the number of vertices of degree i; we have

$$112 = km = \sum_{x \in X} \deg(x) = 12n_{12} + 6n_6 + 4n_4 = 36 + 6n_6 + 64$$

so  $n_6 = 2$ , which is impossible.

If k = 6, then  $\varepsilon \in \{1,2\}$  by Theorem 6.4.1. Let  $n_i$  denote the number of vertices of degree *i*. If  $\varepsilon = 2$ , then  $n_1 = 2$ ,  $n_4 = 9$ , and  $n_{10} = 4$  by a result of Vanstone [111]. Using (6.4.10), we see that  $\mathscr{F}$  is uniquely determined and must be the family

$$\{\{1,2,3,5,6,7\},\{1,2,3,8,9,10\},\{1,2,3,11,12,13\},\{1,2,4,5,8,11\}, \\ \{1,2,4,6,9,12\},\{1,2,4,7,10,13\},\{1,3,4,5,10,12\},\{1,3,4,6,8,13\}, \\ \{1,3,4,7,9,11\},\{2,3,4,5,9,13\},\{2,3,4,6,10,11\},\{2,3,4,7,8,12\}, \\ \{1,2,3,4,14,15\}\}.$$
 (6.5.5)

Hence,  $|\partial^2 \mathscr{F}| = 87 > \binom{13}{2}$ . If k = 6 and  $\varepsilon = 1$ , then either  $n_1 = 2$ ,  $n_3 = 3$ ,  $n_4 = 6$ ,  $n_9 = 3$ , and  $n_{10} = 1$  or  $n_4 = 9$  and  $n_9 = 4$  by a result of Vanstone [111]. Again using (6.4.10), we can conclude that  $\mathscr{F}$  is uniquely determined in both cases. In the first case,  $\mathscr{F}$  must

$$\{\{1,2,4,5,6,11\},\{1,2,4,7,8,12\},\{1,2,4,9,10,13\},\{1,3,4,7,9,11\}, \\ \{1,3,4,5,10,12\},\{1,3,4,6,8,13\},\{2,3,4,8,10,11\},\{2,3,4,6,9,12\}, \\ \{2,3,4,5,7,13\},\{1,2,3,5,8,9\},\{1,2,3,6,7,10\},\{1,2,3,4,14,15\}\};$$
(6.5.6)

hence,  $|\partial^2 \mathscr{F}| = 84 > \binom{12}{2}$ . In the latter case,  $\mathscr{F}$  is the complement of a projective plane of order 3 with respect to a line; hence  $|\partial^2 \mathscr{F}| = 78 > \binom{12}{2}$ .

Chapter 6, in part, is a reprint of the material as it appears in "On a conjecture of Frankl and Füredi," 2011. Chowdhury, Ameera. *Electron. J. Combin.*, 18(1):Paper 56, 16, 2011. The dissertation author was the primary investigator and author of this paper.

## Chapter 7

# On the Manickam-Miklós-Singhi Conjecture

For  $k \in \mathbb{Z}^+$ , let f(k) be the minimum integer N such that for all  $n \ge N$ , every set of *n* real numbers with nonnegative sum has at least  $\binom{n-1}{k-1}$  *k*-element subsets whose sum is also nonnegative. In 1988, Manickam, Miklós, and Singhi proved that f(k) exists and conjectured that  $f(k) \le 4k$ . In this chapter, we prove f(3) = 11 and  $f(4) \le 24$ , which improves previous upper bounds in these cases. With more patience, our arguments could yield improved upper bounds on f(k) for larger *k*. Moreover, we show how our method could potentially yield a quadratic upper bound on f(k). We end this chapter by discussing a vector space analog of the Manickam-Miklós-Singhi conjecture.

#### 7.1 Nonnegative Sums

Manickam, Miklós, and Singhi conjectured in [88] and [89] that

**Conjecture 7.1.1.** For any integers n, k with  $n \ge 4k$ , every set of n real numbers with nonnegative sum has at least  $\binom{n-1}{k-1}$  k-element subsets whose sum is also nonnegative.

This conjecture is similar to the Erdős-Ko-Rado theorem, Theorem 2.2.2, not only in the appearance of the binomial coefficient  $\binom{n-1}{k-1}$ , but also because of the following tight example:  $x_1 = n - 1, x_2 = \cdots = x_n = -1$ . In this example, a *k*-subset of

 $x_1, \ldots, x_n$  has nonnegative sum exactly when it contains  $x_1$ , and hence there are exactly  $\binom{n-1}{k-1}$  *k*-element subsets with nonnegative sum.

As in the Erdős-Ko-Rado theorem, *n* must be large enough with respect to *k*, otherwise there exist *n* real numbers  $x_1, \ldots, x_n \in \mathbb{R}$  with nonnegative sum and fewer than  $\binom{n-1}{k-1}$  *k*-element subsets with nonnegative sum. Here are two counterexamples when n < 4k. For n = 2k + r, where  $1 \le r \le 3k/5$ , let  $x_1 = \cdots = x_{2k+r-2} = 2$  and  $x_{2k+r-1} = x_{2k+r} = -(2k+r-2)$ . Note that a *k*-subset is nonnegative exactly when it does not contain  $x_{2k+r-1}$  or  $x_{2k+r}$  and that  $\binom{2k+r-2}{k} < \binom{2k+r-1}{k-1}$  when  $1 \le r \le 3k/5$ . For n = 3k + r, where  $1 \le r \le k/7$ , a similar counterexample sets  $x_1 = \cdots = x_{3k+r-3} = 3$  and  $x_{3k+r-2} = \cdots = x_{3k+r} = -(3k+r-3)$ . Again, note that a *k*-subset is nonnegative exactly when it does not contain  $x_{3k-1}, x_{3k}$ , or  $x_{3k+1}$  and that  $\binom{3k+r-3}{k} < \binom{3k+r-1}{k-1}$  when  $1 \le r \le k/7$ . These counterexamples do not generalize to larger *n* because  $t^{t-1} \le (t-1)^t$  for  $t \ge 3$ .

Although the  $n \ge 4k$  requirement is probably not sharp, one reason the conjecture is written with this bound is because Baranyai's theorem, Theorem 3.4.33, or the Greene-Kleitman type argument in Section 4.6.2 verifies Conjecture 7.1.1 when k|n; see Lemma 7.4.1.

Very recently, Alon, Huang, and Sudakov [1] verified Conjecture 7.1.1 when  $n \ge \min\{33k^2, 2k^3\}$ , which substantially improves previous results. They also obtained a Hilton-Milner analog. See their paper for references and historical remarks.

#### 7.2 Notation

We begin with some definitions and notation that are special to this chapter.

**Definition 7.2.1.** Given  $x_1 \ge \cdots \ge x_n \in \mathbb{R}$ , a subset  $S \subset \{x_1, \dots, x_n\}$  is <u>nonnegative</u> if  $\sum_{x \in S} x \ge 0$  and is negative otherwise.

**Definition 7.2.2.** *Given*  $x_1 \ge \cdots \ge x_n \in \mathbb{R}$ *, we define* 

$$\mathscr{F}_k := \{S \subset \{x_1, \dots, x_n\} : |S| = k, S \text{ is nonnegative}\}$$

to be the set of nonnegative k-subsets of  $x_1, \ldots, x_n$ .

**Definition 7.2.3.** We say a pair  $(n,k) \in \mathbb{Z}^+ \times \mathbb{Z}^+$  is good if whenever we are given  $x_1 \geq \cdots \geq x_n \in \mathbb{R}$  with nonnegative sum, we have  $|\mathscr{F}_k| \geq \binom{n-1}{k-1}$ .

**Definition 7.2.4.** Let  $\underline{f(k)}$  be the minimum integer N such that for all  $n \ge N$ , the pair (n,k) is good.

**Definition 7.2.5.** A family  $\mathscr{F} \subset {\binom{X}{k}}$  is a <u>star</u> if there exists  $x \in X$  such that

$$\mathscr{F} = \left\{ F \in \binom{X}{k} : x \in F \right\}.$$

**Definition 7.2.6.** We say a pair  $(n,k) \in \mathbb{Z}^+ \times \mathbb{Z}^+$  is <u>strongly good</u> if it is good and  $|\mathscr{F}_k| = \binom{n-1}{k-1}$  if and only if  $\mathscr{F}_k$  is the star on  $x_1$ .

#### 7.3 Results

Alon, Huang, and Sudakov [1] proved that  $f(k) \le \min\{33k^2, 2k^3\}$  and Manickam, Miklós, and Singhi conjectured that  $f(k) \le 4k$ . Our main result shows that the veracity of Conjecture 7.1.1 boils down to proving its veracity in k-1 base cases.

**Theorem 7.3.1** (Chowdhury). If (n,k) is a good pair, then (n+k,k) is a good pair. Moreover, if (n,k) is strongly good and  $n \ge 2k - 1$ , then (n+k,k) is strongly good.

Hence, we can verify Conjecture 7.1.1 for small k.

**Theorem 7.3.2** (Chowdhury). We have f(3) = 11 and, more precisely, that the pair (n,3) is strongly good when  $n \ge 11$ . We have  $f(4) \le 24$  and, more precisely, that the pair (n,4) is strongly good if n = 22 or  $n \ge 24$ .

The previous best upper bounds for f(3) and f(4) were  $f(3) \le 12$  from [87, 91] and  $f(4) \le 128$  from [1]. Although Conjecture 7.1.1 for k = 3 was previously tackled, our result is stronger because we determine f(3) exactly and we characterize the case of equality; moreover our proof is simpler and provides a nice application of the Kruskal-Katona theorem. With more patience, our arguments could yield improved upper bounds on f(k) for larger k. In Section 7.4, we also show how our method might yield a quadratic upper bound on f(k) that improves on the current best upper bound [1] of  $f(k) \le \min\{33k^2, 2k^3\}$ .

#### 7.4 Reductions to Base Cases

This section contains a proof of Theorem 7.3.1. We first show that Baranyai's theorem, Theorem 3.4.33, or the Greene-Kleitman type argument in Section 4.6.2 yields a lemma due to Bier and Manickam [13] that proves Conjecture 7.1.1 in the case that k|n.

**Lemma 7.4.1** (Bier-Manickam). For  $k \in \mathbb{Z}^+$ , the pair (ck, k) is good for any  $c \in \mathbb{Z}^+$  and is strongly good if  $c \neq 2$ .

**Proof.** We first show that Baranyai's theorem implies that the pair (ck,k) is good for any  $c \in \mathbb{Z}^+$ . Let n = ck and suppose that  $x_1 \ge \cdots \ge x_{ck} \in \mathbb{R}$  has nonnegative sum. We will prove that if  $\mathscr{G} \subset {X \choose k}$  is a perfect matching, then there is a nonnegative *k*-set  $G \in \mathscr{G}$ . Since the total sum of  $x_1, \ldots, x_{ck}$  is nonnegative and the *k*-subsets in  $\mathscr{G}$  are disjoint,

$$0 \leq \sum_{i=1}^{ck} x_i = \sum_{G \in \mathscr{G}} \left( \sum_{i \in G} x_i \right).$$

Hence, at least one *k*-subset  $G \in \mathscr{G}$  must have nonnegative sum. As n = ck, Baranyai's theorem asserts that the complete hypergraph on ck vertices  $\binom{X}{k}$  has a partition  $\mathscr{P}$  into  $\binom{ck-1}{k-1}$  perfect matchings. Each perfect matching in  $\mathscr{P}$  contains a nonnegative *k*-subset, and as  $\mathscr{P}$  is a partition, the corresponding *k*-subsets are distinct. Hence, there are at least  $\binom{ck-1}{k-1}$  *k*-element subsets with nonnegative sum.

We do not need the full power of Baranyai's theorem to prove the statement, however. The Greene-Kleitman type argument in Section 4.6.2 not only yields the same conclusion, but also allows us to prove that (ck,k) is strongly good when  $c \neq 2$ . Let  $\mathscr{G} \subset {X \choose k}$  be a perfect matching and let  $\pi \in S_{ck}$  be a permutation. As in Lemma 4.5.7, the family  $\pi(\mathscr{G}) := {\pi(G) : G \in \mathscr{G}}$  is also a perfect matching. Recall that  $\mathscr{F}_k \subset {X \choose k}$ from Definition 7.2.2 is the family of nonnegative *k*-subsets in  $x_1, \ldots, x_{ck}$ . We showed in the preceding paragraph that any perfect matching contains a nonnegative *k*-subset so  $|\mathscr{F}_k \cap \pi(\mathscr{G})| \ge 1$  for any  $\pi \in S_{ck}$ . On the other hand, given  $G \in \mathscr{G}$  and  $F \in \mathscr{F}_k$ , there are k!(ck-k)! permutations  $\pi \in S_{ck}$  such that  $\pi(G) = F$ . Consequently,

$$\left(\frac{ck}{k}\right)|\mathscr{F}_k|k!(ck-k)! = |\mathscr{G}||\mathscr{F}_k|\{\pi \in S_{ck} : \pi(G) = F\} = \sum_{\pi \in S_{ck}}|\mathscr{F}_k \cap \pi(\mathscr{G})| \ge (ck)!$$
(7.4.1)

Rearranging (7.4.1) yields that  $|\mathscr{F}_k| \ge \binom{ck-1}{k-1}$ , as desired.

If there are exactly  $\binom{ck-1}{k-1}$  nonnegative *k*-subsets, then equality holds everywhere in (7.4.1). Hence, for each  $\pi \in S_{ck}$ , we have  $|\mathscr{F}_k \cap \pi(\mathscr{G})| = 1$ . This proves that every perfect matching contains exactly one nonnegative *k*-subset in  $\mathscr{F}_k$ , which implies that  $\mathscr{F}_k \subset \binom{X}{k}$  is intersecting. By the Erdős-Ko-Rado theorem, Theorem 2.2.2, if  $c \neq 2$ , then  $\mathscr{F}_k$  must be the star on  $x_1$ . Hence, (ck, k) is strongly good if  $c \neq 2$ .

Our next lemma shows that if a set of n' real numbers has at least  $\binom{n'-1}{d-1}$  nonnegative *d*-subsets and (d,k) is a good pair, then it has at least  $\binom{n'-1}{k-1}$  nonnegative *k*-subsets.

**Lemma 7.4.2.** Suppose  $x_1 \ge \cdots \ge x_{n'} \in \mathbb{R}$  has  $|\mathscr{F}_d| \ge \binom{n'-1}{d-1}$ . If (d,k) is a good pair, then  $|\mathscr{F}_k| \ge \binom{n'-1}{k-1}$ . Moreover, if  $|\mathscr{F}_k| = \binom{n'-1}{k-1}$ ,  $d \ge 2k-1$ , and the pair (d,k) is strongly good, then  $\mathscr{F}_k$  is the star on  $x_1$ .

**Proof.** Count pairs (A,B) where  $A \in \mathscr{F}_d$ ,  $B \in \mathscr{F}_k$ , and  $B \subset A$ . Since (d,k) is a good pair, each  $A \in \mathscr{F}_d$  contains at least  $\binom{d-1}{k-1}$  sets  $B \in \mathscr{F}_k$ . On the other hand, each  $B \in \mathscr{F}_k$  is contained in at most  $\binom{n'-k}{d-k}$  sets  $A \in \mathscr{F}_d$ . Putting all this together, we have

$$\binom{n'-1}{d-1}\binom{d-1}{k-1} \le |\{(A,B): A \in \mathscr{F}_d, B \in \mathscr{F}_k, B \subset A\}| \le |\mathscr{F}_k|\binom{n'-k}{d-k}.$$
 (7.4.2)

Hence,  $|\mathscr{F}_k| \ge \binom{n'-1}{k-1}$ .

If  $|\mathscr{F}_k| = \binom{n'-1}{k-1}$ , then (7.4.2) implies that  $|\mathscr{F}_d| = \binom{n'-1}{d-1}$ , that each  $A \in \mathscr{F}_d$  contains exactly  $\binom{d-1}{k-1}$  sets  $B \in \mathscr{F}_k$ , and that if  $B \in \mathscr{F}_k$ , then every *d*-set that contains *B* lies in  $\mathscr{F}_d$ . Clearly,  $\{x_1, \ldots, x_k\} \in \mathscr{F}_k$  as it is the *k*-set with largest sum. Let  $B' \subset \{x_1, \ldots, x_{n'}\}$  be any *k*-subset containing  $x_1$ . We will show that  $B' \in \mathscr{F}_k$ . Since  $d \ge 2k-1$ , there exists a *d*-subset *A'* that contains  $B' \cup \{x_1, \ldots, x_k\}$ . Observe that  $A' \in \mathscr{F}_d$  because it contains  $\{x_1, \ldots, x_k\} \in \mathscr{F}_k$ . Now *A'* contains exactly  $\binom{d-1}{k-1}$  sets  $B \in \mathscr{F}_k$  and since (d, k) is strongly good, these sets form the star on  $x_1$ . Hence,  $B' \in \mathscr{F}_k$  as it contains  $x_1$  and lies in *A'* so  $\mathscr{F}_k$  is the star on  $x_1$ .

Lemma 7.4.2 has two corollaries; the first is Theorem 7.3.1.

**Proof of Theorem 7.3.1.** Suppose, for a contradiction, that (n,k) is a good pair, but that (n+k,k) is not a good pair. Consequently, there exist  $x_1 \ge \cdots \ge x_{n+k} \in \mathbb{R}$  with nonnegative sum and  $|\mathscr{F}_k| < \binom{n+k-1}{k-1}$ . By the Pascal rule, there are greater than  $\binom{n+k-1}{k}$ 

negative *k*-subsets. As  $\sum_{i=1}^{n+k} x_i \ge 0$ , the complement of a negative *k*-subset must be a positive *n*-subset. Hence,  $|\mathscr{F}_n| > \binom{n+k-1}{k} = \binom{n+k-1}{n-1}$ . We may now apply Lemma 7.4.2 with n' = n + k and d = n to conclude that  $|\mathscr{F}_k| \ge \binom{n+k-1}{k-1}$ , which contradicts our initial assumption. Hence, if (n,k) is good, then (n+k,k) is also good. If (n,k) is strongly good and  $n \ge 2k - 1$ , then a similar argument yields that  $\mathscr{F}_k$  is the star on  $x_1$ .

The second corollary of Lemma 7.4.2 is a result of Manickam and Singhi that shows that if (n,k) is a good pair, then (cn,k) is a good pair for any  $c \in \mathbb{Z}^+$ .

**Corollary 7.4.3** (Manickam-Singhi, [89]). If (n,k) is a good pair, then (cn,k) is a good pair for any  $c \in \mathbb{Z}^+$ . If (n,k) is a strongly good pair and  $n \ge 2k - 1$ , then (cn,k) is strongly good for any  $c \in \mathbb{Z}^+$ .

**Proof.** Let  $x_1 \ge \cdots \ge x_{cn} \in \mathbb{R}$  have nonnegative sum. By Lemma 7.4.1, the pair (cn, n) is good so  $|\mathscr{F}_n| \ge \binom{cn-1}{n-1}$ . If the pair (n,k) is good, then we may apply Lemma 7.4.2 with n' = cn and d = n to conclude that  $|\mathscr{F}_k| \ge \binom{cn-1}{k-1}$ . If  $|\mathscr{F}_k| = \binom{cn-1}{k-1}$ , the pair (n,k) is strongly good, and  $n \ge 2k - 1$ , then we may apply Lemma 7.4.2 again to conclude that  $\mathscr{F}_k$  is the star on  $x_1$ .

We now show that if gcd(r,k) = 1 and  $(\alpha k + r,k)$  is a (strongly) good pair, then (n,k) is a (strongly) good pair for any  $n \ge (\alpha k + r)(k-1)$  and so  $f(k) \le (\alpha k + r)(k-1)$ . Hence, if gcd(r,k) = 1 and we could show  $(\alpha k + r,k)$  is a good pair for some  $\alpha < 33$ , then we could improve the current best upper bound  $f(k) \le \min\{33k^2, 2k^3\}$  in [1]. Two natural candidates for this approach would be the pairs (4k - 1, k) and (4k + 1, k), but we were not able to show that these were good pairs in general.

**Corollary 7.4.4.** If gcd(r,k) = 1 and  $(\alpha k + r,k)$  is a (strongly) good pair, then (n,k) is a (strongly) good pair for any  $n \ge (\alpha k + r)(k-1)$ , and hence  $f(k) \le (\alpha k + r)(k-1)$ .

**Proof** If  $(\alpha k + r, k)$  is a (strongly) good pair, then  $((\alpha k + r)i, k)$  is a (strongly) good pair, where  $i \in [k-1]$ , by Corollary 7.4.3. We have  $(\alpha k + r)i \equiv ri \pmod{k}$  and since gcd(r,k) = 1, we have hit all nonzero congruence classes modulo k. Hence, (n,k) is a (strongly) good pair for any  $n \ge (\alpha k + r)(k - 1)$  by Theorem 7.3.1 and Lemma 7.4.1. As a result,  $f(k) \le (\alpha k + r)(k - 1)$ .

#### 7.5 The MMS Conjecture when k is Three

It is not difficult to see that f(2) = 6 and that (n,2) is a strongly good pair if and only if n = 6 or  $n \ge 8$ . In this section, we prove Theorem 7.3.2 in the case that k = 3. That is, we show f(3) = 11 and that (n,3) is a strongly good pair if  $n \ge 11$ . By Theorem 7.3.1, it suffices to show that the pairs (11,3) and (13,3) are strongly good. This is best possible since when  $n \in \{4, 5, 7, 8, 10\}$ , the pair (n,3) is not good. In our proof, we will use the full version of the Kruskal-Katona theorem, Theorem 2.1.4. We first observe that we can get a lower bound on  $|\mathscr{F}_k|$  given that it contains a certain sum.

**Lemma 7.5.1.** Define  $F_1(i_1) = i_1$  and  $F_k(i_1, \ldots, i_k)$  recursively by

$$F_k(i_1,...,i_k) = \binom{i_k}{k} - \binom{i_k - i_1}{k} - \sum_{j=1}^{k-2} F_j(i_1,...,i_j) \binom{i_k - i_{j+1}}{k-j}.$$

If  $\sum_{l=1}^{k} x_{i_l} \in \mathscr{F}_k$ , where  $i_1 < \cdots < i_k$ , then  $|\mathscr{F}_k| \ge F_k(i_1, \ldots, i_k)$ .

**Proof.** We assume  $x_1 \ge \cdots \ge x_n$ . Hence, if  $\sum_{l=1}^k x_{i_l} \in \mathscr{F}_k$  then  $\mathscr{F}_k$  contains  $\sum_{l=1}^k x_{j_l}$ , where  $j_l \le i_l$  for  $l \in [k]$  and  $j_1 < \cdots < j_k$ . By induction,  $F_k(i_1, \ldots, i_k)$  counts the number of *k*-tuples  $(j_1, \ldots, j_k)$  satisfying  $j_l \le i_l$  for  $l \in [k]$  and  $j_1 < \cdots < j_k$ . Hence, if  $\sum_{l=1}^k x_{i_l}$  lies in  $\mathscr{F}_k$ , where  $i_1 < \cdots < i_k$ , then  $|\mathscr{F}_k| \ge F_k(i_1, \ldots, i_k)$ .

#### Lemma 7.5.2. The pair (11,3) is strongly good.

**Proof.** We have  $x_1 \ge \cdots \ge x_{11}$  satisfying  $\sum_{i=1}^{11} x_i \ge 0$ . We first show that  $|\mathscr{F}_3| > {\binom{10}{2}}$  if  $x_1 + x_{11} \le 0$ . Since  $\sum_{i=1}^{11} x_i \ge 0$ , we have  $\sum_{i=2}^{10} x_i \ge 0$ . Let  $\mathscr{S}$  be the family of nonnegative 3-subsets among  $x_2, \ldots, x_{10}$ . By Lemma 7.4.1,  $|\mathscr{S}| \ge {\binom{8}{2}} = 28$  and Theorem 2.1.4 yields that  $|\partial \mathscr{S}| \ge 20$ . As we can add  $x_1$  to any 2-subset in  $\partial \mathscr{S}$  to form a nonnegative 3-set, we have  $|\mathscr{F}_3| \ge 28 + 20 > {\binom{10}{2}}$ . Hence, we may assume that  $x_1 + x_{11} > 0$ .

We may assume  $x_1 + x_{10} + x_{11} < 0$  as otherwise  $\mathscr{F}_3$  contains the star on  $x_1$ . Since  $\sum_{i=1}^{11} x_i = 0$ , we must have  $\sum_{i=2}^{7} x_i \ge (6/8) \sum_{i=2}^{9} x_i > 0$ . Let  $\mathscr{W}$  be the family of nonnegative 3-subsets among  $x_2, \ldots, x_7$ . Lemma 7.4.1 yields that  $|\mathscr{W}| \ge 10$ . Note that  $x_1 + x_2 + x_{11} \in \mathscr{F}_3$  since  $x_2$  and  $x_1 + x_{11}$  are nonnegative. If  $x_1 + x_8 + x_{10} \ge 0$ , then Lemma 7.5.1 implies that  $|\mathscr{F}_3| \ge 35 + 10 + 1 > {10 \choose 2}$  by taking into account the sets in  $\mathscr{W}$  and  $x_1 + x_2 + x_{11}$ . Hence, by Lemma 7.5.1, we may assume that

$$x_1 + x_8 + x_{10} < 0, \ x_2 + x_7 + x_9 < 0, \ x_3 + x_4 + x_{11} < 0.$$

$$(7.5.1)$$

Equation (7.5.1) and  $\sum_{i=1}^{11} x_i \ge 0$  yield that  $x_5 + x_6 > 0$  so  $x_5 > 0$ .

By (7.5.1), we have  $x_1 + x_9 + x_{10} \le x_1 + x_8 + x_{10} < 0$ . Since  $\sum_{i=1}^{11} x_i \ge 0$ , either  $x_2 + x_{11} > 0$  or  $\sum_{i=3}^{8} x_i > 0$ . In the first case,  $x_2 + x_5 + x_{11} > 0$  since  $x_5 > 0$ , so  $|\mathscr{F}_3| > {10 \choose 2}$  by Lemma 7.5.1. We now show that if  $\sum_{i=3}^{8} x_i \ge 0$ , then  $|\mathscr{F}_3| > {10 \choose 2}$ . Let  $\mathscr{T}$  be the family of nonnegative 3-subsets among  $x_3, \ldots, x_8$ . By Lemma 7.4.1,  $|\mathscr{T}| \ge {5 \choose 2} = 10$  and Theorem 2.1.4 yields that  $|\partial \mathscr{T}| \ge 10$ . We can add either  $x_1$  or  $x_2$  to any 2-subset in  $\partial \mathscr{T}$  to form a nonnegative 3-subset. The family  $\mathscr{F}_3$  also contains

$$x_1 + x_2 + x_i, i \in \{3, \dots, 11\}; x_1 + x_b + x_c, 3 \le b \le 5, c \in \{9, 10, 11\},$$
 (7.5.2)

since  $x_1 + x_{11} > 0$  and  $x_5 > 0$ . Taking into account the sets in  $\mathscr{T}$ , the sets formed by adding  $x_1$  or  $x_2$  to a set in  $\partial \mathscr{T}$ , and the sets in (7.5.2),  $|\mathscr{F}_3| \ge 10 + 10 + 10 + 18 > \binom{10}{2}$ .

The preceding arguments show that if  $x_1 + x_{10} + x_{11} < 0$ , then  $|\mathscr{F}_3| > \binom{10}{2}$ . Hence,  $|\mathscr{F}_3| \ge \binom{10}{2}$  and  $|\mathscr{F}_3| = \binom{10}{2}$  if and only if  $\mathscr{F}_3$  is the star on  $x_1$ . Consequently, (11,3) is a strongly good pair.

Lemma 7.5.3. The pair (13,3) is strongly good.

**Proof.** We have  $x_1 \ge \cdots \ge x_{13}$  satisfying  $\sum_{i=1}^{13} x_i \ge 0$ . As in the proof of Lemma 7.5.2, if  $x_1 + x_{13} \le 0$ , then Theorem 2.1.4 implies  $|\mathscr{F}_3| > \binom{12}{2}$ . Hence, we may assume that  $x_1 + x_{13} > 0$ .

We may assume  $x_1 + x_{12} + x_{13} < 0$  as otherwise  $\mathscr{F}_3$  contains the star on  $x_1$ . Let  $\mathscr{T}$  be the family of nonnegative 3-subsets among  $x_2, \ldots, x_{10}$ . As in the proof of Lemma 7.5.2, we may conclude that  $|\mathscr{T}| \ge {\binom{8}{2}} = 28$ . Note that  $x_1 + x_2 + x_i \in \mathscr{F}_3$  for  $i \in \{11, 12, 13\}$  since  $x_2$  and  $x_1 + x_{13}$  are nonnegative. If  $x_1 + x_9 + x_{10} \ge 0$ , then Lemma 7.5.1 implies that  $|\mathscr{F}_3| \ge 36 + 28 + 3 > {\binom{12}{2}}$  by taking into account the sets in  $\mathscr{T}$  and the sets  $x_1 + x_2 + x_i$  for  $i \in \{11, 12, 13\}$ . Lemma 7.5.1 thus implies

$$x_1 + x_9 + x_{10} < 0, \ x_3 + x_8 + x_{12} \le x_2 + x_7 + x_{11} < 0, \ x_4 + x_5 + x_{13} < 0.$$
(7.5.3)

Equation (7.5.3) and  $\sum_{i=1}^{13} x_i \ge 0$  yield that  $x_6 > 0$  so  $x_1 + x_6 + x_{13} > 0$ . Consequently Lemma 7.5.1 implies that  $|\mathscr{F}_3| \ge 45 + 28 > \binom{12}{2}$  by taking into account the sets in  $\mathscr{T}$ . We conclude that (13,3) is a strongly good pair.

#### 7.6 The MMS Conjecture when k is Four

The preceding arguments yield improved upper bounds on f(k) for k > 3. For example, in this section, we prove Theorem 7.3.2 in the case that k = 4. That is, we show  $f(4) \le 24$  and that (n, 4) is a strongly good pair if n = 22 or  $n \ge 24$ . Taking into account the counterexamples in Section 7.1, this shows  $14 \le f(4) \le 24$ . By Lemma 7.4.1 and Theorem 7.3.1, it suffices to prove that (22, 4), (25, 4), and (27, 4) are strongly good pairs.

#### Lemma 7.6.1. The pair (22,4) is strongly good.

**Proof.** We have  $x_1 \ge \cdots \ge x_{22}$  satisfying  $\sum_{i=1}^{22} x_i \ge 0$ . As in the proof of Lemma 7.5.2, if  $x_1 + x_{22} \le 0$ , then Theorem 2.1.4 implies  $|\mathscr{F}_4| > {\binom{21}{3}}$ . Hence, we may assume that  $x_1 + x_{22} > 0$ .

We may assume  $x_1 + x_{20} + x_{21} + x_{22} < 0$  as otherwise  $\mathscr{F}_4$  contains the star on  $x_1$ . Let  $\mathscr{S}$  be the family of nonnegative 4-subsets among  $x_2, \ldots, x_{17}$ . As in the proof of Lemma 7.5.2, we may conclude that  $|\mathscr{S}| \ge {\binom{15}{3}}$ . Consequently, if there are greater than  ${\binom{21}{3}} - {\binom{15}{3}} = 875$  nonnegative 4-sets on  $x_1$ , then  $|\mathscr{F}_4| > {\binom{21}{3}}$ . Hence, by Lemma 7.5.1, we may assume that

$$x_1 + x_{11} + x_{14} + x_{22} < 0, \quad x_2 + x_9 + x_{15} + x_{21} < 0, \quad x_3 + x_7 + x_{16} + x_{19} < 0, \quad (7.6.1)$$
  
$$x_4 + x_{10} + x_{12} + x_{18} < 0, \quad x_5 + x_6 + x_{13} + x_{20} < 0.$$

Since  $\sum_{i=1}^{22} x_i \ge 0$ , equation (7.6.1) implies that  $x_8 + x_{17} > 0$ . We consequently have that  $x_1 + x_8 + x_{17} + x_{22} > 0$  since  $x_1 + x_{22} > 0$ , which implies that  $|\mathscr{F}_4| > {21 \choose 3}$  by Lemma 7.5.1. We conclude that (22,4) is a strongly good pair.

#### Lemma 7.6.2. The pair (25,4) is strongly good.

**Proof.** We have  $x_1 \ge \cdots \ge x_{25}$  with  $\sum_{i=1}^{25} x_i \ge 0$ . We may assume  $x_1 + x_{23} + x_{24} + x_{25} < 0$ , as otherwise  $\mathscr{F}_4$  contains the star on  $x_1$ . Let  $\mathscr{S}$  be the family of all nonnegative 4-sets in  $x_2, \ldots, x_{21}$ . As in the proof of Lemma 7.5.2, we may conclude that  $|\mathscr{S}| \ge {19 \choose 3}$ . Consequently, if there are greater than  ${24 \choose 3} - {19 \choose 3} = 1055$  nonnegative 4-sets on  $x_1$  then

 $|\mathscr{F}_4| > \binom{24}{3}$ . Hence, by Lemma 7.5.1, we may assume that

$$x_1 + x_7 + x_{18} + x_{25} < 0, \quad x_2 + x_8 + x_{19} + x_{24} < 0, \quad x_3 + x_9 + x_{16} + x_{23} < 0, \quad (7.6.2)$$
  
$$x_4 + x_{10} + x_{15} + x_{22} < 0, \quad x_5 + x_{11} + x_{14} + x_{21} < 0, \quad x_6 + x_{12} + x_{13} + x_{20} < 0.$$

Since  $\sum_{i=1}^{25} x_i \ge 0$ , equation (7.6.2) implies that  $x_{17} > 0$ . Hence,  $|\mathscr{F}_4| \ge {\binom{17}{4}} > {\binom{24}{3}}$ . We conclude that (25,4) is a strongly good pair.

#### Lemma 7.6.3. The pair (27,4) is strongly good.

**Proof.** We have  $x_1 \ge \cdots \ge x_{27}$  with  $\sum_{i=1}^{27} x_i \ge 0$ . We may assume  $x_1 + x_{25} + x_{26} + x_{27} < 0$ , as otherwise  $\mathscr{F}_4$  contains the star on  $x_1$ . Let  $\mathscr{S}$  be the family of all nonnegative 4-sets in  $x_2, \ldots, x_{21}$ . As in the proof of Lemma 7.5.2, we may conclude that  $|\mathscr{S}| \ge {19 \choose 3}$ . Consequently, if there are greater than  ${26 \choose 3} - {19 \choose 3} = 1631$  nonnegative 4-sets on  $x_1$  then  $|\mathscr{F}_4| > {26 \choose 3}$ . Hence by Lemma 7.5.1, we may assume that

$$x_1 + x_{12} + x_{17} + x_{27} < 0, \quad x_2 + x_8 + x_{22} + x_{26} < 0, \quad x_3 + x_{11} + x_{16} + x_{23} < 0, \quad (7.6.3)$$
  
$$x_4 + x_{10} + x_{13} + x_{25} < 0, \quad x_5 + x_7 + x_{15} + x_{24} < 0, \quad x_6 + x_9 + x_{14} + x_{21} < 0.$$

Since  $\sum_{i=1}^{27} x_i \ge 0$ , equation (7.6.3) implies that  $x_{18} + x_{19} + x_{20} > 0$ . We consequently have  $|\mathscr{F}_4| \ge {\binom{18}{4}} > {\binom{26}{3}}$ . We conclude that (27,4) is a strongly good pair.  $\Box$ 

### 7.7 Open Problems

Like the preceding combinatorial questions in this thesis, the Manickam-Miklós-Singhi conjecture, Conjecture 7.1.1, has a vector space analog about which we know distressingly little. In this section, we show that the methods discussed in Section 7.4 may also be useful for attacking the vector space analog of Conjecture 7.1.1.

Recall that *V* is an *n*-dimensional vector space over a finite field  $\mathbb{F}_q$ . Suppose for each one-dimensional subspace  $v \in {V \brack 1}$ , we assign a weight  $f(v) \in \mathbb{R}$  such that  $\sum_{v \in {V \brack 1}} f(v) = 0$ . Define the weight of a subspace  $S \subset V$  to be the sum of the weights of all its one-dimensional subspaces,

$$f(S) := \sum_{v \in [{V \atop 1}], v \subseteq S} f(v).$$
(7.7.1)

The family of k-dimensional subspaces with nonnegative weight will be denoted by

$$\mathscr{F}_{V,f,k} := \left\{ S \in \begin{bmatrix} V \\ k \end{bmatrix} : f(S) \ge 0 \right\}.$$

The vector space analog of Conjecture 7.1.1 states

**Conjecture 7.7.1** (Manickam-Singhi, [89]). Let V be an n-dimensional vector space over a finite field  $\mathbb{F}_q$  and let  $f: \begin{bmatrix} V \\ 1 \end{bmatrix} \to \mathbb{R}$  be a weighting of the one-dimensional subspaces of V such that  $\sum_{v \in \begin{bmatrix} V \\ 1 \end{bmatrix}} f(v) = 0$ . If  $n \ge 4k$ , then  $|\mathscr{F}_{V,f,k}| \ge \begin{bmatrix} n-1 \\ k-1 \end{bmatrix}$ .

Unlike Conjecture 7.1.1, we do not have a good reason for the  $n \ge 4k$  stipulation. The counterexamples in Section 7.1 do not generalize to the vector space case. In fact, there are no known counterexamples to Conjecture 7.7.1 for n > 2k, and it is easy to construct counterexamples to Conjecture 7.7.1 when k < n < 2k. Hence, it is possible that Conjecture 7.7.1 is true for  $n \ge 2k$ .

In this section, we show that the veracity of Conjecture 7.7.1 for k = 2 also boils down to proving its veracity in a few base cases. As in Section 7.4, given integers  $n, k \in \mathbb{Z}^+$  we say the pair [n, k] is good if whenever we are given an *n*-dimensional vector space *V* over  $\mathbb{F}_q$  and a weighting  $f : \begin{bmatrix} V \\ 1 \end{bmatrix} \to \mathbb{R}$  with  $\sum_{v \in \begin{bmatrix} V \\ 1 \end{bmatrix}} f(v) = 0$ , we have

$$|\mathscr{F}_{V,f,k}| \ge {n-1 \brack k-1}.$$
(7.7.2)

Similarly, the pair [n,k] is strongly good if it is good and (7.7.2) holds with equality if and only if  $\mathscr{F}_{V,f,k}$  is the star on the one-dimensional subspace  $\hat{v} \in \begin{bmatrix} V \\ 1 \end{bmatrix}$  with largest weight,

$$\mathscr{F}_{V,f,k} := \left\{ S \in \begin{bmatrix} V \\ k \end{bmatrix} : \hat{v} \subset S \right\}, \ f(\hat{v}) \ge f(v) \ \forall \ v \in \begin{bmatrix} V \\ 1 \end{bmatrix}.$$

The results and proofs of Lemma 7.4.2, Lemma 7.4.1, and Corollary 7.4.3 generalize straightforwardly to the vector space setting. The proof of Theorem 7.3.1, however, does not readily generalize because of the issues discussed in Section 4.6.1.2. We have not yet been able to surmount these difficulties to prove a vector space analog of Theorem 7.3.1. Nevertheless, we can still show that for k = 2, the veracity of Conjecture 7.7.1 boils down to proving its veracity in a few base cases.

**Lemma 7.7.2.** If  $m \in \mathbb{Z}^+$  is odd and [m, 2] is a (strongly) good pair, then [cm - 1, 2] and [cm + 1, 2] are (strongly) good pairs for any  $c \in \mathbb{Z}^+$ .

**Proof.** Since *m* is odd, we may assume that  $c \ge 2$ . Let *V* be a (cm-1)-dimensional vector space over  $\mathbb{F}_q$  and let  $f : {V \choose 1} \to \mathbb{R}$  be a weighting of the one-dimensional spaces such that  $\sum_{v \in {V \choose 1}} f(v) = 0$ . If  $|\mathscr{F}_{V,f,m}| \ge {cm-2 \choose m-1}$ , then by the vector space analog of Lemma 7.4.2 with n = cm-1, d = m, and k = 2, we have  $|\mathscr{F}_{V,f,2}| \ge {cm-2 \choose 1}$ . Hence, we may assume that  $|\mathscr{F}_{V,f,m}| < {cm-2 \choose m-1}$ .

Embed V in W, a *cm*-dimensional vector space over  $\mathbb{F}_q$  and extend f to a weighting  $\hat{f}: \begin{bmatrix} W \\ 1 \end{bmatrix} \to \mathbb{R}$  by giving every one-dimensional  $w \in \begin{bmatrix} W \\ 1 \end{bmatrix} \setminus \begin{bmatrix} V \\ 1 \end{bmatrix}$  a weight  $\hat{f}(w) = 0$ . Since  $\sum_{w \in \begin{bmatrix} W \\ 1 \end{bmatrix}} \hat{f}(w) = 0$ , the vector space analog of Lemma 7.4.1 yields that  $|\mathscr{F}_{W,\hat{f},m}| \ge \begin{bmatrix} cm-1 \\ m-1 \end{bmatrix}$ . If  $U \in \mathscr{F}_{W,\hat{f},m}$  then  $U \in \mathscr{F}_{V,f,m}$  or  $U \cap V \in \mathscr{F}_{V,f,m-1}$ . Also note that each S in  $\mathscr{F}_{V,f,m-1}$  lies in  $q^{(c-1)m}$  spaces  $U \in \mathscr{F}_{W,\hat{f},m}$ . By the q-Pascal identity,

 $\begin{bmatrix} cm-2\\m-1 \end{bmatrix} + q^{(c-1)m} \begin{bmatrix} cm-2\\m-2 \end{bmatrix} = \begin{bmatrix} cm-1\\m-1 \end{bmatrix} \le |\mathscr{F}_{W,\hat{f},m}| = |\mathscr{F}_{V,f,m}| + q^{(c-1)m} |\mathscr{F}_{V,f,m-1}|.$ (7.7.3)

By assumption,  $|\mathscr{F}_{V,f,m}| < {\binom{cm-2}{m-1}}$  so (7.7.3) implies that  $|\mathscr{F}_{V,f,m-1}| > {\binom{cm-2}{m-2}}$ . Since *m* is odd, we have that m-1 is even, and hence [m-1,2] is a good pair by Lemma 7.4.1. The vector space analog of Lemma 7.4.2 with n = cm-1, d = m-1, and k = 2 thus yields that  $|\mathscr{F}_{V,f,2}| > {\binom{cm-1}{1}}$ .

If  $|\mathscr{F}_{V,f,2}| = {\binom{cm-1}{1}}$ , then (7.7.3) and the vector space analog of Lemma 7.4.2 imply that  $|\mathscr{F}_{V,f,m}| = {\binom{cm-1}{m-1}}$ . Hence, if [m,2] is strongly good, then the vector space analog of Lemma 7.4.2 implies that  $\mathscr{F}_{V,f,2}$  is the star on the one-dimensional subspace  $\hat{v} \in {\binom{V}{1}}$  with largest weight. This shows that if *m* is odd and the pair [m,2] is (strongly) good, then the pair [cm-1,2] is also (strongly) good for any  $c \in \mathbb{Z}^+$ ; a similar argument shows that the pair [cm+1,2] is also (strongly) good.

A corollary of the vector space analog of Lemma 7.4.1 and Lemma 7.7.2 is that if the pairs [5,2] and [7,2] are good, then the pair [n,2] is good for any  $n \ge 4$ . If the pair [5,2] is not good, but the pairs [7,2], [9,2] and [11,2] are good, then the vector space analog of Lemma 7.4.1 and Lemma 7.7.2 yield that the pair [n,2] is good for any  $n \ge 6$ .

Chapter 7, in part, is currently being prepared for submission for publication of the material. Chowdhury, Ameera. The dissertation author was the primary investigator and author of this material.

# **Bibliography**

- [1] N. Alon, H. Huang, and B. Sudakov. Nonnegative k-sums, fractional covers, and probability of small deviations. preprint, 2011.
- [2] N. Alon and G. Kalai. A simple proof of the upper bound theorem. *European J. Combin.*, 6(3):211–214, 1985.
- [3] L. Babai. On the nonuniform Fisher inequality. *Discrete Math.*, 66(3):303–307, 1987.
- [4] L. Babai. A short proof of the nonuniform Ray-Chaudhuri-Wilson inequality. *Combinatorica*, 8(1):133–135, 1988.
- [5] L. Babai and P. Frankl. Linear algebra methods in combinatorics. manuscript, 1992.
- [6] R. Baer. Partionen abelscher Gruppen. Arch. Math. (Basel), 14:73–83, 1963.
- [7] R.D. Baker. Partitioning the planes of  $AG_{2m}(2)$  into 2-designs. *Discrete Math.*, 15(3):205–211, 1976.
- [8] R.D. Baker, J.H. van Lint, and R.M. Wilson. On the Preparata and Goethals codes. *IEEE Trans. Inform. Theory*, 29(3):342–345, 1983.
- [9] I. Bárány. A short proof of Kneser's conjecture. J. Combin. Theory Ser. A, 25(3):325–326, 1978.
- [10] Zs. Baranyai. On the factorization of the complete uniform hypergraph. In *Infinite and finite sets (Colloq., Keszthely, 1973; dedicated to P. Erdős on his 60th birthday), Vol. I*, pages 91–108. Colloq. Math. Soc. Jánōs Bolyai, Vol. 10. North-Holland, Amsterdam, 1975.
- [11] A. Beutelspacher. On parallelisms in finite projective spaces. *Geometriae Dedicata*, 3:35–40, 1974.
- [12] S. Bezrukov and A. Blokhuis. A Kruskal-Katona type theorem for the linear lattice. *European J. Combin.*, 20(2):123–130, 1999.

- [13] T. Bier and N. Manickam. The first distribution invariant of the Johnson-scheme. Southeast Asian Bull. Math., 11(1):61–68, 1987.
- [14] A. Blokhuis, A. E. Brouwer, A. Chowdhury, P. Frankl, T. Mussche, B. Patkós, and T. Szőnyi. A Hilton-Milner theorem for vector spaces. *Electron. J. Combin.*, 17(1):Research Paper 71, 12, 2010.
- [15] A. Blokhuis, A.E. Brouwer, T. Szőnyi, and Zs. Weiner. On q-analogues and stability theorems. J. Geometry, 101:31–50, 2011.
- [16] B. Bollobás. On generalized graphs. Acta Math. Acad. Sci. Hungar, 16:447–452, 1965.
- [17] R. C. Bose. A note on Fisher's inequality for balanced incomplete block designs. *Ann. Math. Statistics*, 20:619–620, 1949.
- [18] M. Braun, T. Etzion, and A. Vardy. Linearity and complements in projective space. Preprint, available at http://arxiv.org/abs/1103.3117, 2011.
- [19] W. Bridges. Some results on  $\lambda$ -designs. J. Combinatorial Theory, 8:350–360, 1970.
- [20] W. Bridges and E. Kramer. The determination of all  $\lambda$ -design with  $\lambda = 3$ . J. *Combinatorial Theory*, 8:343–349, 1970.
- [21] W. Bridges and T. Tsaur. Some structural characterizations of  $\lambda$ -designs. Ars Combin., 44:129–135, 1996.
- [22] A. E. Brouwer and A. Schrijver. Uniform hypergraphs. In *Packing and Covering in Combinatorics*. Mathematical Centre Tracts, Amsterdam, 1979.
- [23] R. H. Bruck and H. J. Ryser. The nonexistence of certain finite projective planes. *Canadian J. Math.*, 1:88–93, 1949.
- [24] A. Chowdhury. Colouring subspaces. Master's thesis, University of Waterloo, 2005.
- [25] A. Chowdhury. On a conjecture of Frankl and Füredi. *Electron. J. Combin.*, 18(1):Paper 56, 16, 2011.
- [26] A. Chowdhury, C. Godsil, and G. Royle. Colouring lines in projective space. J. Combin. Theory Ser. A, 113(1):39–52, 2006.
- [27] A. Chowdhury and B. Patkós. Shadows and intersections in vector spaces. J. Combin. Theory Ser. A, 117(8):1095–1106, 2010.
- [28] S. Chowla and H. J. Ryser. Combinatorial problems. *Canadian J. Math.*, 2:93–99, 1950.

- [29] W. S. Connor, Jr. On the structure of balanced incomplete block designs. Ann. Math. Statistics, 23:57–71, 1952.
- [30] W.H. Cunningham and A.S. Lewis. *Mathematical Optimization: Course Notes for C&O 355*. University of Waterloo, Waterloo, Canada, 2000.
- [31] E. Czabarka and L. Székely. An alternative shifting proof to Hsieh's theorem. In Proceedings of the Twenty-ninth Southeastern International Conference on Combinatorics, Graph Theory and Computing (Boca Raton, FL, 1998), volume 134, pages 117–122, 1998.
- [32] G. B. Dantzig. Application of the simplex method to a transportation problem. In Activity Analysis of Production and Allocation, Cowles Commission Monograph No. 13, pages 359–373. John Wiley & Sons Inc., New York, N. Y., 1951.
- [33] D. E. Daykin. Erdös-Ko-Rado from Kruskal-Katona. J. Combinatorial Theory Ser. A, 17:254–255, 1974.
- [34] N. G. de Bruijn and P. Erdős. On a combinatorial problem. *Nederl. Akad. Wetensch.*, *Proc.*, 51:1277–1279 = Indagationes Math. 10, 421–423 (1948), 1948.
- [35] P. Delsarte. An algebraic approach to the association schemes of coding theory. *Philips Res. Rep. Suppl.*, (10):vi+97, 1973.
- [36] P. Delsarte. Association schemes and t-designs in regular semilattices. J. Combinatorial Theory Ser. A, 20(2):230–243, 1976.
- [37] M. Deza. Une propriété extrémale des plans projectifs finis dans une classe de codes équidistants. *Discrete Math.*, 6:343–352, 1973.
- [38] M. Deza. Solution d'un problème de Erdős-Lovász. J. Combinatorial Theory Ser. B, 16:166–167, 1974.
- [39] M. Deza and P. Frankl. The Erdős-Ko-Rado theorem—22 years later. *SIAM J. Algebraic Discrete Methods*, 4(4):419–431, 1983.
- [40] I. Dinur and S. Safra. On the hardness of approximating minimum vertex cover. *Ann. of Math.* (2), 162(1):439–485, 2005.
- [41] D. Dummit and R. Foote. *Abstract algebra*. John Wiley & Sons Inc., Hoboken, NJ, third edition, 2004.
- [42] J. Eisfeld and L. Storme. (partial) t-spreads and minimal t-covers in finite spaces. lecture notes, available at http://cage.ugent.be/~fdc/intensivecourse2/ storme2.ps, 2000.

- [43] S. El-Zanati, H. Jordon, G. Seelinger, P. Sissokho, and L. Spence. The maximum size of a partial 3-spread in a finite vector space over GF(2). *Des. Codes Cryptogr.*, 54(2):101–107, 2010.
- [44] D. Ellis. Irredundant families of subcubes. Math. Proc. Cambridge Philos. Soc., 150(2):257–272, 2011.
- [45] D. Ellis, Y. Filmus, and E. Friedgut. Triangle-intersecting families of graphs. J. *Eur. Math. Soc. (JEMS)*, to appear.
- [46] D. Ellis, E. Friedgut, and H. Pilpel. Intersecting families of permutations. J. *Amer. Math. Soc.*, 24(3):649–682, 2011.
- [47] P. Erdős, C. Ko, and R. Rado. Intersection theorems for systems of finite sets. *Quart. J. Math. Oxford Ser.* (2), 12:313–320, 1961.
- [48] P.L. Erdős, U. Faigle, and W. Kern. A group-theoretic setting for some intersecting Sperner families. *Combin. Probab. Comput.*, 1(4):323–334, 1992.
- [49] T. Etzion and A. Vardy. Error-correcting codes in projective space. *IEEE Trans. Inform. Theory*, 57(2):1165–1173, 2011.
- [50] T. Etzion and A. Vardy. On *q*-analogs of Steiner systems and covering designs. *Adv. Math. Commun.*, 5(2):161–176, 2011.
- [51] R. A. Fisher. An examination of the different possible solutions of a problem in incomplete blocks. *Ann. Eugenics*, 10:52–75, 1940.
- [52] P. Frankl. On Sperner families satisfying an additional condition. *J. Combinatorial Theory Ser. A*, 20(1):1–11, 1976.
- [53] P. Frankl. A new short proof for the Kruskal-Katona theorem. *Discrete Math.*, 48(2-3):327–329, 1984.
- [54] P. Frankl and Z. Füredi. Nontrivial intersecting families. *J. Combin. Theory Ser. A*, 41(1):150–153, 1986.
- [55] P. Frankl and Z. Füredi. A sharpening of Fisher's inequality. *Discrete Math.*, 90(1):103–107, 1991.
- [56] P. Frankl and R.L. Graham. Intersection theorems for vector spaces. *European J. Combin.*, 6(2):183–187, 1985.
- [57] P. Frankl and N. Tokushige. The katona theorem for vector spaces. preprint, 2011.
- [58] P. Frankl and N. Tokushige. On *r*-cross intersecting families of sets. Preprint, available at http://www.cc.u-ryukyu.ac.jp/~hide/3c7.pdf, 2011.

- [59] P. Frankl and R. M. Wilson. Intersection theorems with geometric consequences. *Combinatorica*, 1(4):357–368, 1981.
- [60] P. Frankl and R. M. Wilson. The Erdős-Ko-Rado theorem for vector spaces. J. Combin. Theory Ser. A, 43(2):228–236, 1986.
- [61] D. Gale, H. W. Kuhn, and A. W. Tucker. Linear programming and the theory of games. In *Activity Analysis of Production and Allocation*, Cowles Commission Monograph No. 13, pages 317–329. John Wiley & Sons Inc., New York, N. Y., 1951.
- [62] B. Gardner. On coverings and  $(r, \lambda)$ -systems. PhD thesis, University of Waterloo, 1972.
- [63] B. Gardner and S. A. Vanstone. Some results on irreducible  $(r, \lambda)$ -designs. *Utilitas Math.*, 18:291–300, 1980.
- [64] C. Godsil and K. Meagher. A new proof of the Erdős-Ko-Rado theorem for intersecting families of permutations. *European J. Combin.*, 30(2):404–414, 2009.
- [65] C. Godsil and G. Royle. *Algebraic graph theory*, volume 207 of *Graduate Texts in Mathematics*. Springer-Verlag, New York, 2001.
- [66] C.D. Godsil and M.W. Newman. Independent sets in association schemes. Combinatorica, 26(4):431–443, 2006.
- [67] C. Greene and D.J. Kleitman. Proof techniques in the theory of finite sets. In *Studies in combinatorics*, volume 17 of *MAA Stud. Math.*, pages 22–79. Math. Assoc. America, Washington, D.C., 1978.
- [68] J. I. Hall. On two conjectures of Stanton and Mullin. J. Combinatorial Theory Ser. A, 22(2):153–162, 1977.
- [69] L. H. Harper and F. Hergert. The isoperimetric problem in finite projective planes. In Proceedings of the Twenty-fifth Southeastern International Conference on Combinatorics, Graph Theory and Computing (Boca Raton, FL, 1994), volume 103, pages 225–232, 1994.
- [70] A. J. W. Hilton and E. C. Milner. Some intersection theorems for systems of finite sets. *Quart. J. Math. Oxford Ser.* (2), 18:369–384, 1967.
- [71] A. J. Hoffman and J. B. Kruskal. Integral boundary points of convex polyhedra. In *Linear inequalities and related systems*, Annals of Mathematics Studies, no. 38, pages 223–246. Princeton University Press, Princeton, N. J., 1956.
- [72] W.N. Hsieh. Intersection theorems for systems of finite vector spaces. *Discrete Math.*, 12:1–16, 1975.

- [74] J. Kahn and G. Kalai. A counterexample to Borsuk's conjecture. Bull. Amer. Math. Soc. (N.S.), 29(1):60–62, 1993.
- [75] G. Katona. A theorem of finite sets. In *Theory of graphs (Proc. Colloq., Tihany, 1966)*, pages 187–207. Academic Press, New York, 1968.
- [76] G. Katona. A simple proof of the Erdős-Chao Ko-Rado theorem. J. Combinatorial Theory Ser. B, 13:183–184, 1972.
- [77] P. Keevash. Shadows and intersections: stability and new proofs. *Adv. Math.*, 218(5):1685–1703, 2008.
- [78] M. Kneser. Aufgabe 300. Jber. Deutsch. Math.-Verein., 58, 1955.
- [79] E. Kramer. On  $\lambda$ -designs. PhD thesis, University of Michigan, 1969.
- [80] J.B. Kruskal. The number of simplices in a complex. In *Mathematical optimization techniques*, pages 251–278. Univ. of California Press, Berkeley, Calif., 1963.
- [81] C. W. H. Lam, L. Thiel, and S. Swiercz. The nonexistence of finite projective planes of order 10. *Canad. J. Math.*, 41(6):1117–1123, 1989.
- [82] L. Lovász. Flats in matroids and geometric graphs. In *Combinatorial surveys* (*Proc. Sixth British Combinatorial Conf., Royal Holloway Coll., Egham, 1977*), pages 45–86. Academic Press, London, 1977.
- [83] L. Lovász. Kneser's conjecture, chromatic number, and homotopy. J. Combin. Theory Ser. A, 25(3):319–324, 1978.
- [84] L. Lovász. Combinatorial problems and exercises. North-Holland Publishing Co., Amsterdam, second edition, 1993.
- [85] K. Majindar. On some theorems in combinatorics relating to incomplete block designs. Ann. Math. Statistics, 24:377–389, 1953.
- [86] K. Majindar. On the parameters and intersection of blocks of balanced incomplete block designs. Ann. Math. Statist., 33:1200–1205, 1962.
- [87] N. Manickam. *On the distribution invariants of the association scheme*. PhD thesis, Ohio State University, 1986.
- [88] N. Manickam and D. Miklós. On the number of nonnegative partial sums of a nonnegative sum. In *Combinatorics (Eger, 1987)*, volume 52 of *Colloq. Math. Soc. János Bolyai*, pages 385–392. North-Holland, Amsterdam, 1988.
- [89] N. Manickam and N. M. Singhi. First distribution invariants and EKR theorems. J. Combin. Theory Ser. A, 48(1):91–103, 1988.
- [90] W. Mantel. Problem 28. Wiskundige Opgaven, 10:60-61, 1907.
- [91] G. Marino and G. Chiaselotti. A method to count the positive 3-subsets in a set of real numbers with non-negative sum. *European J. Combin.*, 23(5):619–629, 2002.
- [92] D. McCarthy and S. A. Vanstone. On the structure of regular pairwise balanced designs. *Discrete Math.*, 25(3):237–244, 1979.
- [93] D. Mubayi and J. Verstraëte. Proof of a conjecture of Erdős on triangles in setsystems. *Combinatorica*, 25(5):599–614, 2005.
- [94] M.W. Newman. *Independent Sets and Eigenspaces*. PhD thesis, University of Waterloo, 2004.
- [95] R. Peltesohn. *Das Turnierproblem für Spiele zu je drein*. PhD thesis, Berlin, 1936.
- [96] J.-E. Pin. On two combinatorial problems arising from automata theory. In Combinatorial mathematics (Marseille-Luminy, 1981), volume 75 of North-Holland Math. Stud., pages 535–548. North-Holland, Amsterdam, 1983.
- [97] D.K. Ray-Chaudhuri and R.M. Wilson. On *t*-designs. Osaka J. Math., 12(3):737– 744, 1975.
- [98] H. J. Ryser. A note on a combinatorial problem. *Proc. Amer. Math. Soc.*, 1:422–424, 1950.
- [99] H. J. Ryser. An extension of a theorem of de Bruijn and Erdős on combinatorial designs. J. Algebra, 10:246–261, 1968.
- [100] B. Segre. Teoria di Galois, fibrazioni proiettive e geometrie non desarguesiane. *Ann. Mat. Pura Appl.* (4), 64:1–76, 1964.
- [101] Å. Seress. On  $\lambda$ -designs with  $\lambda = 2P$ . In Coding theory and design theory, *Part II*, volume 21 of *IMA Vol. Math. Appl.*, pages 290–303. Springer, New York, 1990.
- [102] Å. Seress. All lambda-designs with  $\lambda = 2p$  are type-1. Des. Codes Cryptogr., 22(1):5–17, 2001.
- [103] S. S. Shrikhande and N. M. Singhi. Some combinatorial problems. In *Combinatorics and applications (Calcutta, 1982)*, pages 340–359. Indian Statist. Inst., Calcutta, 1984.

- [104] R. G. Stanton and R. C. Mullin. Inductive methods for balanced incomplete block designs. Ann. Math. Statist., 37:1348–1354, 1966.
- [105] M. Steele. *The Cauchy-Schwarz master class*. MAA Problem Books Series. Mathematical Association of America, Washington, DC, 2004. An introduction to the art of mathematical inequalities.
- [106] N. Tokushige. The eigenvalue method for cross *t*-intersecting families. preprint, 2011.
- [107] S. Topalova and S. Zhelezova. 2-spreads and transitive and orthogonal 2-parallelisms of PG(5,2). *Graphs Combin.*, 26(5):727–735, 2010.
- [108] T. Tsaur. Variants of symmetric block designs. PhD thesis, The University of Wyoming, 1993.
- [109] P. Ure. A study of (0, n, n+1)-sets and other solutions of the isoperimetric problem in finite projective planes. PhD thesis, California Institute of Technology, 1996.
- [110] J. H. van Lint and R. M. Wilson. *A course in combinatorics*. Cambridge University Press, Cambridge, second edition, 2001.
- [111] S. A. Vanstone. Irreducible regular pairwise balanced designs. *Utilitas Math.*, 15:249–259, 1979.
- [112] J. von Neumann. Collected works. Vol. VI: Theory of games, astrophysics, hydrodynamics and meteorology. General editor: A. H. Taub. A Pergamon Press Book. The Macmillan Co., New York, 1963.
- [113] J. Wang. Intersecting antichains and shadows in linear lattices. *J. Combin. Theory Ser. A*, 118(7):2092–2101, 2011.
- [114] I. Weisz. *Lambda-designs with small lambda are type-1*. PhD thesis, The Ohio State University, 1995.
- [115] C. Whitehead, editor. Surveys in combinatorics 1987, volume 123 of London Mathematical Society Lecture Note Series. Cambridge University Press, Cambridge, 1987. Papers from the eleventh British combinatorial conference held in New Cross, July 13–17, 1987.
- [116] R.M. Wilson. The exact bound in the Erdős-Ko-Rado theorem. *Combinatorica*, 4(2-3):247–257, 1984.
- [117] D. R. Woodall. Square  $\lambda$ -linked designs. *Proc. London Math. Soc.* (3), 20:669–687, 1970.